Southampton

University of Southampton Research Repository

Copyright © and Moral Rights for this thesis and, where applicable, any accompanying data are retained by the author and/or other copyright owners. A copy can be downloaded for personal non-commercial research or study, without prior permission or charge. This thesis and the accompanying data cannot be reproduced or quoted extensively from without first obtaining permission in writing from the copyright holder/s. The content of the thesis and accompanying research data (where applicable) must not be changed in any way or sold commercially in any format or medium without the formal permission of the copyright holder/s.

When referring to this thesis and any accompanying data, full bibliographic details must be given, e.g.

Thesis: Ruktanonchai CW (2020) "Mapping Spatial and Temporal Inequalities in Utilisation of Maternal and Newborn Care in Five East African Countries", University of Southampton, Faculty of Environmental and Life Sciences, Geography and Environment, PhD Thesis, pp 182.

Data: Ruktanonchai CW (2020) Mapping Spatial and Temporal Inequalities in Utilisation of Maternal and Newborn Care in Five East African Countries. www.worldpop.org

University of Southampton

Faculty of Environmental and Life Sciences

Geography and Environmental Science

Mapping Spatial and Temporal Inequalities in Utilisation of Maternal and Newborn Care in Five East African Countries

by

Corrine Warren Ruktanonchai

ORCID ID: 0000-0002-7889-3473

Thesis for the degree of **Doctor of Philosophy**

June 2020

University of Southampton

<u>Abstract</u>

Faculty of Environmental and Life Sciences

Geography and Environmental Science

Thesis for the degree of **Doctor of Philosophy**

Mapping Spatial and Temporal Inequalities in Utilisation of Maternal and Newborn Care in Five East African Countries

by

Corrine Warren Ruktanonchai

Historically, maternal and newborn health (MNH) outcomes used to monitor progress in achieving global and national targets have been measured at an aggregate level, showing vast inequalities between and within countries. To ensure no one is left behind in improving health, researchers have called for the spatial and temporal disaggregation of MNH data. This thesis aims to generate high spatial resolution data over time that can be used to monitor progress in reducing inequalities amongst utilisation of key MNH services in the East African Community (EAC) region, including Burundi, Kenya, Rwanda, Tanzania, and Uganda.

Following a 'three-paper' format, the first paper in this thesis employs a hierarchical mixed effects logistic regression framework, to estimate the odds of: 1) skilled birth attendance (SBA), 2) receiving 4+ antenatal care (ANC) visits, and 3) receiving a postnatal health check-up (PNC) within 48 hours of delivery. Model results are applied to an accessibility surface to visualise the probabilities of obtaining MNH care at both high-resolution and sub-national levels after adjusting for live births in 2015. Across all outcomes, decreasing wealth and education levels are associated with lower odds of obtaining MNH care, while increasing geographic inaccessibility scores are associated with the strongest effect in lowering odds of obtaining care observed across outcomes, with the widest disparities observed among skilled birth attendance.

The second paper explores temporal trends in absolute and relative spatial inequalities in utilisation of these MNH services between 1990 and 2015. A Bayesian framework is employed to generate sub-national estimates of utilisation of SBA, ANC, and PNC over several time points. Absolute change in estimates over time is reported, as well as relative change in ratios of the best-to-worst performing districts per country. Across all countries, the greatest spatial equality is observed among ANC, while SBA and PNC tend to have greater spatial variability. Lastly, while progress has been made to reduce coverage gaps between districts, improvement in PNC coverage has stagnated and should be monitored closely over the coming decades.

The final paper comprising this work explores the trade-off between increasing spatial resolution in model inputs and resulting model uncertainty, with aims of understanding the optimal spatial resolution to report health outcomes. Prevalence of childbirth via c-section is estimated in Tanzania, using geospatial covariates at varying levels of spatial coarseness within a Bayesian model framework. Uncertainty in posterior outcomes is reported as the distribution of 95% credible intervals at each spatial resolution, and visualised at the spatial resolution with the greatest model precision. Overall, higher spatial resolution input increases model uncertainty, while model precision is best approximated at the highest spatial resolution, suggesting an important policy trade-off between identifying concealed spatial heterogeneities in health indicators.

This thesis makes substantive contributions to the literature by outlining where spatial inequalities in key MNH services are occurring within the EAC region and how these disparities are evolving over time. This work also makes methodological contributions by demonstrating how spatial approaches can be used to monitor health indicators, as well as exploring uncertainty in the application of these techniques, with important implications in communicating results to policy makers. These techniques can be applied across health and development outcomes, notably across Sustainable Development Goal indictors, ensuring "no one left behind" by 2030.

Table of Contents

Table	of Conter	1ts	i
List of	f Tables		v
List of	f Figures.		vi
Resear	rch Thesis	s: Declaration of Authorship	X
Ackno	wledgem	ents	xi
Defini	tions and	Abbreviations	xii
Chapt	er 1:	Introduction	1
1.1	Backgro	ound	1
	1.1.1	Global initiatives in maternal and newborn health	4
	1.1.2	Maternal and newborn health in sub-Saharan Africa	6
1.2	Study o	bjectives	7
Chapt	er 2:	Literature Review	9
2.1	Materna	al and newborn health	9
	2.1.1	Causes of maternal death	10
	2.1.2	Causes of newborn deaths	12
	2.1.3	Maternal and newborn health services	15
	2.1.4	Barriers to utilisation of health services	18
	2.1.5	Monitoring maternal and newborn health outcomes in the SDG era	22
	2.1.6	Inequalities in maternal and newborn health	24
2.2	Spatial	demographic data for health in Sub-Saharan Africa	29
	2.2.1	GIS utilisation and the availability of spatial demographic data	30
	2.2.2	The evolution of mapping demographic data and current approaches	37
	2.2.3	Applications in producing spatial demographic data	43
	2.2.4	Future directions and recommendations	47
Chapt	er 3:	Study framework	49
3.1	Concept	tual framework	49
3.2	Researc	h questions	51
	3.2.1	Chapter 4 research questions	51
	3.2.2	Chapter 5 research questions	51

	3.2.3	Chapter 6 research questions	. 52
3.3	Study in	ndicators	. 52
	3.3.1	Antenatal care	. 52
	3.3.2	Skilled birth attendance	. 54
	3.3.3	Postnatal care	. 55
	3.3.4	Delivery via caesarean section	. 56
3.4	Study a	rea	. 57
	3.4.1	Burundi	. 58
	3.4.2	Kenya	. 59
	3.4.3	Rwanda	. 60
	3.4.4	Tanzania	. 62
	3.4.5	Uganda	. 63
3.5	Study n	nethods	. 64
	3.5.1	Hierarchical mixed effects modelling	. 65
	3.5.2	Bayesian modelling	. 66
3.6	Scientif	ic contribution	. 67
	3.6.1	Knowledge gaps	. 67
	3.6.2	Contribution to the literature	. 68
	3.6.3	Impact	. 69
	3.6.4	Intellectual contribution	. 70
Chapt	ter 4:	Equality in maternal and newborn health: Modelling geographic	
	dispar	rities in utilisation of care in five East African countries	.71
4.1	Abstrac	t	.71
4.2	Introduc	ction	.71
4.3	Method	S	. 73
	4.3.1	Overview	.73
	4.3.2	Data	.74
	4.3.3	Creating an impedance surface and inaccessibility score	.76
	4.3.4	Analysis of MNH outcomes	. 78
4.4	Results		. 80
	4.4.1	Gridded accessibility surface	. 82
	4.4.2	Explanatory variables of MNH care utilisation	. 82

	4.4.3	High-resolution mapping of MNH care utilisation	
	4.4.4	Sub-national mapping of MNH care utilisation	86
	4.4.5	Model validation	
4.5	Discuss	sion	
	4.5.1	Limitations	90
4.6	Conclus	sions	
4.7	Acknow	vledgements	
4.8	Intellec	tual contribution	92
Chapt	er 5:	Temporal trends in spatial inequalities of maternal and new	born
	health	ı services among four East African countries, 1999 – 2015	
5.1	Abstrac	rt	
5.2	Introdu	ction	
5.3	Method	ls	
	5.3.1	Data	
	5.3.2	Absolute indices of temporal change	
	5.3.3	Relative indices of temporal change	
	5.3.4	Model validation	
	5.3.5	Ethics approval	
5.4	Results		
	5.4.1	Absolute indices of temporal change	
	5.4.2	Relative indices of temporal change	105
	5.4.3	Model validation	108
5.5	Discuss	sion	110
	5.5.1	Limitations	112
5.6	Conclus	sions	
5.7	Acknow	vledgements	113
5.8	Intellec	tual contribution	113
Chapt	er 6:	Estimating uncertainty in geospatial modelling at multiple s	patial
	resolu	itions: the pattern of delivery via caesarean section in Tanzan	114 ia
6.1	Abstrac	t	
6.2	Introdu	ction	

6.3	Method	ls	117
	6.3.1	DHS data	
	6.3.2	Covariate data	
	6.3.3	Model framework	
	6.3.4	Ethics approval	
6.4	Results		
6.5	Discuss	sion	
	6.5.1	Limitations	
6.6	Conclus	sions	
6.7	Acknow	wledgments	
6.8	Intellec	tual contribution	
Chapt	er 7:	Conclusions	
7.1	Key fin	ıdings	
7.2	Thesis of	contributions	134
7.3	Limitati	ions	
7.4	Future	work	
7.5	Policy a	and development implications	
7.6	Conclus	sions	141
Appen	dices		
Appen	dix A	Chapter 4 Appendix	
A.1	Chapter	r 4 Supplementary Figures	
Appen	dix B	Chapter 5 Appendix	
B .1	Chapter	r 5 Supplementary Tables	145
B.2	Chapter	r 5 Supplementary Figures	
Appen	dix C	Chapter 6 Appendix	
C.1	Chapter	r 6 Supplementary Figures	
Biblio	graphy		

List of Tables

Table 2.1 Selected MDG and SDG targets and indicators pertaining to maternal and newborn
health23
Table 2.2 Sources of spatially explicit demographic data, as outlined by Tatem et al., 201232
Table 4.1 MNH outcomes within the study area disaggregated by sub-national DHS region and national total, gathered through STATcompiler (ICF International, 2015)81
Table 4.2 Hierarchical mixed effects logistic regression model odds ratios of MNH outcomes among female DHS respondents in five East African countries (N = 25,325)83
Table 5.1 DHS survey used in study analysis and characteristics. 97
Table 5.2 Model validation results
Table 6.1 Marginal effects of the fixed effects and hyperparameters of the posterior c-section
models at 5km, 50km, and 100km122
Table 7.1 Key research findings, by study objective 131
Table B.2 Unadjusted logistic regression results, with worst performing DHS region as reference
as compared to best performing DHS region145
Table B.3 Model fit, mean posterior estimates and hyperparameters of modelled MNH outcomes

List of Figures

Figure 2.1 Global ca	auses of maternal deaths in 1990 and 2013, as published by Requejo and
Bl	hutta, 2015
Figure 2.2 Distribut	ion of causes of maternal deaths by WHO geographical region, as published by
K	han et al., 2006
Figure 2.3 Global ca	auses of under-5 mortality in 2000 and 2013, as published by Requejo and
Bl	hutta, 2015
Figure 2.4 Global tr 20	ends in under-5 childhood mortality in 2000 - 2010, as published by Liu et al., 012
Figure 2.5 Global ca	auses of under-5 mortality, separated into neonatal deaths (0 - 27 days) and
de	eaths among children aged 1-59 months, as published by Liu et al., 201414
Figure 2.6 The repro	oductive life course across which care packages must be integrated, adapted
fro	om Kerber et al., 2007
Figure 2.7 Median c	coverage of interventions among Countdown countries, across the continuum of re as published by Victora et al. 2016
Cu	
Figure 2.8 Phases of	f delay and factors affecting utilisation of maternal health services, as published
by	Gabrysch and Campbell, 2009 19
Figure 2.9 Framewo	ork for assessing quality of care in maternity services, as published by Hulton et, 2007
Figure 2.10 Skilled	birth attendance in 83 low- and middle-income countries, as published by
W	НО 2015а25
Figure 2.11 Modern	contraceptive prevalence among 71 low- and middle-income countries, as
pu	blished by WHO 2015a
Figure 2.12 DTP3 in	mmunization coverage among 78 low- and middle-income countries, as
pu	blished by WHO 2015a
Figure 2.13 Change	over time (absolute and absolute excess change) in DTP3 immunization
co 20	overage among 41 low- and middle-income countries, as published by WHO
20	15a
Figure 2.14 Compos	site coverage index for 8 selected RMNCH interventions in selected low- and
m	iddle-income countries, as reported by WHO 2015a

Figure 2.15. Availability of demographic datasets informing subnational population attributes,
among a) census microdata, b) national household survey programmes, and c)
national household survey programmes with corresponding GPS information.
Adapted from (Tatem et al., 2012)
Figure 2.16. Population density of the East Africa region (Burundi, Kenya, Rwanda, Uganda,
Tanzania), with selected inset cities, using a land-cover based redistribution
approach, adapted from (Tatem et al., 2007)
Figure 2.17. High-resolution populations counts redistributed using Kenya 1999 census data and
ancillary data sources (freely available at worldpop.org.uk). Adapted from
(Stevens et al., 2015)40
Figure 2.18. Top-down (A) versus bottom-up (B) approaches to mapping gridded populations,
adapted from (Wardrop et al., 2018)41
Figure 2.19. Spatial demographic datasets for Africa. A) Proportion of children under 5; B)
proportion of women of child-bearing age; and, C) children under 5 in 2010 at
1km resolution. Adapted from (A. J. Tatem et al., 2013)
Figure 2.20. Age-specific fertility rates among those aged a) 20 - 24, b) 30 - 34, and c) 40 - 44, plus
estimated live births per 100 x 100 m, Tanzania, 2012. Adapted from (Tatem et
al., 2014a)
Figure 3.1 Study framework for monitoring and evaluating maternal and newborn health care
utilisation
Figure 3.2. Distribution of women of childbearing age in Burundi, 2015. White lines delineate 2010
DHS boundaries, while grey lines delineate administrative II units59
Figure 3.3. Distribution of women of childbearing age in Kenya, 2015. White lines delineate 2014
DHS boundaries, while grey lines delineate administrative II units (as of 2013).
Figure 3.4. Distribution of women of childbearing age in Rwanda, 2015. White lines delineate 2015
DHS boundaries, while grey lines delineate administrative II units61
Figure 3.5. Distribution of women of childbearing age in Tanzania, 2015. White lines delineate
2015 DHS boundaries, while grey lines delineate administrative II units63
Figure 3.6. Distribution of women of childbearing age in Uganda, 2015. White lines delineate 2016
DHS boundaries, while grey lines delineate administrative II units

Figure 4.1 Mapp	ing inaccessibility to the nearest health facility. A) Impedance surface representing
	the difficulty in traversing a given 300 x 300 m cell. B) Health facility locations
	used as destinations. C) Accessibility surface generated via cost-distance analysis
	using inputs A) and B), representing the least accumulative geographic "cost"
	value for a 300 x 300 m cell in accessing the nearest health facility
Figure 4.2 Hierar	chical covariates included in the mixed-effects logistic regression analysis.
	Covariates are listed in boxes with corresponding hierarchy and fixed versus
	random effect79
Figure 4.3 Preval	ence of skilled attendance at delivery, no antenatal care, and postnatal care
	provided within 48 hours of delivery, by DHS region
Figure 4.4 Boxpl	ots of modelled probabilities of skilled birth attendance (SBA), antenatal care
	(ANC), and postnatal care (PNC)
Figure 4.5 Model	lled probability surfaces representing the spatial effect of accessibility at 300 x 300
	m. a) Delivery with a skilled birth attendant (SBA) present, b) Four or more
	antenatal care (ANC) visits at time of delivery, and c) Postnatal care (PNC)
	received within 48 hours of delivery
Figure 4.6 Births adjusted probability maps representing the probability of obtaining MNH care for	
	a given birth at the administrative II unit. A) Delivery with a skilled birth
	attendant (SBA) present, B) Four or more antenatal care (ANC) visits at time of
	delivery, and C) Postnatal care (PNC) received within 48 hours of delivery. 87
Figure 4.7 ROC	curves for skilled birth attendance (SBA), antenatal care (ANC), and postnatal
	care (PNC) models, with associated area under the curve (AUC) metrics88
Figure 5.1 Predic	ted prevalence and absolute change in i) skilled birth attendance (blue), ii) 4+
	antenatal care visits (green), and iii) postnatal check-up within 48 hours (red)
	among administrative II units, among DHS data in a) Kenya, 2003 – 2014 b)
	Tanzania, 1999 – 2015, c) Rwanda, 2005 – 2014, and d) Uganda, 2000 – 2011.
Figure 5.2 DIC v	alues for skilled birth attendance (blue), 4+ antenatal care visits (green), and
	postnatal care (red) models, by country, DHS data, 1999 – 2015 104
Figure 5.3 Ratio	of modelled best-to-worst performing administrative II units over time for skilled
	birth attendance (blue), 4+ antenatal care visits (green), and postnatal care (red),
	by country, DHS data, 1999 - 2015

Figure 5.4 Unadjusted odds ratio of skilled birth attendance (blue), 4+ antenatal care visits (green),
and postnatal care (red) over time, with worst performing DHS region as
reference. DHS data. 1999 – 2015
,
Figure 6.1 Delivery by caesarean section at the administrative 1 level using DHS data, Tanzania,
2015
Figure 6.2 Study analysis flowshort
Figure 0.2 Study analysis nowchart
Figure 6.3 Violin plot of posterior 95% credible intervals for c-section estimates predicted at the
5km. 50km. and 100km scale
Figure 6.4 Modelled c-section prevalence estimates (left) and associated 95% credible interval
(right) at 5km (top), 50km (middle), and 100km (bottom), Tanzania, 2015.125
Figure 4.1 Case located DUS electors (N = 2.211) by symptom of DUS second onto (N = 26.179)
Figure A.1 Geo-located DHS clusters ($N = 5,511$) by humber of DHS respondents ($N = 50,178$)
and urban versus rural location in five East African countries
Figure A.2 Estimated live births in five East African countries. Birth estimates were generated by
the WorldPop project (www.worldpop.org) and are shown for the year 2015 at
$100 \times 100 \text{ m resolution}$
100 x 100 m resolution
Figure B.3 Absolute change in a) 4+ antenatal care visits (green), b) postnatal care check-up within
48 hours (red), and c) skilled birth attendance (blue), Kenya DHS data, 2003 –
2014, ordered by administrative II unit
Figure B.4 Absolute change a) 4+ antenatal care visits (green), b) postnatal care check-up within 48
hours (red), and c) skilled birth attendance (blue), Tanzania DHS data, 1999 -
2015, ordered by administrative II unit
Figure B.5 Absolute change a) 4+ antenatal care visits (green), b) postnatal care check-up within 48
hours (red), and c) skilled birth attendance (blue), Rwanda DHS data, 2005 -
2014, ordered by administrative II unit
Figure B.6 Absolute change in a) 4+ antenatal care visits (green), b) postnatal care check-up within
48 hours (red) and c) skilled birth attendance (blue). Uganda DHS data 2000 -
2011 ordered by administrative II unit
2011, ordered by administrative if unit
Figure C.7 Marginal effect density plots at 5km (top), 50km (middle), and 100km (bottom) 163
Figure C.8. Model variance, model range, and model random effects at the 5km (top), 10km
(middle) and 100km (bottom) scales

Research Thesis: Declaration of Authorship

Print name:	Corrine Warren Ruktanonchai
Title of thesis:	Mapping Spatial and Temporal Inequalities in Utilisation of Maternal and Newborn Care in Five East African Countries

I declare that this thesis and the work presented in it are my own and has been generated by me as the result of my own original research.

I confirm that:

- 1. This work was done wholly or mainly while in candidature for a research degree at this University;
- 2. Where any part of this thesis has previously been submitted for a degree or any other qualification at this University or any other institution, this has been clearly stated;
- 3. Where I have consulted the published work of others, this is always clearly attributed;
- 4. Where I have quoted from the work of others, the source is always given. With the exception of such quotations, this thesis is entirely my own work;
- 5. I have acknowledged all main sources of help;
- 6. Where the thesis is based on work done by myself jointly with others, I have made clear exactly what was done by others and what I have contributed myself;
- 7. Parts of this work have been published as:

Ruktanonchai CW, Ruktanonchai NW, Pezzulo C, Alegana VA, Bosco C, Nove A, Lopes S, Bytyqi A, Ayiko R, Charles A, Lambert N, Msechu E, Matthews Z, Tatem AJ. *Equality in maternal and newborn health: Modelling geographic disparities in utilisation of care in five East African countries.* PLoS ONE. 2016, 11(8): e0162006. DOI: 10.1371/journal.pone.0162006

Ruktanonchai CW, Nilsen K, Alegana VA, Bosco C, Ayiko R, Charles ASK, Matthews Z, Tatem AJ. *Temporal trends in spatial inequalities of maternal and newborn outcomes among four East African countries, 1999 – 2015.* BMC Public Health. 2018, 18:1339. DOI: 10.1186/s12889-018-6241-8

Ruktanonchai CW, Nieves JJ, Ruktanonchai NW, Nilsen K, Steele JE, Matthews Z, Tatem AJ. *Estimating uncertainty in geospatial modelling at multiple spatial resolutions: the pattern of delivery via caesarean section in Tanzania.* BMJ Global Health. 2020, 4: e002092. DOI: 10.1136/bmjgh-2019-002092

Signature:

Date:

Acknowledgements

This work is dedicated to Jesse and Silas, and to all the women who weren't so fortunate.

To Nick, whose intellectual and emotional support provided me with the will to carry on, and whose culinary passion provided me with the strength to do so. I've grown as a person and a scientist because of you, and both my life and my work are enriched for having you in it.

To Bear, who helped me ground-truth the indicators in this thesis, like accessibility to my antenatal care appointments, and having an emergency c-section $(\ensuremath{\overline{0}}\xspace \ensuremath{\overline{0}}\xspace \ensuremath{\overline{0}}\xspace$). You gave this thesis a face, little baby, and now I go to work so everyone can experience the joy you give me every day.

To Olly, who provided cuddles and purrs on demand, no questions asked.

To Mom and Dad, who always believed in me and selflessly pushed me to chase my dreams, even if those dreams took me to the Arctic Circle, the South Pacific, or the United Kingdom. I couldn't have done any of this without your love and support, no matter how far away I was.

To my family, Kate, Preston, Sue, Duke, Melissa, and Joel, who chatted with me every day and video called with me every weekend, and made the Atlantic Ocean seem just a little bit smaller.

To my friends near and far, who kept me sane and never let me lose perspective. Thank you for the endless nights of video games and board games which kept me grounded. Maybe one day we'll finally find Timmy.

To my supervisors, Andy Tatem and Zoe Matthews, whose guidance and insight at all stages of the PhD process provided me with the foundations to become an independent and successful scientist. Thank you for giving me the space and autonomy to be creative, to make mistakes, and to find my footing as a researcher (and musician!), while never letting me get lost.

To the post-doctoral researchers and postgraduate students at WorldPop, who provided daily technical support, guidance, and feedback, and allowed me to stand on the shoulders of giants. Special thanks to Victor Alegana, Claudio Bosco, Kristine Nilsen, Jess Steele, Carla Pezzulo, Jeremiah Nieves, and Nick Ruktanonchai.

To the many colleagues and friends I've made along the way, including Andrea Nove, Sarah Neal, Rogers Ayiko, and Andrew Charles, who provided me with valuable professional opportunities throughout my degree, from gaining cultural perspective through knowledge sharing workshops to working with advocates and policy makers at the WHO.

And finally, to the SCDTP, without whom this work would not have been possible. Thank you for taking a chance on my research, and special thanks to Pauline Leonard, Glenn Miller, and Gemma Harris for the unwavering support, funding, and opportunities you provide us with every year.

Definitions and Abbreviations

AIS	AIDS Indicator Surveys
AMDD	Averting Maternal Death and Disability
ANC	Antenatal care
CI	Confidence/credible interval
DHS	Demographic and Health Surveys
DIC	Deviance Information Criterion
EAC	East African Community
EmOC	Emergency obstetric care
EPMM	Ending Preventable Maternal Mortality
GIS	Geographic information system
GPS	Global positioning system
HMIS	Health management information systems
INLA	Integrated Nested Laplace Approximation
SBA	Skilled birth attendance
SSA	Sub-Saharan Africa
MDG	Millennium Development Goals
MIS	Malaria Indicator Surveys
MMR	Maternal mortality ratio
MNH	Maternal and newborn health
OR	Odds ratio
PMNCH	Partnership for Maternal, Newborn and Child Health
PNC	Postnatal care
RMNCAH	Reproductive, maternal, newborn, child, and adolescent health

- SAE Small area estimation
- SD Standard deviation
- SDG Sustainable Development Goals
- SPA Service Provision Assessment
- SPDE Stochastic partial differential equation
- UN United Nations
- UNICEF United Nations International Children's Emergency Fund
- UNFPA United Nations Fund for Population Activities
- WHO World Health Organization

Chapter 1: Introduction

1.1 Background

With the establishment of the Millennium Development Goals (MDGs) in 2000, maternal mortality was nearly halved worldwide by the year 2015, while under-5 mortality was reduced by over 50%, from 90 to 43 deaths per 1,000 livebirths (United Nations, 2015a). Despite this progress, movement towards maternal and child health MDG targets stubbornly lagged behind, with many countries failing to achieve a reduction of under-5 mortality by two-thirds (MDG 4) and maternal mortality by three-quarters (MDG 5). In 2015, the United Nations (UN) renewed their call for action with the establishment of the Sustainable Development Goals (SDGs), aimed at ensuring no one left behind and that future progress in improving health is seen among all populations among all countries. Particular attention has been placed on monitoring progress both within and between countries in order to close health inequalities amongst the most vulnerable (United Nations General Assembly, 2015). Towards this, sub-national monitoring of health disparities within countries has been emphasised to better ensure sustainable progress is achieved by the year 2030, in addition to better stakeholder engagement and capacity strengthening.

As the world moves further into the post-2015 SDG era, however, health inequalities continue to persist among the world's most vulnerable populations (United Nations General Assembly, 2015). Pregnant women, newborns and children remain particularly at risk, and while progress has been made in reducing the total number of women dying in childbirth, the global maternal mortality ratio in 2015 was estimated at 216 deaths per 100,000 live births (WHO et al., 2015). A further 19 newborns per 1,000 live births were estimated to have died within the first 28 days of life in 2015 (UN Inter-agency Group for Child Mortality Estimation, 2015; WHO et al., 2015). These deaths occur predominantly in low resource settings, and the vast majority can be prevented (Alkema et al., 2016a).

The UN has consequently renewed its call for action in preventing these deaths, with ambitious SDG targets of reducing the global maternal mortality ratio to less than 70 per 100,000 live births and neonatal mortality rate to under 12 per 1,000 live births (United Nations General Assembly, 2015). Yet Boldosser-Boesch and colleagues from the Ending Preventable Maternal Mortality (EPMM) work group argue that in addition to this global maternal mortality ratio target, national level targets are also critical in accelerating reduction of maternal mortality by the 2030 timeframe (Boldosser-Boesch et al., 2017). Specifically, the EPMM recommended country targets to increase equity in maternal mortality, where: 1) countries with a baseline MMR in the year 2010 of less than 420 per 100,000 live births should reduce MMR by two-thirds by the year 2030; 2) countries with an MMR in 2010 of greater than 420 per 100,000 live births should not have an MMR greater than

Chapter 1

twice that of the global target (or 140) by 2030; and, 3) countries with a baseline MMR in 2010 of less than 10 per 100,000 should aim to achieve equity in MMR at the sub-national level (WHO and HRP, 2015).

These national level estimates, however, represent a historical push to monitor progress in maternal and newborn health (MNH) outcomes at aggregate levels such as the national and multi-national level, and themselves show vast inequality between countries (Bhutta ZA and Reddy K, 2012). Examining inequality indices among 12 key MNH intervention indicators, Barros and colleagues found substantial variation in coverage of interventions between over 50 countries that account for nearly 95% of all maternal, newborn and child deaths (Barros et al., 2012; Bhutta ZA and Reddy K, 2012). These inequality measures varied not only between countries, but within country, as well, with the most disparities observed among skilled birth attendance at delivery, followed by 4+ antenatal care visits prior to delivery (Barros et al., 2012). These findings suggest that even among countries with improved national-level outcomes, aggregated measures of health may mask local, worsening inequalities among already marginalized and vulnerable populations, such as the poorest and most remote (Bhutta ZA and Reddy K, 2012; Ebener et al., 2015). In order to therefore ensure sustainable and measureable progress amongst all, policymakers have increasingly called for the spatial and temporal disaggregation of MNH data, as evidenced by the recent emphasis on subnational monitoring within the SDGs (United Nations General Assembly, 2015; WHO, 2015a).

Researchers have heeded this call, and recent literature reviews such as those performed by Makanga and colleagues and Ebener et al. suggest that evidence exploring access to healthcare within an explicitly spatial context is emerging (Ebener et al., 2015; Makanga et al., 2016). Both reviews, however, note that a need remains to apply a geographic framework using a more nuanced and spatially explicit approach, helping to account for variation in MNH indicators and outcomes within the context of geography. Specifically, while Ebener et al. note a substantial increase in the number of studies utilising geographical methods to study MNH indicators since 2010, both reviews found that a large majority of spatial research done within an MNH framework has focused predominantly on quantifying geographic access to health facilities (Ebener et al., 2015). This is in line with Makanga et al., who note that a spatially explicit approach quantifying access to health services is a shared theme among other studies. Despite this, they argue a lack of evidence exists correlating these modelled accessibility measures to observed individual-level MNH outcomes, as well as facility-level indicators such as intervention coverage and quality of care indicators (Makanga et al., 2016).

Yet ensuring SDG targets are achieved necessitates not only an increase in spatial disaggregation of data, but also temporal disaggregation to monitor progress (United Nations General Assembly, 2015; WHO, 2015a). By monitoring spatially disaggregated data over time, progress within and between subgroups can be assessed over two or more time points (WHO, 2015a). Temporal disaggregation of data and comparison between those time points present a unique set of

complications, however, as outlined by Barros and Victora (Barros and Victora, 2013). Addressing these complications, they argue that both absolute and relative measures of temporal inequalities should be reported, as these measures can particularly interact and mask increasing or decreasing inequalities over time. For example, Minujin and Delamonica report relative trends over time in under-5 mortality rates among the poorest and richest quintiles within 24 countries (Minujin and Delamonica, 2003). Specifically, they found differentials between the highest and lowest wealth quintiles actually increased over the span of a decade, driven by a reduction in mortality observed among the middle quintiles. These findings suggest that inequalities between the poorest and richest may actually be widening, despite an overall reduction in mortality over time.

Boerma et al. similarly explored coverage of key MNH interventions among the countries most vulnerable to adverse MNH outcomes, as identified in the Countdown to 2015 for Maternal, Newborn and Child Survival initiative (Boerma et al., 2008). Similar to previous studies, they observed wide gaps both between and within countries, but further noted these inequalities persisted throughout time with little to no change, particularly between the poorest and richest quintiles. Lastly, Nguhiu and colleagues examined the change in effective coverage of maternal and child health services over time within Kenya, an aggregate level indicator incorporating information on quality of care, health systems performance, and service use. They similarly found that while effective coverage had steadily increased over time, inequalities in particular services persisted, namely among antenatal care, skilled birth attendance and family planning (Nguhiu et al., 2017). Such consensus suggests that aggregate level targets may not be adequate in ensuring that the most vulnerable sub-populations of a country will see the same improvement observed at a national level, and monitoring progress among these groups over the coming decades will therefore be key to achieving the targets laid out by the SDGs.

The use of Geographic Information Systems (GIS) as a tool to achieve spatial and temporal disaggregation of data and highlight vulnerable populations is becoming increasingly recognized, particularly within the field of MNH (Ebener et al., 2015; Makanga et al., 2016). In an outline of the current state of the geography of MNH, Ebener and colleagues broadly categorized the published use of GIS methodology into three themes in increasing order of complexity: 1) thematic mapping (creation of basic maps to convey a topic or theme), 2) spatial analyses (creation or extraction of new information from spatial data), and 3) spatial modelling (spatial analysis with the use of mathematical or statistical models to simulate real-world phenomena) (Ebener et al., 2015). Similarly, Makanga et al. noted two major themes in the use of GIS in maternal care: modelling access to health services and analysis of risk factors associated with adverse maternal health outcomes (Makanga et al., 2016).

While the historic use of GIS within health literature was limited until more recent decades (Boulos, 2004; Higgs, 2004), the availability of newly disaggregated geographic data and advanced computational infrastructure have allowed greater application of GIS methodologies and tools to

health research (Auchincloss et al., 2012; Musa et al., 2013), as evidenced by the establishment of academic journals such as *International Journal of Health Geographics* (2002) and *Spatial and Spatio-Temporal Epidemiology* (2009), among others. These methods can help fill knowledge gaps for health geographers and epidemiologists, guide policy discussions and intervention efforts in a comprehensive framework and ensure adequate attention to marginalized populations (Boulos, 2004; Goldenberg and McClure, 2012; Higgs, 2004; Makanga et al., 2016; Musa et al., 2013; Nykiforuk and Flaman, 2009). However, the application of GIS in addressing maternal health research has faced unique challenges, as outlined by Molla and colleagues (Molla et al., 2017). These challenges broadly included: limited high quality, geo-referenced MNH data at sufficient spatial resolutions to allow spatial analysis at policy-relevant geographic units (e.g., vital registration statistics, including births and maternal deaths); limited technical expertise, institutional capacity and human resource availability to conduct analyses within low and middle income countries; and, a lack of community participation and participatory approaches including key MNH workers such as community health workers and nurses.

1.1.1 Global initiatives in maternal and newborn health

The UN Millennium Summit in 2000 set the global stage for monitoring and reducing maternal and child mortality, with specific targets to reduce under-5 mortality by two-thirds and maternal mortality by three-quarters by the year 2015 (United Nations, 2015a). The Countdown to 2015 for Maternal, Newborn, and Child Survival was established shortly thereafter in 2005 to ensure accountability for achieving these targets among the 75 most vulnerable countries (the 'Countdown countries') accounting for over 95% of global maternal and child mortality (UNICEF and WHO, 2015). Towards this, the Countdown initiative monitored progress among key maternal, newborn and child mortality interventions within and between countries, identifying key gaps in both intervention coverage and data since 1990. Specifically, priorities of the initiative as outlined by Bhutta and colleagues included:

1) selection and monitoring of evidence-based key interventions proven to reduce maternal, newborn and child mortality;

2) production of key country profiles to support policy guidance and decision making;

- 3) tracking progress towards MDGs 4 and 5;
- 4) recommendations on action to promote equitable coverage of key interventions;

5) ensuring accountability from both national governments and the international development community to ensure MDG targets are met; and,

6) identifying gaps in knowledge and implementation which are hindering progress towards meeting these goals (Bhutta et al., 2010).

As the MDGs reached their conclusion in 2015, Victora and colleagues assessed global progress among these 75 Countdown countries to take stock of progress and lessons learned, as well as

identify next steps (Victora et al., 2016). Overall, they found that the global maternal mortality ratio decreased by 45% over the span of two decades from 380 maternal deaths per 100,000 live births to 210, with the most progress achieved between 2000 and 2013 (United Nations, 2015a). Globally, Southern and Eastern Asia saw the greatest progress in reducing maternal mortality, down 64% and 65%, respectively. Despite this accelerated progress, many countries failed to achieve the MDG 5 target of reduction in maternal mortality ratio by three quarters. These failures were largely a result of causes which could have been prevented through basic and essential intervention packages as outlined by the Partnership for Maternal, Newborn & Child Health (PMNCH), ensuring adequate provision of services throughout pregnancy, birth and the postpartum period (PMNCH, 2011; Say et al., 2014a). Indeed, among these 75 countries, Victora and colleagues note only four countries met targets for both MDGs 4 and 5 (Cambodia, Eritrea, Nepal, and Rwanda).

Although only a small minority of countries met the specified targets outlined by MDGs 4 and 5, the Countdown initiative found that coverage increased for most interventions that are known to combat maternal and child deaths, such as skilled birth attendance, antenatal and postnatal care, and vaccination coverage amongst both women and children. They further found that coverage gaps between the richest and poorest quintiles in 47 countries narrowed among eight key interventions outlined by the WHO in reducing maternal and child deaths (UNICEF and WHO, 2015; Victora et al., 2016). In addition to coverage, financing for maternal, child and newborn health increased and many countries adopted more MNCH friendly policies. Yet regardless of this progress, many disparities (largely socioeconomic) continued to persist in nearly every Countdown country, both in terms of intervention coverage and resources, as well as financing and policies. Further, coverage of these interventions varied substantially between countries in addition to within country variation (UNICEF and WHO, 2015). Finally, the Countdown initiative also noted problems faced by data availability, and the research community's subsequent overall dependence on modelled mortality estimates. They therefore recommended development of innovative measurement techniques as the world moved into the SDG era, with particular emphasis on equity and sub-national monitoring of social and environmental determinants and their impact on health (Victora et al., 2016).

Moving forward, the UN has defined a global MMR target of less than 70 maternal deaths per 100,000 live births by the year 2030. Contention exists, however, as to what global agenda over the coming decades is best suited to not only reach this target, but also ensure equitable progress is seen among the most vulnerable populations (Boldosser-Boesch et al., 2017). Towards this, the UN Secretary-General launched a Global Strategy for ensuring the health and rights of women, children and adolescents, with key connections to the Every Woman Every Child initiative launched in mid-2015. Together, these frameworks aimed to outline a clear roadmap to ending preventable maternal deaths, newborn and child deaths, while promoting human rights and equitable progress (United

Nations, 2016). To accomplish this, the Ending Preventable Maternal Mortality (EPMM) and Every Newborn Action Plant (ENAP) working groups through the World Health Organization (WHO) were created with a grounding in human rights principles. These groups aim to end preventable maternal and neonatal deaths through elimination of disparities in access to and utilisation of care across the reproductive life course (WHO, 2014; WHO and HRP, 2015). In addition to SDG targets of less than 70 maternal deaths per 100,000 live births, EPMM further outlined that no country should have an MMR greater than 140, while ENAP set targets of less than 10 newborn deaths per 1,000 live births by the year 2035.

1.1.2 Maternal and newborn health in sub-Saharan Africa

With almost two-thirds of Countdown countries falling in the African continent, sub-Saharan Africa (SSA) saw a nearly 50% reduction in maternal mortality since 1990 (United Nations, 2015a). Despite this progress, this region accounts for the highest regional maternal mortality ratio in the world, at just over 500 maternal deaths per 100,000 live births in 2013, compared to a global rate of 210 deaths. Further, only four countries achieved the 75% reduction set out by MDG 5 (United Nations, 2015a). Together with Southern Asia, maternal deaths in SSA account for 86% of global maternal deaths, with over half due to causes such as haemorrhage (25%), hypertensive disorders (16%) and sepsis (10%) (Say et al., 2014a). By increasing access and equitable coverage of quality care including antenatal and postnatal care and skilled birth attendance, these deaths could be prevented (UNICEF and WHO, 2015; United Nations, 2015a).

Encouragingly, the global proportion of births attended by skilled personnel increased, and in many parts of the world such as Central and Eastern Asia and the Americas the proportion of deliveries attended by skilled birth attendants in 2013 reached well over 90% (Say et al., 2014a; UNICEF and WHO, 2015). In SSA, however, the increase in proportions of births attended by skilled personnel was more modest and still represents some of the lowest proportions globally, with nearly half of babies delivered without skilled care (United Nations, 2015a, 2015b). Further, the coverage of skilled attendance at delivery was inequitably distributed across regions, with the greatest inequalities observed in Central Africa at only 32% skilled attendance in rural areas as compared to 84% in urban (United Nations, 2015a). Similar to skilled attendance at delivery, only half of pregnant women receive the recommended number of four antenatal care visits, and SSA represented the least amount of change regionally with only 2% increase between 1990 and 2014 (United Nations, 2015a). These gaps suggest that even though global progress has been made in reducing maternal mortality, these advances may mask sub-national and regional health disparities. These disparities are driven by inequalities in intervention coverage which continue to persist, and are disproportionately impacting vulnerable women who need these services the most (United Nations, 2015a).

1.2 Study objectives

As the development world moves away from the MDGs into the SDG era, disaggregated spatial and temporal data are needed along which to monitor change and measure progress the development world is to ensure that new global targets are achieved by the year 2030. This is particularly relevant in resource-poor countries with disproportionately high burdens of maternal and child deaths, such as countries falling within the SSA region. The objective of this thesis is to address this need within several African countries by generating high-resolution data that can be used to monitor spatial and temporal change, with specific aims to:

1) Produce estimates of the probability of utilising MNH services as predicted by geographic accessibility at both high spatial resolution and policy relevant scales, highlighting spatial heterogeneity and sub-national inequalities;

2) Track how spatial inequalities in utilisation of MNH services have changed over time, identifying sub-national regions which have made progress in narrowing inequalities over time, as well as regions where progress still needs to be made; and,

3) Quantify the trade-off between increasing spatial resolution of modelled estimates of delivery via caesarean section in Tanzania, and associated uncertainty in model outcomes.

This work follows a "three-paper PhD" structure and is therefore be organized to address the above aims, with chapters 4 through 6 following traditional journal article format. Chapter 1 includes a broad introduction to the global and regional epidemiology of maternal and newborn health, including overall study objectives of this thesis. Chapter 2 consists of a substantive literature review outlining the fields of maternal and newborn health and spatial demographic data for health, while Chapter 3 outlines the study framework, including study indicators, study area, research questions and study methods used. Chapter 3 concludes with how this work contributes to the current literature as well as intellectual contribution for each analytic chapter, and finally how this work generates broader impact. Chapter 4 consists of an expanded version of the article, "Equality in Maternal and Newborn Health: Modelling Geographic Disparities in Utilisation of Care in Five East African Countries", as published in PLoS ONE's special collection, Neglected Populations: Decreasing Inequalities & Improving Measurement in Maternal Health (C. W. Ruktanonchai et al., 2016). Chapter 5 explores the temporal evolution of spatial inequalities in MNH services within the East Africa Community, "Temporal trends in spatial inequalities of maternal and newborn outcomes among four East African countries, 1999 – 2015", as published in BMC Public Health (C. W. Ruktanonchai et al., 2018). Chapter 6 explores the trade-off between increasing spatial resolution in model inputs and associated model outcomes and measures of uncertainty among delivery via caesarean section in Tanzania, titled "Estimating uncertainty in geospatial modelling at multiple spatial resolutions: the pattern of delivery via caesarean section in Tanzania", as accepted

Chapter 1

to BMJ Global Health. Finally, Chapter 7 includes a synthesis of substantive and methodological findings and broad conclusions from the preceding chapters, with aims of outlining academic and policy impact generated from this work as well as broad policy recommendations, where appropriate.

Chapter 2: Literature Review

To address the objectives of this thesis, a detailed understanding of how, why and where maternal and newborn deaths and complications are occurring is needed, as well as what health services can prevent these deaths. To answer these questions within a geographic framework, this chapter first explores causes of maternal and newborn deaths, what health services are recommended to encourage healthy pregnancy and childbirth, and barriers to obtaining these services within predominantly resource-poor settings. Next, a synthesis of the current framework for monitoring progress of maternal and newborn health within the SDG era is outlined, followed by how, where and to whom spatial and temporal health inequalities occur, as well as previous methods for measuring inequalities in space and time. The role of spatial demographic data in improving maternal and newborn health is then discussed, with a synthesis of existing spatial data within the study region, as well as where knowledge gaps still exist.

The purpose of this chapter is two-fold: firstly, to synthesize the extant literature on the above topics to identify knowledge gaps; and secondly, to provide a theoretical foundation for the remaining chapters of this work (Paré and Kitsiou, 2017). This chapter therefore represents a qualitative narrative review of the existing literature (Sylvester et al., 2013), and while no structured or standardised approach was employed to exhaustively search the literature, a 'snowball' methodology was generally employed wherein relevant citations and additional references were identified within pertinent pieces of work, such as review articles and meta-analyses (Wohlin, 2014).

2.1 Maternal and newborn health

Pregnancy and childbirth should be a time of joy and empowerment for all women, but in many countries across the world, this period can pose critical health risks and complications which too often prove fatal for either the woman, newborn, or both. The health of a woman and her newborn child are inextricably linked, and addressing the health concerns of one inevitably impacts the other. The complications arising out of pregnancy, labour and childbirth, however, vary for woman and child, and resulting fatalities can be caused by vastly different reasons. This section therefore explores the causes of deaths for women and newborns separately, followed with what services are recommended for the health of women and children, plus common barriers associated with obtaining those services on both the supply and demand side. This section concludes with a synthesis of progress in monitoring and evaluation in the field of maternal and newborn health, and outlines the current state of inequalities in maternal and newborn health across the globe.

2.1.1 Causes of maternal death

To achieve global development targets outlined by the SDGs, a clear and nuanced understanding of the causes of maternal and newborn deaths worldwide is needed. The WHO has defined maternal deaths as deaths occurring among pregnant women, or within 42 days of termination of pregnancy (either via delivery or other means), attributed to any cause related to pregnancy, delivery, or its management, excluding incidental or accidental causes (WHO et al., 2015). This definition allows for identification of both direct and indirect causes of maternal deaths, provided routine health records are captured. In reality, however, such routine data is rarely captured amongst the most vulnerable women, particularly in countries with the highest rates of maternal mortality (Say et al., 2014a). Therefore, systematic analyses have been necessary over the decades to adequately report and monitor both direct and indirect causes of maternal mortality.

Analysing 34 datasets representing over 35,000 maternal deaths across various geographic regions throughout the world, Khan et al. describe primary causes of maternal deaths (Khan et al., 2006). Globally, they attributed the leading causes of deaths to haemorrhage and hypertensive disorders, yet they found leading causes of death varied across regions. While progress has been made in reducing the overall number of maternal deaths since 1990, Requejo and Bhutta note that more than half of women who die from complications due to pregnancy are dying from severe bleeding, hypertension and sepsis (Figure 2.1), which can be alleviated through skilled birth attendance and timely postnatal care check-ups (Requejo and Bhutta, 2015).



Figure 2.1 Global causes of maternal deaths in 1990 and 2013, as published by Requejo and Bhutta, 2015

Khan et al. found these trends also occurred regionally—within Africa and Asia, the leading cause of death was haemorrhage, accounting for 33.9% of deaths in Africa (CI 13.3% - 43.6%) and 30.8% of deaths in Asia (CI 5.9% - 48.5%). Alternatively, within Latin American and Caribbean countries, hypertensive disorders accounted for over a quarter of deaths (25.7%, CI 7.9% - 52.4%). Lastly, deaths due to abortions were noted to be particularly high in some Latin American countries, while deaths due to sepsis were higher among study countries as compared to developed countries (Figure 2.2) (Khan et al., 2006).



Figure 2.2 Distribution of causes of maternal deaths by WHO geographical region, as published by Khan et al., 2006

Following similar methodology as Khan et al., Say and colleagues followed up with another systematic analysis of maternal deaths in 2014. Using 417 datasets comprising 115 countries, they examined nearly 61,000 maternal deaths between 2003 and 2009 (Say et al., 2014a). Globally, they similarly found that haemorrhage, hypertensive disorders and sepsis were the leading causes of maternal deaths, accounting for more than half of maternal deaths worldwide. They also noted that indirect causes of deaths, such as medical disorders and HIV-related maternal deaths, could be attributed to nearly a quarter of deaths worldwide (Say et al., 2014a).

The majority of complications discussed above arise during labour, delivery and the immediate post-partum period—however, the majority of deaths occurring from these complications can be

prevented, provided immediate and appropriate medical care (Alkema et al., 2016a). Why, then, do these deaths continue to occur, despite progress in reducing mortality over the previous decades? Thaddeus and Maine proposed a seminal framework to contextualize these maternal deaths in low and middle-income countries, and explain why preventable fatalities continue to occur (Thaddeus and Maine, 1994). Specifically, they focus on the critical period of risk for a woman between the onset of labour and delivery, proposing that adequate treatment or care provided within this interval will usually result in a positive outcome, whether a complication has arisen or not. Many deaths are therefore preventable through provision of treatment, but often the decision to seek or obtain this care can be fraught with life-threatening delays. They subdivide these barriers into a three-delay framework: 1) the decision to seek care, 2) arrival at a care facility, and 3) the provision of care after arrival at a care facility. At each stage of these decisions, barriers and delays may be experienced, resulting in continuation of life-threatening complications or death. Maternal mortality prevention programs must therefore contextualize each of these delays within any intervention effort, in order to effectively treat labour and delivery complications and ultimately reduce preventable maternal deaths (Thaddeus and Maine, 1994).

2.1.2 Causes of newborn deaths

The MDGs established in 2000 aimed to reduce not only maternal mortality by three-quarters, but also under-five child mortality by two-thirds. Among children dying under the age of 5 years old, the largest proportion of deaths have historically been comprised of neonates, or those between 0 -27 days old (Black et al., 2010; Liu et al., 2015, 2012; Rajaratnam et al., 2010). Globally, neonatal deaths declined 2.1% per year and were nearly halved between 1970 through 2010, with similar patterns of accelerating reductions post-2000 (as compared to 1990 – 2000). The most progress was seen in the Western Pacific and Europe, while the least progress seen in Africa and the Eastern Mediterranean (Liu et al., 2012). Yet despite global progress in reducing child mortality, 3.1 million neonatal deaths continued to occur in 2010, comprising over 40% of global under-five mortality (Rajaratnam et al., 2010). Requejo and Bhutta also note that the number of deaths occurring within the first 4 weeks of life are in fact a growing proportion of under-5 mortality (Figure 2.3), and that progress in reducing neonatal mortality specifically, is lagging globally (Requejo and Bhutta, 2015). These findings echoed results published by Liu and colleagues, showing that global trends in the reduction of neonatal deaths have remained relatively consistent since 2000, especially as compared to other causes of under-5 mortality such as diarrhoeal diseases and measles, as shown in Figure 2.4 (Liu et al., 2012).



Figure 2.3 Global causes of under-5 mortality in 2000 and 2013, as published by Requejo and



Figure 2.4 Global trends in under-5 childhood mortality in 2000 - 2010, as published by Liu et al., 2012

The systematic analysis conducted by Liu et al. in 2012 explored global, regional and national causes of deaths among under-fives, and found in 2010 that the leading causes of neonatal deaths included preterm birth and intrapartum-related complications, sepsis or meningitis, and pneumonia (Figure 2.5) (Liu et al., 2012). These findings were echoed by Black and colleagues, who found leading causes of neonatal deaths were attributed to preterm birth complications, birth asphyxia and sepsis in 2008 (Black et al., 2010). Further, neonatal deaths comprised high proportions of underfive mortality (ranging from 48% to 54%) in the Americas, Southeast Asia, the Western Pacific and Europe, with leading causes of death attributed to preterm birth complications and birth asphyxia. Lastly, Africa and Southeast Asia saw the greatest number of total under-five deaths at 4.199 million and 2.390 million deaths, respectively (Black et al., 2010).



Figure 2.5 Global causes of under-5 mortality, separated into neonatal deaths (0 - 27 days) and deaths among children aged 1-59 months, as published by Liu et al., 2014

Performing a follow-up systematic analysis five years later, Liu and colleagues later found similar results globally among leading causes of death among neonates, with preterm birth and intrapartum-related complications representing the leading causes, followed by sepsis and congenital abnormalities. Notably, neonatal tetanus, diarrhoea and pneumonia represented substantial yearly reductions from 2000 – 2013, at 8.9%, 6.0% and 5.2% annual reductions, respectively. The majority of reductions in under-five deaths have, however, been largely isolated to deaths among children aged 1-59 months, with little decline seen among those in the first 27 days of life (Figure 2.4). Lastly, they outline that if present trends continued, sub-Saharan Africa will account for a disproportionately high rate of under-five deaths as compared to births,

estimating it will account for 33% of global births in 2030, yet 60% of under-five deaths (Liu et al., 2015).

2.1.3 Maternal and newborn health services

Preventing the deaths of women and newborns ultimately requires delivery of timely and high quality service packages and interventions across the continuum of care (Kerber et al., 2007). Many definitions of 'continuum of care' exist, but the WHO notes that 'continuum of care' is two-pronged, occurring over both time and space. This firstly means care is provided throughout the life course, spanning the neonatal period to adolescence to reproductive years through pregnancy, and secondly means that care is provided seamlessly, spanning "the home, the community, the health centre and the hospital" (WHO, 2005). Kerber et al. build upon this definition, arguing that prevention of lives is dependent upon functional linkages of services and health care systems across the reproductive life course, resulting in care that "contributes to the effectiveness of all the linked packages" (Kerber et al., 2007).

Countries with the highest levels of maternal and neonatal deaths often also tend to be limited in resources, such as health personnel, funds and supplies. Given these limitations, studies have suggested that essential health intervention packages exist which could avert up to 67% of neonatal deaths within the most vulnerable countries worldwide (Kerber et al., 2007). Kerber and colleagues outlined eight such essential packages across the continuum of care, grouped into services provided through clinical care, outpatient and outreach services, and family and community care. Within these provisional groupings, services were further categorised along the continuum of care, as defined in Figure 2.6.



Figure 2.6 The reproductive life course across which care packages must be integrated, adapted from Kerber et al., 2007

Across the spectrum of pregnancy, childbirth, and the immediate postnatal period, the most lifethreatening conditions can occur suddenly and without warning, and require timely and high quality management. Victora et al. identified critical services across the continuum of care which are crucial in determining the survival of women, newborns, or both (Figure 2.7) (Victora et al., 2016).





Figure 2.7 Median coverage of interventions among Countdown countries, across the continuum of care, as published by Victora et al, 2016

Nyamtema and colleagues further performed a systematic review of health interventions which were most likely to have positive impacts on maternal health outcomes, specifically within limited resource settings (Nyamtema et al., 2011). Out of nearly 60 publications, they found no single intervention alone was sufficient in reducing maternal mortality, reiterating integration of multiple service packages were most effective in influencing health outcomes. Generally, however, they found that the most impactful interventions focused on increasing functional access and delivery of skilled birth attendance with sufficient equipment, such as supplies, drugs and referrals to higher-level health care systems in the event of complications. Other services key to ensuring healthy delivery for both the woman and child also included four or more antenatal care visits (focused on malaria prevention, provision of insecticide-treated bed nets, and other preventative treatments), and postnatal care for both woman and newborn in the hours after delivery.

Even with high coverage of essential interventions such as skilled birth attendance and adequate antenatal care, maternal mortality may not be substantially reduced without equally high coverage of comprehensive emergency care (Souza et al., 2013). Globally, nearly 75% of maternal deaths were caused by obstetric complications between 2003 and 2009, with haemorrhage and hypertension as the leading causes of death (Say et al., 2014a). Access to readily available emergency obstetric care (EmOC) services where life-threatening obstetric complications can be quickly identified and readily treated are therefore key in preventing maternal deaths (Paxton et al., 2005). To combat these preventable deaths, UNICEF, WHO, and the UNFPA, outlined a baseline package of services that health facilities should offer, aimed at treating the 7 major direct causes of

Chapter 2

obstetric complications leading to death, collectively known as emergency obstetric care (EmOC) (UNICEF et al., 1997).

To quantify the availability of these medical interventions and to measure progress in reducing maternal and perinatal deaths, UNICEF, WHO, the UNFPA, and the AMDD (Averting Maternal Death and Disability) initiative through Columbia University identified a set of nine 'signal functions' and common facility indicators (WHO et al., 2009). Using this framework, facilities are classified as either Basic EmOC facilities, or facilities offering 1) parenteral antibiotics, 2) uterotonic drugs such as parenteral oxytocin, 3) parenteral anticonvulsants for pre-eclampsia and eclampsia, 4) manual removal of placenta, 5) removal of retained products, 6) assisted vaginal delivery, and 7) basic neonatal resuscitation. Facilities offering these services plus 8) surgery such as caesarean section and 9) blood transfusion, are classified as comprehensive EmOC facilities (Paxton et al., 2005; WHO et al., 2009).

To assess the impact of EmOC interventions on preventing maternal deaths in developing countries, Paxton and colleagues performed a systematic literature review including studies with evidence from over 80 countries. Reviewing studies with a variety of experimental designs such as quasi-experimental, observational and ecological designs, they found strong consensus that incorporating EmOC facility coverage into national programme strategies is a key component in reducing maternal mortality among developing countries (Paxton et al., 2005). Specifically, among a sample of 12 countries with varying degrees of met need for emergency obstetric services, a preliminary correlation analysis showed that estimated maternal mortality ratio (MMR), or maternal deaths per 100,000 live births, was inversely associated with a higher degree of met need for EmOC ($\mathbb{R}^2 = 0.64$). These findings indicate that countries having higher proportions of women who receive the emergency care they required (and therefore wider EmOC facility coverage), tended to also have lower estimated maternal mortality ratios (Paxton et al., 2005).

In 2006, Paxton and colleagues followed this systematic review up by examining standardized needs assessment created through the AMDD program at Columbia University throughout 24 countries, with aims of reporting global trends of EmOC facility coverage (Paxton et al., 2006). Out of this analysis, they report three key themes which emerged: 1) the minimum number of comprehensive EmOC facilities required in relation to population size were typically available among even the least-developed countries; 2) the number of basic EmOC facilities, however, were typically not sufficient, even among countries with only moderate levels of maternal mortality; and, 3) the majority of facilities surveyed were not able to satisfy the full spectrum of recommended interventions comprising key signal functions. As a result of this analysis, Paxton and colleagues therefore recommended that existing facilities such as hospitals and maternity care wards be 'upgraded' to qualify as meeting basic EmOC facility standards (Paxton et al., 2006).

Chapter 2

2.1.4 Barriers to utilisation of health services

Care packages focusing only on increasing the absolute number of MNH care services may miss a crucial component of why emergency complications can turn into fatalities. Integration of successful intervention packages requires a detailed understanding of the socioeconomic and health systems barriers surrounding the decision to seek and obtain care on both the supply and demand side. Here, the seminal three-delay model is discussed as proposed by Thaddeus and Maine, followed by a discussion of common economic supply and demand-side barriers influencing affordability, acceptability, and quality of care provided.

2.1.4.1 Three-delay framework

The vast majority of obstetric complications can be prevented through routine health services, such as antenatal care or skilled birth attendance, or treated through timely identification and prevention (Bailey et al., 2006). Despite this, the duration between a woman's decision to seek care to that of actually obtaining the care is fraught with delays, thereby preventing life-saving treatment and contributing to morbidity and mortality. Thaddeus and Maine articulated this process when they proposed the three-delay model framework, suggesting phases of delays exist in 1) the decision to seek care, 2) the accessibility of a health facility, and 3) the provision of adequate care (Thaddeus and Maine, 1994). Within the first phase, the decision to seek care is shaped by broad socioeconomic and cultural compositional factors, including demand-side factors and decision-making behaviours outside of the woman's control, such as those by her husband or mother-in-law. For example, Byrne and colleagues found among a synthesis of studies that delayed care-seeking behaviours in the mountainous regions of Nepal was determined by costs of care, and low autonomy of a woman in obtaining her care (Byrne et al., 2013). Importantly, these delays in care-seeking behaviour only served to further exacerbate later delays.

Once the decision to seek care has been made, challenges in accessing care may be encountered through the physical landscape such as mountainous terrain, transportation, or long distances to the nearest health facility. Gething and colleagues explored the accessibility delay within an explicitly geographical context in Ghana, finding that over a third of women lived beyond a 2-hour journey to the nearest facility offering basic or partial emergency obstetric care (EmOC) (Gething et al., 2012). These findings were further exaggerated amongst the most remote regions, with upwards of 80% of women living within "accessible" distance, or less than a 2-hour journey, to a comprehensive EmOC facility.

Gabrysch and Campell further explored how physical accessibility affects service utilisation, specifically in the context of the three delay model, as shown in Figure 2.7 (Gabrysch and Campbell, 2009a). In their literature review, they found the vast majority of studies explore region
and place of residence, while fewer studies have explored distance or travel time and transport to facilities, despite its impact on both the decision to obtain care and the accessibility of that care. The incorporation of geospatial data sources with explicitly geographical approaches are arguably crucial in supporting systemic maternal care planning, yet remain underutilised (Ebener et al., 2015; Gabrysch and Campbell, 2009a; Gething et al., 2012). With the growing availability of georeferenced data, however, opportunities are arising to link these data sources with maternal health outcomes, and represent a promising new avenue of research (Ebener et al., 2015; Gabrysch and Campbell, 2009a).



Figure 2.8 Phases of delay and factors affecting utilisation of maternal health services, as published by Gabrysch and Campbell, 2009

The final phase of delay, provision of adequate care, is predominantly driven by quality of care provided. Thaddeus and Maine argue this is explained by satisfaction or dissatisfaction with the service received and outcome, noting that these mechanisms are driven by perceived quality of care, rather than medical quality, potentially conflicting with patient perceptions (Thaddeus and Maine, 1994). This conflict is echoed by Gabrysch and Campbell, who note that studies found medical culture often clashed with traditional beliefs, resulting in a lack of respect or cultural competence which impacted perceived quality of care (Gabrysch and Campbell, 2009a). Despite its importance in driving the decision to seek care and the timeliness in ultimately receiving that care, few quantitative studies have examined quality of care, and even fewer within a geographical construct. Gabrysch and Campbell argue this could be due to a number of restraints between linking individual level data with facility-based data. Yet with the growing of availability of

georeferenced facilities, household and individual surveys, and spatially explicit Health Management Information systems (HMIS), future research may be able to examine quality of care and its interaction with physical accessibility and facility choice.

2.1.4.2 Supply and demand side barriers

A plethora of economic barriers in accessing care on both the supply and demand side influence an individual's decision to seek, access, and obtain care. Ensor and Cooper outlined this clearly in their review of demand-side barriers in low- and middle-income countries, and present suggestions for how to overcome these barriers (Ensor and Cooper, 2004). They begin by defining 'demand' side barriers as those "factors that influence demand and operate at the individual, household or community level". These may include factors such as lack of knowledge of obstetric complications or available services, high financial costs in accessing care, lack of empowerment in making decisions about seeking care, religious objections, etc.

Alternatively, 'supply' side barriers are those "that influence the slope and position of the supply curve", and may manifest as lack of adequate medicine, equipment or personnel, a lack of nearby health facilities, difficulty in obtaining a referral to a given hospital, or a lack of respect or cultural competence among health staff. Ensor and Cooper go on to further categorize these demand side barriers, suggesting barriers can be broadly grouped into: 1) information on health care choices/providers, 2) education, 3) indirect consumer costs such as distance and opportunity costs, 4) household preferences, 5) community and cultural preferences, attitudes and norms, and 6) price and availability of substitute services. By grouping barriers into these categories, they argue interventions may be targeted to overcome these obstacles, including improved education, provision of transportation to health services, financial restitution for lost wages in accessing care, and targeted subsidies for the most vulnerable members of households (Ensor and Cooper, 2004).

Jacobs et al. further explore both supply and demand side barriers in their literature review of barriers in accessing care, and incorporate dimensionality of barriers as they relate to geographic accessibility, availability, affordability and acceptability (Jacobs et al., 2012). They build upon Ensor and Cooper's work on demand side barriers, and further outline several supply side barriers within these dimensions: service locations affecting geographic accessibility; waiting times, staff qualifications and stock of drugs or other equipment affecting availability; costs and prices of services affecting affordability; and staff interpersonal skills and cultural attitudes, including trust, affecting acceptability.

They further note potential interventions targeting these crucial supply side barriers, including provision of essential health services, regulatory approaches, integrated outreach services, provision of maternity waiting homes, emergency transport and adequate facility staffing, and increased education on culturally sensitive health care delivery (Ensor and Cooper, 2004; Jacobs et al., 2012). They note that interventions exist addressing all four dimensions of access barriers but

do not address both supply and demand side barriers, while other less comprehensive interventions may touch upon only a few dimensions influencing both supply and demand side barriers. As a result, no one intervention can be considered a 'magic bullet' approach, and often successful approaches to reducing these barriers rely on a combination of interventions (Jacobs et al., 2012; McNamee et al., 2009).

2.1.4.3 Quality of care barriers

The existence, accessibility and provision of care does not, however, guarantee that women will use these services, nor does it guarantee improved health outcomes for the woman or child. Hulton and colleagues argue that only provision of high quality, acceptable care can necessitate both timely use and improved maternal and newborn mortality metrics (Hulton et al., 2000). Indeed, among the most marginalised groups within the least developed countries, numerous studies have reported the correlation between low quality of care measures and high rates of maternal and newborn mortality (van den Broek and Graham, 2009). While the specific mechanisms which drive the relationship between health seeking behaviour and quality of care are not well understood, studies have noted that a woman's perception of quality of care can often predict facility choice more than proximity (Hulton et al., 2000). Hulton and colleagues argue these perceptions are shaped by both the experience of care and the provision of care. They go on to report case studies where too often women are provided with inadequate information about what services are being provided to them, are not treated with dignity or respect, such as a lack of privacy during examination, or are sometimes ignored altogether and left to labour in crowded waiting rooms without acknowledgement (Hulton et al., 2007).

Yet despite its importance in predicting health seeking behaviour and in influencing utilisation of maternity services, quality of care within maternal health has been historically neglected, with no universal definition of "quality of care" nor systematic frameworks by which to assess quality of care in maternity services (Hulton et al., 2000; van den Broek and Graham, 2009). To address this need, Hulton and colleagues proposed a framework by which to assess quality of care, incorporating key elements influencing the experience of care and the provision of care (Figure 2.8) (Hulton et al., 2000). They argue that experience of care is influenced by: a woman's perception of the human and physical resources available to her at a health facility; her awareness and understanding of the situation; the perceived respect, dignity and equity afforded to her by staff; and the emotional support received during care.

On the other side of 'quality of care' is a facility's provision of care, which can be reflected by: the objective quality of human and physical resources available; the quality of the referral systems and health management systems in country; the use of adequate and appropriate technology in maternity care; and lastly, the use of and adherence to internationally recognised good practice procedures (Hulton et al., 2007). With the proposal and application of this framework, 'quality of

care' can be objectively assessed and monitored, and indeed represents a neglected agenda which has historically failed to be incorporated into key global strategies and intervention efforts (van den Broek and Graham, 2009).



Figure 2.9 Framework for assessing quality of care in maternity services, as published by Hulton et al., 2007

2.1.5 Monitoring maternal and newborn health outcomes in the SDG era

Establishing a standardized and rigorous framework by which to monitor and evaluate progress in maternal and newborn health will be key to achieving meaningful reductions in maternal and newborn deaths, as well as ensuring greater well-being for all. Since the development of the MDGs in 2000, international agencies around the world have agreed on a short list of key indicators such as maternal mortality ratio, antenatal care coverage, births attended by skilled health personnel, contraceptive prevalence, etc., to be used in monitoring the progress of global reproductive health (WHO, 2006). The WHO outlined a subset of these indicators, together with indicators relating to newborn and child health, to be used as key targets in MDGs 4 and 5, which have continued into the SDG platform under target 3 (Table 2.1). Despite these guidelines, debate surrounding how to measure these key indicators persists, and definitions used among indicators, such as what constitutes 'skilled health personnel', vary widely between countries (Adegoke et al., 2011; Victora et al., 2016).

Miller	nnium Development Goal Targets and Inc	licators (2000 – 2015)			
	Goal 4: Reduce child mortali	ty			
	Target	Indicator			
Target 4A	Reduce by two-thirds, between	4.1: Under-five mortality rate			
	1990 and 2015, the under-five mortality rate	4.2: Infant mortality rate			
		4.3: Proportion of 1 year-old children immunized against measles			
	Goal 5: Improve maternal health by 2015				
Target 5A	Reduce by three-quarters, between 1990 and 2015, the	5.1: Maternal mortality ratio (MMR)			
	maternal mortality ratio	5.2: Proportion of births attended by skilled health personnel			
Target 5B	Achieve by 2015 universal access	5.3: Contraceptive prevalence rat			
	to reproductive health	5.4: Adolescent birth rate			
		5.5: Antenatal care coverage (at least one visit and at least four visits)			
		5.6: Unmet need for family planning			
Sustai	inable Development Goal Targets and Inc	licators (2015 – 2030)			
Goal	Goal 3: Ensure healthy lives and promote well-being for all at all ages				
	Target	Indicator			
Target 3.1	By 2030, reduce the global	3.1.1: Maternal mortality ratio			
	than 70 per 1000,000 live births	3.1.2: Proportion of births attended by skilled health personnel			
Target 3.2	By 2030, end preventable deaths	3.2.1: Under-five mortality rate			
	years of age, with all countries aiming to reduce neonatal mortality to at least as low as 12 per 1,000 live births and under-5 mortality to at least as low as 25 per 1,000 live births	3.2.2: Neonatal mortality rate			
Target 3.8	Achieve universal health coverage, including financial risk protection, access to quality essential health-care services and access to safe, effective, quality	3.8.1: Coverage of essential health services (defined as the average coverage of essential services based on tracer interventions that include reproductive, maternal, newborn and child health, infectious			

Table 2.1 Selected MDG and SDG targets and indicators pertaining to maternal and newborn health

	and affordable essential medicines and vaccines for all	diseases, non-communicable diseases and service capacity and access, among the general and the most disadvantaged population)
		3.8.2: Number of people covered by health insurance or a public health system per 1,000 population
Target 3.c	Substantially increase health financing and the recruitment, development, training and retention of the health workforce in developing countries, especially in least developed countries and small island developing States	3.c.1: Health worker density and distribution

Evidence shows that the majority of maternal deaths occur around the time of labour, delivery, and the immediate postpartum period (Ronsmans and Graham, 2006). Therefore, the reduction and prevention of adverse maternal and newborn health outcomes will require a focused understanding of individual risk before, during and after delivery, in addition to a facility-level understanding of accessibility to skilled workforce, emergency services, and the quality of services provided (Kerber et al., 2007; Requejo and Bhutta, 2015). Improving fundamentally multidimensional maternal and newborn health outcomes, however, necessitates targeted efforts and quantifiable indicators of progress. Towards this, monitoring and evaluation frameworks have been applied to identify structural indicators, input process and resulting outcomes which can be used to monitor progress effectively and efficiently (WHO, 2010a, 2016a).

2.1.6 Inequalities in maternal and newborn health

The MDGs prioritised a reduction in national level development indicators such as overall poverty or literacy, with a focus on ensuring rates between countries were more equitably distributed. Little focus, however, was paid to within-country inequalities, and indeed while national progress improved, pockets of localized, hidden disparities containing the most vulnerable populations persisted (Bhutta ZA and Reddy K, 2012). The SDGs agenda aims to address this oversight with Goal 10 specifically calling for reduction in inequalities both between and within countries, stressing "no one left behind" (Hosseinpoor et al., 2015). Disaggregated data comparable across countries will be key to achieving these targets and monitoring the reduction of inequalities both nationally and sub-nationally, and indeed other SDG targets call on countries to increase high-quality, readily accessible data, disaggregated by geographic location and socio-economic indicators such as race, age, and income. The WHO's Health Equity Monitor has led the way on best practices in compiling databases which consist of disaggregated, comparable data spanning

nearly 100 low- and middle-income countries, and reporting how these inequalities should be monitored and evaluated over time and space (Hosseinpoor et al., 2015; WHO, 2015a).

In this seminal report, the WHO's Health Equity Monitor focused on inequalities within the field of reproductive, maternal, newborn and child health (RMNCH), noting that inequalities persisted both between and within countries in terms of maternal health outcomes and intervention coverage (WHO, 2015a). Specifically, they noted the largest disparities were found "between the richest and poorest, the most and least educated, and urban and rural areas", among skilled birth attendance (Figure 2.9) and 4+ antenatal care visits, followed by use of modern contraception (Figure 2.10) and immunization coverage (Figure 2.11). Just as inequalities evolve over space, however, they have also changed and improved over time, with national figures increasing over the previous decade, and some countries noting more rapid rates of improvement among vulnerable sub-populations within country.



O African Region; 🗖 Region of the Americas; × South-East Asia Region; 🛆 European Region; + Eastern Mediterranean Region; 🛇 Western Pacific Region

Coloured shapes indicate countries – each study country is represented on the graph by one coloured shape. Each type of shape represents one WHO region. Dashed orange lines indicate the median values (middle points).

Figure 2.10 Skilled birth attendance in 83 low- and middle-income countries, as published by WHO 2015a

Chapter 2



Circles indicate countries – each study country is represented on the graph by three circles. Horizontal lines indicate the median value (middle point) for each subgroup. Light grey bands indicate the interquartile range (middle 50% of study country estimates). Highlighted country: Ethiopia.





Circles indicate countries – each study country is represented on the graph by five circles. Horizontal lines indicate the median value (middle point) for each subgroup. Light grey bands indicate the interquartile range (middle 50% of study country estimates).

Figure 2.12 DTP3 immunization coverage among 78 low- and middle-income countries, as published by WHO 2015a

For example, in half of the countries included within the WHO report, immunization coverage for polio and DTP3 were 'pro-poor', with children in the poorest quintiles having margins of nearly 10 points greater than children in the wealthiest quintiles, as shown in Figure 2.12 (WHO, 2015a).

However, some gains in intervention coverage over time varied even by socioeconomic status, such as unmet demand for family planning. Specifically, this indicator saw improvements over time by education (with those in the least educated group having a higher rate of increase in met demand, as compared to those with secondary education or higher), but with slower rates of increase among those in rural areas. These findings suggest the complicated future of monitoring disaggregated data within the post-2015 SDG era. This implies that in order to track the progress of maternal and newborn health within countries (and importantly, how this progress manifests within all populations), a multidisciplinary approach is needed, tracking outcomes disaggregated by geography, time, and socioeconomic status.





Coloured shapes indicate countries – each study country is represented on the graph by one coloured shape. Each type of shape represents one WHO region. **Dashed orange lines** indicate the median values (middle points).

Figure 2.13 Change over time (absolute and absolute excess change) in DTP3 immunization coverage among 41 low- and middle-income countries, as published by WHO 2015a

2.1.6.1 Measuring inequalities

Monitoring and evaluating progress towards achieving SDGs will rely not only on individual and facility level indicators, but also coverage of one or many interventions. Towards this, several approaches exist quantifying coverage among a set of interventions, including co-coverage, composite coverage, and effective coverage indices. A co-coverage score can be calculated to reflect what proportion of the population is receiving a set of interventions, typically among a set of essential or lifesaving interventions (Victora et al., 2005). This index can be helpful in identifying to what extent women and children are receiving a set of complimentary interventions, but can also

Chapter 2

prove analytically intensive as it requires reanalysis of input survey data (Barros and Victora, 2013).

Alternatively, the composite coverage index score represents a weighted score among a set of essential interventions, providing information on the overall coverage of a set of intervention services, with emphasis (or weight) of each constituent service varying across the continuum of care as determined by the researcher (Boerma et al., 2008). This score can prove more computationally efficient, particularly across multi-country analyses or time-trend assessments, particularly where a single, group-level estimate is desired (Barros and Victora, 2013). Lastly, an effective coverage score can be calculated, representing health systems' performance score by combining information quality of care, need and use of services of interest. This approach reflects not only the proportion of the population or individuals using the service, but also adjusts for service use by the quality of care of the service provided (Nguhiu et al., 2017).

This approach can be effective in prioritizing country-specific priorities, as well as estimating inequalities among particular sub-groups. The WHO exemplified this approach when calculating a composite coverage index highlighting 8 key RMNCH interventions, including met demand for family planning, ANC coverage, SBA, BCG/DTP3/measles immunization coverage, health seeking behaviour among children with pneumonia symptoms, and oral rehydration therapy for children with diarrhoeal diseases (WHO, 2015a). By combining these interventions into one easily interpretable score, inequalities over space, time and socioeconomic characteristics can be reported, as exemplified in Figure 2.13 where the composite coverage indicator score is presented for several socioeconomic characteristics.



* Education refers to the highest level of schooling attained by the woman or, in the case of newborn and child health interventions, the child's mother. **Circles** indicate countries – each study country is represented on each graph by one circle per subgroup. **Horizontal lines** indicate the median value (middle point) for each subgroup. **Light grey bands** indicate the interquartile range (middle 50% of study country estimates).

Figure 2.14 Composite coverage index for 8 selected RMNCH interventions in selected low- and middle-income countries, as reported by WHO 2015a

2.2 Spatial demographic data for health in Sub-Saharan Africa

With recent advancements of more computationally efficient resources and user-friendly applications, the past decade has seen an exponential rise in applications of geographical information systems (GIS) towards addressing public health research questions (Banerjee, 2016). Adoption and uptake of GIS techniques within public health has generally been slower within Africa, however, due to a number of obstacles such as infrastructure and cost constraints, lack of donor investment involving capacity strengthening and local engagement, and lack of sustainable research funding moving past the donor-led phase of research projects (Tanser and le Sueur, 2002). Despite this, GIS tools offer great potential in addressing some of the most significant public health threats facing the African continent, particularly in the context of MNH (Ebener et al., 2015; Molla et al., 2019, 2017).

Over the coming decades, data visualization and disaggregation through disease mapping and spatial analysis can play an increasingly significant role in ensuring that preventable maternal and newborn deaths are avoided and disease burden is lessened amongst the most vulnerable populations (Molla et al., 2017; Tanser and le Sueur, 2002). Despite this, monitoring MNH trends over time and space remains limited due to unreliable spatial data, lack of expertise and capacity within countries, and inconsistent data production resulting in conflicting results (Molla et al.,

2019). Avoiding preventable deaths therefore necessitates that these challenges are adequately addressed and recommendations made in ensuring high quality spatial data that can be collected even within resource-poor settings.

In this section, I address current trends and emerging innovations in freely available spatial demographic data production that can be used in GIS applications throughout Africa. Firstly, I outline the utilisation of GIS techniques and availability of spatial demographic data in Africa, followed by the evolution of methods and current approaches used to estimate these data, where other reliable sources might be lacking. I then present case studies highlighting applications in producing freely available spatial demographic data, including subnational age and gender composition, high-resolution live births and pregnancies, and quantifying human mobility derived from mobile phone data. Lastly, I discuss gaps in knowledge in improving the production of spatial demographic data, and applications to low-resource settings and policy-making.

2.2.1 GIS utilisation and the availability of spatial demographic data

2.2.1.1 GIS utilisation within Africa

With an increasing number of health researchers adopting GIS techniques, the potential to explore spatial variation in disease and its relationship to the environment, population, and health systems is growing. The application of these methods can be limited, however, by infrastructure and cost constraints, as well as specialist knowledge capacity. While GIS can help to fill these gaps, the technology must be aimed at health problems and priorities relevant to the region. Towards this, (Tanser and le Sueur, 2002) performed a systematic review of the health literature within African settings at the turn of the millennium, examining GIS research amongst top public health threats within Africa, such as HIV/AIDS, malaria, and tuberculosis. Overall, they found that GIS applications have been widely applied within Africa to diseases with strong environmental components, such as malaria elimination settings, but fewer studies have utilized GIS to study non-vector borne diseases such as tuberculosis and HIV/AIDS. Despite the ability to address research questions central to African health issues, they note that the use of GIS techniques remained underutilized, and suggested a need for greater capacity strengthening efforts to train GIS experts within-countries (Tanser and le Sueur, 2002).

Yet with global trends favouring more powerful computational techniques and hardware, greater availability of user-friendly software, and increasing reliability of demographic data collection and vital health registrations, the use of GIS continues to hold promise for addressing health research priorities and policy planning within Africa (Tanser and le Sueur, 2002). More recently, Ebener and colleagues performed a rapid literature review over a decade later to examine utilisation of GIS techniques within the field of maternal health (Ebener et al., 2015). Overall, they note that despite the increasing interest and promise in the use of GIS techniques to address health problems,

methods remained largely underutilized within the field of maternal and newborn health in lowand middle-income countries. Specifically, they report three primary emerging areas of research within the field: 1) thematic mapping (i.e., choropleth maps conveying information on a topic or theme), 2) spatial analysis (i.e., creation of new information using spatial data), and 3) spatial modelling (i.e., statistical or mathematical models simulating a spatial process or phenomena). Despite the emergence of these broad research areas, they recommend better communication and coordination amongst institutions to build GIS capacity and expertise, including amongst incountry health sectors and non-academic institutions (Ebener et al., 2015).

2.2.1.2 Availability of spatial demographic data

However, to facilitate greater application and uptake of GIS methods in addressing health research questions specific to Africa, reliable, high-resolution, and contemporary spatial data is necessary which is also freely available. Over recent decades, the use of spatially explicit demographic data, particularly quantifying human population distributions, has seen broad applications across spatial epidemiology, urban planning, accessibility modelling, disaster management, resource allocation, and more (Tatem et al., 2012, 2007).

Towards this, Tatem and colleagues outline several freely available sources of spatially explicit data (Table 2.2) to inform demographics such as population counts, age and gender composition, urbanization, etc., including census data, census microdata, and survey data. Figure 2.15 displays the availability of these datasets by country, as published by Tatem et al., 2012. This figure represents (a) the number of census microdata records freely available through the International Public Use Microdata Series repository (https://international.ipums.org/international/), (b) the combined number of DHS, AIDS Indicator Surveys (AIS), and Malaria Indicator Surveys (MIS) surveys conducted for each country, and (c) the combined number of DHS, AIS, and MIS surveys available with GPS coordinates available (Tatem et al., 2012). While these sources may not represent a comprehensive list of available datasets, and crucially do not contain information on availability of HMIS data, they nevertheless represent major venues of freely available geospatial demographic data accessible across low and middle-income countries.

Table 2.2 Sources of spatially explicit demographic data, as outlined by Tatem et al., 2012

Data (standard survey name)/source	Time intervals	Typical spatial coverage	Typical strata	Relevant variables
Census	•	1		1
National Statistical Offices	Typically 10 years	Census enumerator area or courser level	Urban/rural, race or ethnic groups (often)	Sex, age, education, migration status, household and dwelling characteristics
Census Microdata				
https://international.ipums.org/international/	Typically 10 years	Admin 1-3	Urban/rural	Household and dwelling characteristics, sex, age, education, migration status, children ever born, children surviving
DHS (Demographic and Health Survey)				
Household, women 15–49, men 15–59, children born in the last five years				
http://www.measuredhs.com/	Varies by country, typically every 5 years	National, Admin 1/region, GPS coordinates of cluster locations for	Urban/rural	Household and dwelling characteristics, sex, age, education, maternal and child health, fertility and full birth history, family planning, domestic

Data (standard survey name)/source	Time intervals	Typical spatial coverage	Typical strata	Relevant variables
		most recent surveys (last 15 years)		violence, biomarkers, nutrition
MICS (Multi-indicator cluster survey)				
http://www.unicef.org/statistics/index_24302.html	UNICEF (Round 2, 1999–2001; round 3 2005–2007; round 4 is in the field 2009– present)	National, Admin 1	Urban/rural	Household and dwelling characteristics, sex, age, education, status, maternal and child health, child labour, domestic violence, summary birth history, anthropometry
LSMS (Living Standard Measure Survey)				
(Integrated Household Budget Survey and many others that are locally adapted)				
http://iresearch.worldbank.org/lsms/lsmssurveyFinder.htm	Irregular	National, Admin 1, some GPS coordinates	Urban/rural	Household and dwelling characteristics, sex, age, education, migration status, consumption, expenditures, income, nutrition, anthropometry, summary birth history

Data (standard survey name)/source	Time intervals	Typical spatial coverage	Typical strata	Relevant variables
MIS (Malaria Indicator Survey)	·		·	·
http://www.measuredhs.com/				
http://www.malariasurveys.org/	Varies by country, typically every 3 years	National, Admin 1/region, GPS coordinates of cluster locations for some surveys (last five years)	Urban/rural	Household and dwelling characteristics, sex, age, education, biomarkers
AIS (AIDS Indicator Survey)				
http://www.measuredhs.com/	Varies by country, typically every 3 years	National, Admin 1/region, GPS coordinates of cluster locations for some surveys (last eight years)	Urban/rural	Household and dwelling characteristics, sex, age, education, biomarkers

Census data represent some of the most detailed information regarding spatial demographic data within a country (Hay et al., 2005), typically carried out around once every 10 years (Tatem et al., 2012), and in more recent decades often include harmonization into digital census databases (Balk et al., 2006). However, while high-income countries have extensive resources available to conduct census surveys, low-income countries can suffer from constraints in resources, lack of knowledgeable experts, and competing development priorities, with the most recent mapping and census efforts for some countries occurring thirty to forty years ago (Tanser and le Sueur, 2002; Tatem et al., 2007). In addition, these data can be aggregated to bespoke and irregular administrative units based on enumeration areas (Balk et al., 2006), and are often proprietary and difficult to obtain. To supplement these challenges as well as fill in temporal gaps, census microdata are comprised of large, representative subsamples of households surveyed during a census generally stored in data repositories like the International Public Use Microdata Series (https://international.ipums.org/international/), making data easier to obtain than full censuses. These data may be available at smaller spatial units than typical surveys, with similar levels of spatial demographic data such as age, gender, population, etc. (Tatem et al., 2012).

Survey data collected through international survey programmes such as the Demographic and Health Surveys (DHS), Multiple Indicator Clustery Survey (MICS), and AIDS Indicator Survey (AIDS) are collected on regular intervals and standardized across countries to allow for comparison across borders. These surveys represent some of the primary sources of information on demographic data such as fertility, health, family planning, maternal and child health, provision of health services, and more (Burgert and Prosnitz, 2014), and further allow links between individual household survey data and health facility data. Not only are these data rich in demographic and health information on local populations, but they are becoming increasingly spatially detailed, as represented by geo-referencing of clusters among DHS surveys (Burgert et al., 2013). Further, these data are freely available to researchers across the world (Figure 2.15), and therefore represent a promising avenue for obtaining spatial demographic data, particularly amongst developing countries.



Figure 2.15. Availability of demographic datasets informing subnational population attributes, among a) census microdata, b) national household survey programmes, and c) national household survey programmes with corresponding GPS information. Adapted from (Tatem et al., 2012)

Lastly, many countries across Africa rely on country-wide health management information systems (HMIS) for routine health data and disease surveillance (Gething et al., 2006). These systems, when regularly updated and uniformly applied, can result in reliable and timely information that

can be used to robustly monitor and evaluate a variety of public health conditions. This is especially pertinent in resource-limited settings, where interventions and funds must be strategically and efficiently allocated (Health Metrics Network, 2008). Over recent years, HMIS systems are increasingly providing geo-located data at the health facility-level, where data is collected and then compiled at the regional and national level. In a highly functioning HMIS setting, this results in large datasets which can be analysed at both high spatial resolutions and across fine temporal scales, such as monthly estimates. Despite this, the reality of HMIS settings within Africa is less than ideal, with many countries facing competing priorities, resulting in poor data coverage and incomplete routine surveillance efforts (Gething et al., 2006; Health Metrics Network, 2008). Regardless, HMIS remain a robust and underutilised source of spatial health data, holding promise for further strengthening over the coming years.

Yet in the absence of contemporary sources of spatial data, in addition to decentralized, uncoordinated efforts that can often lead to conflicting, unreliable data sources, the production of spatial demographic data in resource-poor settings often relies upon augmentation with ancillary, satellite-derived data sources (Molla et al., 2017; Tatem et al., 2007). These data sources may include proximity to roads, elevation and land cover, human settlement locations, distance to nearest urban centre, night-time lights, and more (Dobson et al., 2000; Tatem et al., 2012), and are often employed in geo-statistical models to predict distributions of human populations (Balk et al., 2006), including age and gender composition (Pezzulo et al., 2017), births and pregnancies (Tatem et al., 2014), and development indicators such as literacy and childhood stunting (Bosco et al., 2017).

Mapping efforts utilizing these data sources have grown over the past few decades, evolving from simple areal weight redistribution methods (Tatem et al., 2007) to more sophisticated machine-learning approaches requiring computationally intensive time and resources (Stevens et al., 2015; Wardrop et al., 2018). In the following sections, we discuss the evolution of efforts to map demographic data, as well as current approaches in quantifying human population distributions at a high spatial resolution.

2.2.2 The evolution of mapping demographic data and current approaches

2.2.2.1 The evolution of mapping demographic data

Global populations are rapidly changing, with more than 50% of global population growth predicted to fall within urban settings (Stevens et al., 2015; Tatem and Hay, 2004), and considerable variation in age and gender composition of these populations (Alegana et al., 2015). The dynamics and distribution of these populations have crucial implications for public health applications (Alegana et al., 2015), with human population totals used to generate disease burden estimates at the global, national, and subnational scale (Hay et al., 2005). However, reliable and

contemporary census and settlement data often do not exist, particularly amongst African countries (Tatem and Linard, 2011; Tatem et al., 2007). This can have important ramifications in disease outbreak, as exemplified in the case of the 2013 – 2016 Ebola outbreak in Western Africa, where emergency response was hampered by not knowing the size and locations of population settlements (Varshney et al., 2015). In the absence of such data, high-resolution population estimates can be produced using ancillary data sources available (Molla et al., 2017; Tatem et al., 2007). Historically, efforts to estimate demographic data have employed methods such as areal weighting (Tatem et al., 2007) and land-cover based redistribution (Tatem et al., 2004), while more current approaches include dasymetric redistribution (Stevens et al., 2015) and bottom-up population mapping techniques (Wardrop et al., 2018).

Areal weighting techniques constituted some of the first efforts to quantify human populations on a gridded (or raster) surface, such as the Gridded Population of the World, version 2 (Deichmann et al., 2001). These techniques consist of a simple overlay between a gridded surface, or raster surface, and an administrative polygon containing population information (such as census units), and evenly redistributing population counts to grid cells falling within the polygon of interest (Hay et al., 2005). While conceptually straightforward and easy to calculate within GIS software, these techniques critically assume a uniform distribution of population within an administrative unit (Hay et al., 2005), which is often unrealistic.

The land cover-based redistribution approach builds on the areal weighting technique to generate population estimates using ancillary satellite imagery data to map detailed layers of human settlements (Linard et al., 2012), which are then linked with census estimates to generate accurate and high-resolution population estimates. Tatem and colleagues utilized this approach in the East Africa region, combining satellite imagery data with remotely sensed land cover data to generate reproducible, adaptable estimates of populations across large geographic areas at high spatial resolutions such as 100 meters (Figure 2.16) (Tatem et al., 2007). By combining these freely available data sources such as land cover, elevation, and topography data, they demonstrated that it was possible to generate accurate estimates of human settlements and population estimates at a higher spatial resolution than any previous work. While this approach was widely scalable and flexible across low- and middle-income countries, it relied heavily on input data sources such as land cover, climate, and crucially, high-resolution census data (Tatem et al., 2007). In light of this, this approach has generally been replaced by an alternative, machine-learning algorithm such as the random forest approach (Stevens et al., 2015), which incorporates data from a wider range of ancillary data, including distance to health facilities, schools, road networks, and more.

Chapter 2



Figure 2.16. Population density of the East Africa region (Burundi, Kenya, Rwanda, Uganda, Tanzania), with selected inset cities, using a land-cover based redistribution approach, adapted from (Tatem et al., 2007)

2.2.2.2 Current approaches to mapping demographic data

Stevens et al. outline a random forest mapping technique used to disaggregate census data and generate high-resolution population estimates for Kenya, Vietnam, and Cambodia (Figure 2.17) (Stevens et al., 2015). This approach utilizes a semi-automated modelling approach, incorporating continuous global datasets, such as night-time lights, land cover and topography, as well as discrete covariates, such as distance to nearest urban area. By combining these data with census data, this approach employs dasymetric redistribution, where population counts are 'intelligently' redistributed from a coarser spatial resolution (such as census enumeration areas) to a finer spatial resolution, while preserving the total counts at the original input source (Mennis, 2003). However, instead of redistributing these population counts evenly across the finer spatial area of interest, a regression tree model (or random forest) can be used to 'unequally' distribute counts, or weight their distribution (Figure 2.18) (Stevens et al., 2015). This weighted layer contains information from a range of ancillary sources, as outlined above, and reflects underlying mechanisms contributing to unequal population distributions. This approach is robust, semi-automated and scalable, and shows improvement over more traditional mapping approaches, such as previously outlined land cover-based techniques (Stevens et al., 2015).

Chapter 2



Figure 2.17. High-resolution population counts redistributed using Kenya 1999 census data and ancillary data sources (freely available at worldpop.org.uk). Adapted from (Stevens et al., 2015)

Where census data are unreliable or out of date, "bottom-up" population mapping techniques have been employed, as outlined by Wardrop and colleagues (Wardrop et al., 2018). Similar to the previous land cover-based and dasymetric redistribution techniques as outlined above, the goal of bottom-up population mapping is to generate high-resolution population estimates using a range of robust, and ideally freely available, data sources. In contrast to the above approaches (which can be considered 'top-down' population mapping), bottom-up population mapping relies on micro-census data, which collects information within smaller regions than census enumeration areas, collected at more frequent and rapid intervals (Figure 2.18) (Wardrop et al., 2018). Using statistical models to account for residual spatial autocorrelation, these micro-census population counts are linked to spatial covariates to predict population counts in un-sampled locations. Few studies have validated outputs or accuracy of these methods over other techniques such as top-down approaches, however, suggesting an emerging area for future research. Regardless, bottom-up approaches offer potential for predicting population counties in countries where census data are unreliable or sparse, offering a robust method to predict into un-sampled locations with associated model errors and uncertainty.



Figure 2.18. Top-down (A) versus bottom-up (B) approaches to mapping gridded populations, adapted from (Wardrop et al., 2018)

2.2.2.3 Challenges in spatial data production

Maintaining participant confidentiality, particularly when data contain sensitive health or identifying information, is a crucial consideration for any database. As spatial databases generally contain some sort of locational information, confidentiality is an important challenge, particularly amongst datasets containing highly localized movements such as call data records (CDRs) or human movement patterns (de Montjoye et al., 2013). In some cases, removing patient identifiers such as name, address, and date of birth, is a reasonable step to protecting confidentiality— however, in the case of locational coordinates (such as geo-located DHS clusters) or other spatially identifiable situations (such as CDRs), this may not be an option (Burgert et al., 2013; de Montjoye et al., 2013). Approaches exist, however, to overcome these challenges and effectively anonymize data while maintaining relevant spatial information, such as those used by Ruktanonchai and colleagues to anonymize CDR data (N. W. Ruktanonchai et al., 2016).

Working with the primary mobile phone provider across Namibia (Mobile Telecommunications Limited) representing nearly 85% of the mobile phone market, Ruktanonchai et al. obtained an anonymized dataset consisting of 1.2 million unique SIM cards in Namibia representing over 9 billion communications from October 2010 to September 2011. These data include calls or SMS text messages sent during the study period and the phone towers these events were routed through. By observing the towers used by an individual over time, the researchers were able to infer their

mobility patterns. These data therefore represent sensitive and localized human movement patterns, necessitating strict confidentiality. To facilitate this, MTC removed individual SIM phone numbers and replaced them with random study numbers to be used by researchers, effectively removing any personally identifying information in the dataset. However, research has suggested that even as few as four movement data points can be used to uniquely identify up to 90% of individuals, necessitating further protocols to ensure participant confidentiality (de Montjoye et al., 2013). Towards this, Ruktanonchai et al. averaged the movement of all residents at each tower (n = 402 towers across Namibia), effectively aggregating movement data to the tower catchment level and further ensuring anonymity (N. W. Ruktanonchai et al., 2016).

Lastly, challenges unique to Africa exist in uptake of GIS technology, as outlined by (Tanser and le Sueur, 2002). Firstly, they note a lack of qualified GIS technicians and experts which prevents GIS applications from moving past the donor-initiated phase. Many health research studies in the area are funded through international donor agencies, and are often led by non-African scientists and principal investigators with sparse local engagement and contribution (Molla et al., 2019, 2017; Tanser and le Sueur, 2002). To facilitate sustainable GIS applications to address local health problems, capacity training and support must be prioritized such that local scientists who understand both cultural and health systems factors can prioritize the research agenda (Molla et al., 2019). Secondly, Tanser and le Sueur note a lack of suitable spatial datasets in addressing health and geographical research, which suffer from decentralized and uncoordinated efforts (Molla et al., 2017), in addition to time and cost constraints. They argue that cross-sector projects may help to address these challenges, creating a more unified and systematic approach to creating large spatial databases. They further argue cost-effective methods of creating these datasets include establishing large-scale sentinel demographic and health surveillance datasets. Molla et al. make similar recommendations for spatial data used in addressing maternal and newborn health questions, suggesting universal datasets should be created with common data features such as geo-coding, health facility lists, and more (Molla et al., 2017).

However, in the absence or interim of strengthening sustainable capacity strengthening efforts, it is vital to ensure that high-resolution, contemporary and reliable spatial demographic data being produced are made available to researchers across the globe. Towards this, the WorldPop project (www.worldpop.org) based at the University of Southampton works to ensure that every person is accounted for in the decision making and resource allocation process (Tatem, 2017). Fundamental to this vision is making both spatial demographic data and methods open, freely available, and transparent. The datasets available through the WorldPop project have been used not only by academic researchers across low- and middle-income countries, but also policy makers, statistical agencies, and international development agencies across the world (Howes et al., 2015; C. W. Ruktanonchai et al., 2016; N. W. Ruktanonchai et al., 2016; Sorichetta et al., 2016; Tatem, 2017; C. Edson Utazi et al., 2018b).

2.2.3 Applications in producing spatial demographic data

Despite the growing availability of spatial data, the problem of how to generate data for unsampled locations persists. While national-level household surveys such as DHS surveys are increasingly employing geo-referencing techniques, data are often presented at the national or regional level, and may mask important inequalities at higher spatial resolutions (Bhutta ZA and Reddy K, 2012). The production of spatial demographic datasets has therefore become an increasingly recognized tool to fill in these spatial gaps by generating high-resolution surfaces of environmental covariates, population estimates, health outcomes, and more (Ebener et al., 2015). Here, I highlight case studies outlining the production and application of datasets, demonstrating how these innovative datasets can be used for cost-effective decision making and resource allocation efforts, particularly in the absence of other reliable data sources or while capacity strengthening and training efforts are ongoing.

2.2.3.1 Mapping spatial demographics

Health metrics and intervention planning often aim to target specific populations, such as family planning interventions targeted towards women of childbearing age (Alegana et al., 2015; Hay et al., 2005; A. J. Tatem et al., 2013; Tatem et al., 2014). In addition to knowing where populations are, it is therefore also vital to know who comprises these populations. Historically, however, studies mapping indicators and health risks have used statistics at the national level, and do not account for localized demographic variation and may contribute to hidden inequalities amongst the most vulnerable populations (Gwatkin, 2005; Reidpath et al., 2009; Tatem et al., 2014). Demographics such as age, gender, and estimates of births and pregnancies have important monitoring and surveillance applications, such as malaria surveillance (Alegana et al., 2015), vaccination coverage planning (C. Edson Utazi et al., 2018b), and family planning interventions (Tatem et al., 2014), amongst many others. Capturing subnational variation in these populations is therefore crucial to ensure effective resource allocation.

2.2.3.2 Age and gender composition

To quantify subnational variation in age and gender composition, Tatem et al. constructed a detailed and contemporary population dataset for the African continent incorporating information on subnational age and gender structure (A. J. Tatem et al., 2013). To do this, they compiled and collated estimates for age and gender from a range of data sources representing more than 20,000 subnational administrative units, primarily via census data, household-level microdata collected via census surveys, and census microdata. Census datasets represent the most reliable counts of age and gender, as they contain large sample sizes per small area, but other sources of data can be limited to small sample sizes, such as the case with microcensus or household survey data. In instances where microscensus or household survey data were collected within one year of census

derived estimates, these were collated in order to increase sample size and reduce error. Where contemporary census data were not available, however, national household survey data were used, such as those collected through the DHS, Malaria and AIDS Indicator Surveys, and Multiple Indicator Cluster Surveys.

Proportions of individuals by age and sex were then derived from these counts, which were matched to the corresponding administrative boundary (Figure 2.19a/b). Using this Africa-wide GIS database, population estimates at the 100m resolution as produced by the AfriPop project (Linard et al., 2012) were adjusted to create a gridded dataset of distributions of populations by 5-year age groups and sex (Figure 2.19c). Additionally, these data were projected and adjusted from 2000 to 2015 at 5-year intervals by applying United Nations population growth estimates (A. J. Tatem et al., 2013).

As proof of concept, Tatem et al. further quantified how previously unaccounted for subnational demographic variation can influence important health and policy metrics, such as travel time to healthy facility and malaria transmission. By comparing these metrics generated using national level age structure data against the assembled subnational dataset described above, they found that modelled estimates could be as much as 100% discordant between each other, with as many as half of the study countries demonstrating greater than 10% discrepancies in estimates (A. J. Tatem et al., 2013). With crucial international development targets such as those associated with the Millennium Development Goals and Sustainable Development Goals tied to these estimates, in addition to national-level monitoring and surveillance funds, ensuring the accuracy of modelled estimates by capturing subnational demographic heterogeneity is vital to ensuring cost-effective decision making and policy efforts.



Figure 2.19. Spatial demographic datasets for Africa. A) Proportion of children under 5; B) proportion of women of child-bearing age; and, C) children under 5 in 2010 at 1km resolution. Adapted from (A. J. Tatem et al., 2013)

While this approach is useful in collating several data sources across a variety of countries, it is less useful for those countries where census data are not reliable and out of date (A. J. Tatem et al., 2013). In the absence of such data sources, Alegana et al. developed a Bayesian hierarchical spatio-temporal model based framework to integrate georeferenced household survey data (such as those collected through the DHS) with a range of spatial data to predict age and gender structure at a 1km by 1km spatial resolution in Nigeria (Alegana et al., 2015). Similar to the bottom-up population mapping approach described above (Wardrop et al., 2018), this approach links geo-located cluster surveys with ancillary satellite data such as land cover, travel time to major settlements, night-time lights, and vegetation index, to predict age structure in un-sampled locations, while statistically accounting for spatial autocorrelation.

Overall, Alegana et al. found that these freely available high-resolution datasets (namely, land cover, night-time lights, travel time to major settlements and vegetation index) were strong predictors for estimating the proportion of the population under 5 years old. Because this framework utilizes only freely available household survey data, in combination with freely available remotely sensed data, this method fills an important niche for modelling high-resolution population distributions in the absence of reliable and modern census data. This work has important ramifications in health and policy metrics, including malaria surveillance and child and maternal mortality prevention, and has been used in vaccination planning and outreach in Nigeria through the Nigeria Vaccination Tracking System (vts.econg.org) (C. Edson Utazi et al., 2018b).

2.2.3.3 Estimating live births and pregnancies

Building off previous techniques described above in mapping age and sex structure (A. J. Tatem et al., 2013), Tatem et al. produced subnational datasets depicting women of childbearing age by 5year intervals and number of live births and pregnancies (Tatem et al., 2014). To do this, Tatem and colleagues compiled age specific fertility rates (ASFR) from national surveys such as the DHS and Multiple Indicator Cluster Survey (MICS), disaggregated by subnational region and urban versus rural rates. These ASFR estimates were then combined with data on women of childbearing age, produced via the methods outlined above, to generate gridded estimates of live births (Figure 2.20). Lastly, these estimates of live births were combined with data on abortions, stillbirths, and miscarriages provided through the Guttmacher Institute to produce estimates of pregnancies per grid cell. These outputs are crucial in planning maternal and newborn health interventions and resource allocation, and were used in the United Nations Population Division's State of the World's Midwifery report for 2014 (UNFPA et al., 2014).

The evolution of these methods form a foundation by which to explore health disparities in maternal and newborn health by quantifying where women of childbearing age, pregnant women, and newborns are. The increased availability of these data, couple with spatially explicit demographic data as collected through household surveys such as the DHS, are creating new opportunities for researchers to spatially disaggregate health outcomes and monitor progress subnationally (Molla et al., 2019). Towards this, spatially explicit methodologies and geospatial analysis are needed to help reduce disparities in maternal mortality and morbidity (Ebener et al., 2015; Makanga et al., 2016; Molla et al., 2019), and to ensure "no-one is left behind" in the SDG era.





2.2.4 Future directions and recommendations

Data visualization through mapping techniques represents a powerful tool for epidemiologists and demographers, particularly as the need for spatial investigation at a subnational scale becomes more prominent on the global health development agenda. Despite this, most research within public health does not incorporate spatial analysis (Auchincloss et al., 2012), and in fields such as maternal and newborn health, challenges persist in building analytic capacity among resource-poor settings (Ebener et al., 2015). Further, while several datasets exist such as census data, census microdata, and national survey data, these are scattered across disparate sources and can be difficult to obtain (Hay et al., 2005; A. J. Tatem et al., 2013), preventing comparison across time and space.

Towards this, Tatem et al. outline recommendations and future directions in improving spatial demographic datasets (Tatem et al., 2012). Firstly, they note that spatial and temporal population

projections are lacking at a subnational level, even amongst international development agencies, such as the United Nations Population Division. While there are a few countries undertaking these projections at the national level, such as India and China, these projections should be scaled up to include other developing countries to allow for more accurate population estimations and resource planning. The current methods used to project these estimates, however, are subject to sources of bias and model uncertainty, which Tatem et al. note must be better quantified and communicated. While efforts to map disease risk have handled the quantification and propagation of this uncertainty through Bayesian posterior distributions (Gething et al., 2015), less research has addressed understanding and visualizing how uncertainty propagates throughout spatial demographic datasets. This uncertainty might be as a result of outdated census data or the size of input administrative units as compared to output population grid cells, for example (Tatem et al., 2012). Without accounting for these drawbacks, spatial demographic datasets and corresponding population and health metrics may be substantially limited by unquantified sources of uncertainty propagating in unpredictable ways.

Finally, there is further a need for greater analytic capacity and data resources, particularly within resource poor settings (Ebener et al., 2015; Molla et al., 2017). Molla et al. outline future directions and policy recommendations for improving the use of GIS within the field of maternal and newborn health, but with broader applications towards health policy in developing countries. They categorize challenges and recommendations into three broad themes: 1) ancillary geospatial and maternal and newborn health data sources; 2) technical and human resources needs; and, 3) community participation (Molla et al., 2017). Noting that while GIS holds substantial potential in reducing the number of preventable maternal and child deaths, reaching this potential will require high quality spatial data and improving current health systems data, such as vital registration statistics and standardized disease surveillance systems (Rushton, 2003). This will require a quality improvement cycle for georeferenced data, integration of data from both the public and private domain, community involvement in participatory mapping efforts to enhance spatial data infrastructures, and finally, regular monitoring and evaluation of inequities in tandem with collaboration from policy makers. Building in-country technical capacity to perform these tasks will be crucial in ensuring recommendations are implemented and in-country policy needs are met, which in turn will support efforts to end preventable deaths among women and children (Molla et al., 2017).

Chapter 3: Study framework

To understand where women and children are dying, why they are dying, and what factors are driving these inequitable deaths, studies synthesizing geography, maternal health, and health inequalities are needed. This work aims to combine methodologies outlined in previous sections to address this need, producing high-resolution estimates of inequalities in utilisation of maternal and newborn health services, and how these inequalities have evolved over time in the East African Community region. Here, this chapter outlines the overall framework for the three studies comprising this work, including a conceptual framework, followed by detailed research questions and hypotheses for each of the studies. A study justification is outlined for 1) study indicators used, 2) the study area of interest, and 3) study methods used to address research questions. Finally, this chapter concludes by outlining scientific contributions and impact this work makes, along with intellectual contributions made for each substantive chapter.

3.1 Conceptual framework

Figure 3.1 applies a monitoring and evaluation framework to outline selected national-level structural policies and regulations, facility-level processes and inputs, individual-level outcomes and indicators and resulting impact. This study will incorporate individual-level outcomes throughout its analyses, which are broadly influenced by: 1) national-level structural elements, such as national policies relating to maternal deaths notification and maternal health expenditure, the number of women represented in national parliamentary bodies and quality of national health information systems; and, 2) facility-level process and inputs, such as the percent of minimum recommended facilities offering basic and comprehensive EmONC services, the percent of women of childbearing age living within a 2-hour travel time of an emergency care facility and quality of care and satisfaction with services provided. The focus of the analyses in the following chapters therefore comprise factors influencing individual-level outcomes and indicators, including antenatal care, skilled birth attendance, postnatal care, and delivery via caesarean section (c-section).

Structure (National-level policies and regulations)

- Maternal deaths notification
 # of women represented in parliament
- % general government health expenditure relative to total government expenditure
- Per capita total expenditure on health
- •% out of pocket expenditure on health relative to total health expenditure
- Postnatal home visits
 after delivery
- Community-based maternity care programmes
- Availability of routine health information system

- **Process and Inputs** (Facility-level resources, services and programmes)
 - •% of recommended minimum health facilities offering emergency obstetric care
 - •% population within 2 hour travel time of emergency care facility
 - •% signal functions met of 7 for basic emergency obstetric care facilities
 - •% signal functions met of 9 for comprehensive obstetric care facilities
 - •Density of doctors, nurses, midwives per 10,000 population
 - Quality of care provided (as measured through routinely available health information systems)
 - Satisfaction of health services provided



Figure 3.1 Study framework for monitoring and evaluating maternal and newborn health care utilisation

3.2 Research questions

3.2.1 Chapter 4 research questions

Chapter 4, entitled *Equality in Maternal and Newborn Health: Modelling Geographic Disparities in Utilisation of Care in Five East African Countries*, examines several MNH outcomes in the context of geographic accessibility to map the likelihood of receiving care before, during, and after delivery. To do this, I utilise a geo-referenced dataset of nearly 10,000 health facilities to generate high-resolution surfaces and administratively relevant maps reflecting the probability of a woman obtaining critical MNH care throughout the five East African Community partner states. Specifically, I produce high-resolution, disaggregated estimates of probability of MNH utilisation as a function of geographic accessibility. These findings have important policy implications for focusing future intervention efforts and allocation of resources, as well as providing a baseline from which to monitor future progress in the SDG era. Within this study I aim to address the following research questions:

- 1. What are the primary drivers of utilisation of antenatal care, skilled birth attendance, and postnatal care within the East African Community study region?
 - a. Does geographic accessibility to a health facility predict utilisation of these services, and does this vary by service utilised (i.e., ANC, SBA, versus PNC)?
- 2. Provided current health facility locations throughout the study region, where are women most likely to utilise these MNH services at both a high-resolution (300 x 300 m) scale and policy-relevant scale (administrative level II)?

3.2.2 Chapter 5 research questions

While Chapter 4 examines utilisation of MNH care as predicted by geographic accessibility, Chapter 5, entitled *Temporal trends in spatial inequalities of maternal and newborn outcomes among four East African countries, 1999 – 2015*, examines how these spatial inequalities have evolved over time. Here, this study builds upon the previous chapter to analyse sub-nationally where spatial inequalities in coverage of vital MNH services have improved, stayed the same, or grown wider within the EAC region. Specifically, I employ a Bayesian inference framework to spatially interpolate coverage estimates for antenatal care, skilled birth attendance, and postnatal care at the administrative II level throughout the study countries at multiple time points using DHS data. I report both absolute and relative indices of change by visualizing the absolute difference between the first and last survey points available for these estimates, and reporting relative performance ratios between the best and worst performing sub-national units for each country. Lastly, I present logistic regression coefficient plots for the best versus worst performing region among each survey time point available, reflecting the extent to which spatial inequality is occurring as well as its evolution over time. Specifically, within this study I aim to address the following research questions:

- 1. How has coverage in SBA, ANC, and PNC within study countries changed absolutely over time, by administrative unit II?
 - a. Which administrative units have had the most improvement in coverage? Which have had the least improvement?
- 2. How does the change in coverage among these administrative units compare relative to each other?
 - a. How wide is the change-in-coverage gap between administrative units within a country?
 - b. Has this change-in-coverage gap improved over time, gotten worse, or stayed the same?

3.2.3 Chapter 6 research questions

Chapter 6, entitled *Estimating uncertainty in geospatial modelling at multiple spatial resolutions: the pattern of delivery via caesarean section in Tanzania*, aims to explore the trade-off between modelled estimates of delivery via c-section in Tanzania and associated uncertainty at increasing spatial resolutions within a Bayesian framework. To do this, I predict prevalence of delivery via csection at the 5km, 50km, and 100km scales, implemented via a stochastic partial differential equation (SPDE) approach utilising the Integrated Nested Laplace Approximation (INLA) technique within the R-INLA package (Rue et al., 2009). I explore the posterior distribution and fit of these models, and visualise estimates and uncertainty at these spatial resolutions. Specifically, within this chapter I aim to address the following research questions:

- 1. What is the trade-off between increasing spatial resolution and model fit?
 - a. How does the overall posterior distribution of the 95% credible interval change with increasing spatial resolution?
- 2. How does uncertainty in estimates of prevalence of delivery via c-section propagate spatially across the landscape at varying resolution sizes?
 - a. Are there areas with consistently high or low estimates of delivery via c-section across spatial resolutions?
 - b. Are there areas with consistently high or low uncertainty across spatial resolutions?

3.3 Study indicators

3.3.1 Antenatal care

Routine antenatal care appointments are critical to identifying risks, preventing and managing complications, and educating and promoting health throughout pregnancy to ensure a safe delivery (Bloom et al., 1999; World Health Organization, 2016a). Through routine monitoring and contact

of pregnant women, maternal and perinatal mortality is reduced both directly and indirectly: firstly, through detection and management of complications occurring in pregnancy, labour, and delivery; and secondly, through prevention and management of simultaneous disease, such as malaria and HIV. While utilisation of ANC has increased in low and middle-income countries since 2002, less than two-thirds of women obtained the recommended 4 or more ANC visits between 2007 and 2014 (World Health Organization, 2016a). Further, since that time the WHO has increased the number of recommended ANC contacts to eight, based on evidence suggesting the four-visit model did not adequately reduce perinatal mortality or improve women's satisfaction with the quality of care obtained (World Health Organization, 2016a). This suggests even fewer women are obtaining the required number of 8+ ANC contacts.

Many social, cultural, and geographical factors influence the decision and extent to which a woman will obtain the recommended number of antenatal care (ANC) visits, however. Simkhada and colleagues aimed to identify these barriers, performing a literature review to examine the critical determinants of predicting ANC coverage across the socioeconomic spectrum, specifically within the context of developing countries (Simkhada et al., 2008). Among twenty-eight studies, they found a variety of social, political and economic factors such as education, income, marital status, a history of complications, and supply-side factors including cost and availability, to be important predictors of ANC utilisation. More recently, Saad-Haddad performed a cross-country analysis exploring determinants of ANC utilisation throughout seven Countdown Initiative countries, similarly finding large disparities in obtaining 4+ ANC visits by household wealth, woman's education, and place of residence (Saad–Haddad et al., 2016). Further, conducting a systematic literature review of twenty-one qualitative studies among low and middle-income countries, Finlayson and Downe (2013) identified emergent themes in women's experiences as to why they did not obtain or utilise antenatal care services. These primarily included reasons surrounding: i) pregnancy perceived as a 'normal life event' (e.g., only attending ANC appointments when unwell); ii) lack of necessary resources (e.g., transportation and other associated costs, time off work or childcare, or inadequate infrastructure); and, iii) poor prior experiences (e.g., supply-side factors such as lack of staff, long waiting times, or disrespectful care) (Finlayson and Downe, 2013).

Importantly, Simkhada et al. noted that geographic accessibility was a key driver in obtaining antenatal care, comprised of determinants such as place of residence, distant to healthcare facilities, and type of transport available. Specifically, women in urban areas were generally found to be at higher odds of obtaining ANC as compared to those in more rural settings, and were found to be more likely to obtain ANC from a skilled health provider, in line with later findings by Saad-Haddad and colleagues (Saad–Haddad et al., 2016). Further, Simkhada et al. found that increasing distance to health facilities was significantly associated with fewer ANC contacts, supported by both quantitative and qualitative findings (Finlayson and Downe, 2013; Saad–Haddad et al., 2016;

Simkhada et al., 2008). Lastly, they noted that the type of geography encountered was a barrier to obtaining ANC, including poor road conditions, uncomfortable and crowded public transport, and inability to cross major bodies of water (Simkhada et al., 2008).

In addition to these findings, Kyei et al. examined the effect of distance to health facility on number and timing of ANC visits in rural settings within Zambia, finding that increasing distance to the nearest health facility strongly influenced the quality of care a woman received (Kyei et al., 2012). These findings suggest that geography is an important barrier in both the quality and quantity of ANC visits that a woman obtains, yet varies drastically by the social and cultural factors specific to each country. Country and region-specific analyses incorporating geospatial approaches are therefore critical in improving the utilisation of and satisfaction with antenatal care contacts.

3.3.2 Skilled birth attendance

Skilled birth attendance (SBA), along with antenatal care, continues to be a key driver of maternal and newborn mortality with persisting inequalities (WHO, 2015a). As most life-threatening complications occur around the time of labour and delivery, it is vital for women in labour to be monitored by an attendant trained in the identification and management of obstetric emergencies. This is evidenced by the inclusion of coverage of skilled birth attendance in both Millennium Development Goal and Sustainable Development Goal indicators (Table 2.1), and has been identified by the World Health Organisation as one of the "single most important factor(s) in preventing maternal deaths" (World Health Organization, 1999).

Globally, over 80% of births were attended by skilled health personnel, yet while skilled birth attendance in Sub-Saharan Africa increased over 2012 to 2017, just over 50% of births were attended by a skilled health professional (World Health Organization, 2018a), suggesting barriers remain in the utilisation and uptake of skilled attendance at birth. Despite its importance as a key intervention in preventing maternal and newborn deaths, disparities in skilled birth attendance both between and within low- and middle-income countries persist (Gwatkin et al., 2007; WHO, 2010b, 2015a). SBA as a maternal health intervention has further been identified as having the most pronounced economic inequalities (Gwatkin et al., 2007), in line with the "inverse care law" suggesting that the poorest populations in greatest need of health services often receive the fewest services (Kruk et al., 2008; Victora et al., 2003). Towards this, Kruk and colleagues explored SBA utilization among 45 countries with DHS surveys, finding that governmental policies focused on ensuring primary education for all can have a marked effect in increasing skilled attendance at delivery (Kruk et al., 2008). These results suggest that a woman's education is a primary driver in utilising skilled birth attendance, in line with findings among other studies across low- and middleincome countries (Gitimu et al., 2015; Hazarika, 2011; Mpembeni et al., 2007; Yanagisawa et al., 2006).
In their systematic literature review, Gabrysch and Campbell further identified twenty key determinants predicting utilisation of SBA, and grouped them into four major themes, including sociocultural factors, perceived benefit/need, economic accessibility, and finally, physical accessibility (Gabrysch and Campbell, 2009a). Specifically, among 80 original articles and 2 reviews, they found common sociocultural determinants, such as age, parity, education, income, and urban residence to be key in predicting use of SBA during delivery. Importantly, they note that the majority of studies tended to focus on these socioeconomic variables, while neglecting variables related to physical accessibility and qualitative measures of perceived need. Among those studies examining physical accessibility and need, they found distance to health facilities and quality of care received to be influential in facilitating health service use during delivery. Their findings ultimately suggest that future analyses of SBA must critically explore physical and geographic accessibility measures to fully understand the use of skilled birth attendance and health facility use, and ultimately reduce the inequalities persisting amongst these subgroups.

3.3.3 Postnatal care

The hours, days, and months following delivery also represent a crucial time for both women and newborns alike. Half of all postnatal maternal deaths occur within the first week of delivery, with the majority occurring in the first 24 hours of birth; similarly, the majority of newborn deaths occur in the first days and week of life (The Partnership for Maternal, Newborn and Child Health, 2006). Yet while the majority of deaths occur in this timeframe, the PMNCH has identified interventions targeted on this period as among "the weakest of all reproductive and child health programmes in the region" (The Partnership for Maternal, Newborn and Child Health, 2006). Indeed, it is a period that is often characterized by poor quality of care or an absence of care entirely, and is poorly researched (World Health Organization and Department of Maternal, 2013).

Unsurprisingly, fewer studies have examined the effects and determinants of postnatal care (PNC), despite its potential to reduce global maternal and newborn deaths. However, among studies exploring key drivers of postnatal care, similar socioeconomic characteristics arise as those described for ANC and SBA. For example, Dahiru and Oche (2015) explored utilisation of ANC, SBA and PNC in Nigeria, finding education, place of residence, and wealth to be consistent predictors of health service utilisation (Dahiru and Oche, 2015). These findings were echoed by Somefun and Ibisomi (2016), who also explored PNC utilisation among women in Nigeria, finding ANC use, distance to facility, education, place of delivery/skilled attendance at delivery, geographic region and wealth to be important predictors of PNC services (Somefun and Ibisomi, 2016). These results are generalizable to other low- and middle-income countries, where similar factors including ANC use, skilled birth attendance, education, wealth, and distance to facility were found to be key compositional factors for women in Kenya (Akunga et al., 2014), Tanzania (Mohan et al., 2015) and Bangladesh (Islam MR and Odland JO, 2011).

While no systematic literature review has been undertaken examining predictors of PNC utilisation, Fort and colleagues undertook a comprehensive analysis to explore the occurrence, timing, and characteristics associated with utilising PNC throughout 30 developing countries using DHS data (Fort et al., 2006). Their findings showed that as much as 70% of women giving birth in these countries do not receive adequate postnatal care, with the average timing of PNC among noninstitutional deliveries around 3 days after birth. Compositional characteristics associated with utilisation of PNC included wealth, education, urban residence and receipt of ANC prior to birth. These findings are in line with characteristics associated with both ANC and SBA, suggesting that these outcomes are tightly correlated with each other (Akunga et al., 2014; Dahiru and Oche, 2015; Islam MR and Odland JO, 2011; Somefun and Ibisomi, 2016). Ultimately, this suggests that an understanding of more generalized health service utilisation patterns might be an effective approach to increasing service utilisation among all three interventions (Dahiru and Oche, 2015).

The commonalities among these compositional factors suggest they are critical in understanding the factors influencing the utilisation of ANC, SBA and PNC. This understanding can then be applied to identify vulnerable subpopulations for disaggregated analysis. However, the use of these services continues to be unequally distributed. Specifically, the WHO found that while many vulnerable countries made overall progress in key intervention coverages, many inequalities persisted as a result of economic subgroups, including those with less education and wealth (WHO, 2015a). Notably, antenatal care and skilled birth attendance continue to have the greatest socioeconomic inequalities, suggesting an urgent need for disaggregated analysis to identify vulnerable subgroups and targeted interventions (Barros et al., 2012; WHO, 2015a).

3.3.4 Delivery via caesarean section

Rates of caesarean sections (c-section) performed around the world have seen an unprecedented rise over the several few decades across all WHO regions, yet these rates show vast differences in the degree of incline based on high-income versus low- and middle-income countries (Betrán et al., 2007). The optimal rate at which c-sections should be performed remains controversial, with the WHO suggesting rates higher than 10 - 15% were 'unjustifiable' in the mid 1980's (World Health Organization, 1985). Yet while delivery via c-section can save the life of both woman and child when medically indicated, it can conversely pose otherwise unnecessary risks in both the long and short term (Sandall et al., 2018). The variability of c-section rates by country reflects this, with strategies in high-income countries focusing on the reduction of unnecessary c-section, and strategies in low- and middle-income countries focusing on increasing the accessibility and availability of emergency c-section procedures (Betrán et al., 2007). Within African settings, the rate of delivery via c-section has increased since 2010, but remains the lowest WHO region to perform the procedure, at around 10% of births being performed via c-section in 2015, with rates as small as 0.6% in South Sudan (Wise, 2018). Crucially, delivery via c-section remains a vital and

life-saving emergency procedure among low-income countries, strongly correlated with decreasing maternal and newborn deaths, with c-section rates below 10% suggesting that the most vulnerable pregnant women are not getting adequate care (Althabe et al., 2006).

Within resource-poor settings, medically indicated c-sections frequently do not occur largely as a result of barriers including accessibility, availability, and acceptability of care (Betrán et al., 2007; Wise, 2018). Irani and Deering explored challenges affecting accessibility to c-section procedures where medically indicated, performing a systematic review comprising 19 studies throughout 16 Sub-Saharan African countries. They found the average rate of c-section delivery was 3.6%, ranging from 1.5% to 7.1%, and reported major barriers to care including poverty, accessibility to care, and shortage of skilled healthcare personnel (Irani and Deering, 2015). Specifically, they noted that the number and availability of trained healthcare providers and facilities offering 24-hour coverage posed a significant challenge throughout the region, particularly amongst rural areas. Further, poverty was found to be an important barrier to uptake and utilisation of necessary c-section delivery, as many facilities would not provide care until the necessary fees had been paid.

In addition to these availability barriers, they noted accessibility barriers such as poor road conditions, poor transportation infrastructure such as emergency vehicles, inadequate public transportation systems, and long distances to health facilities capable of providing the procedure. They note that these geographic barriers are crucial barriers in obtaining appropriate surgical care, attributing between 88% to 99% of maternal deaths as preventable if women were able to reach emergency obstetric care within a timely manner (Irani and Deering, 2015; Knight et al., 2013; Maternal Health and Safe Motherhood Programme, 1996). These barriers mirror those encountered when obtaining other key MNH services, including ANC, SBA, and PNC, suggesting that a woman's geographical and environmental context can play a critical role in life or death when encountering obstetric emergencies.

3.4 Study area

Inequalities in access to MNH services such as antenatal care, postnatal care, skilled attendance at delivery, and delivery via c-section persist not only on the global level, but within regions, as well. Within sub-Saharan Africa, the Eastern Africa sub-region as defined by the UN MDG groupings saw the greatest reduction in MMR between 1990 and 2015, with a 57% overall change and 3.4% average annual change (WHO et al., 2015). However, this overall reduction masks heterogeneity within even this sub-region, with the countries falling in the East African Community region (comprised in 2015 of Burundi, Kenya, Rwanda, Tanzania, and Uganda, with the addition of South Sudan in 2016) seeing varying amounts of improvements over the past two decades. The EAC region (as it stood in 2015 when these analyses began) therefore represents a relevant sample of sub-Saharan African countries ideal for exploring variation in utilisation of maternal and newborn

care services, comprised of both countries that have both achieved MDG targets, as well as those with persistently high rates of maternal mortality.

3.4.1 Burundi

Figure 3.2 shows the distribution of women of childbearing age in Burundi for the year 2015, using data obtained from WorldPop (www.worldpop.org). Burundi had an estimated 488,000 births in 2015, with a total 3,400 maternal deaths in 2013 and a lifetime risk of maternal death at 1 in 22. In 1990, the country had an MMR of 1220 maternal deaths per 100,000 live births, representing one of the highest in the region, behind only Rwanda (WHO et al., 2015). In 2015, this number had been reduced to 712 maternal deaths per 100,000 live births, with an overall 41.6% reduction in MMR and 2.2% average annual change. These estimates suggest that Burundi has made insufficient progress towards meeting MDG target 5a of 75% reduction in MMR, as the reduction was under greater than 25% but less than 50% (WHO et al., 2015). Using 2010 DHS data, only 1 in 3 women living in Burundi were estimated to have obtained 4+ ANC visits regardless of socioeconomic status, with even less women receiving a postnatal visit within 2 days (30%). Further, less than 2 in 3 women (60%) were estimated to have births attended by a skilled health professional, with wide gaps observed between the poorest and richest quintiles (UNICEF and WHO, 2015; WHO, 2016b). Lastly, only 4% of births between 2005 to 2010 were delivered via c-section (World Health Organization, 2018b).

While having policies in place to ensure postnatal home visits within the first week of birth, Burundi represented the only country among the EAC region to lack a fully costed national implementation plan for maternal, newborn and child health as of 2015. Further, there were no laws or regulations allowing for adolescents to access contraceptives without parental or spousal consent, and no maternal deaths notification system. The density of health workers per 10,000 population was 2.2 in 2004, and met only 27% of the recommended minimum number of Emergency Obstetric Care facilities in 2010(UNICEF and WHO, 2015). These suggest that Burundi's policies, health systems, and financing are lacking for women and newborns, with much progress left to be made.

Chapter 3

Chapter 3



Figure 3.2. Distribution of women of childbearing age in Burundi, 2015. White lines delineate 2010 DHS boundaries, while grey lines delineate administrative II units.

3.4.2 Kenya

Figure 3.3 shows the distribution of women of childbearing age in Kenya for the year 2015, using data obtained from WorldPop (www.worldpop.org). In 2015, Kenya had 1,571,000 births with a total of 6,300 maternal deaths in 2013 and a lifetime risk maternal death at 1 in 53. While Kenya started with fewer maternal deaths per 100,000 live births, it has seen very little progress in reducing its MMR since 1990, designated as 'No progress' towards achieving MDG 5a (WHO et al., 2015). There has been a total reduction of 26% from 687 deaths in 1990 to 510 deaths in 2015, representing an average annual reduction of 1.2%. However, it is important to note this point estimate falls within an 80% uncertainty interval suggesting these numbers could have actually increased per year (lower estimate: -0.5%, upper estimate: 2.8%). Other services, however, such as antenatal care and postnatal care received were higher than neighbouring Burundi, with just over half of women receiving a postnatal visit within 2 days and nearly 60% receiving the recommended number of antenatal care visits. Skilled birth attendance during delivery, however, remained comparable to Burundi, at 62%, with equally high gaps of inequality observed between the poorest and richest households (UNICEF and WHO, 2015; WHO, 2016b). Finally, Kenya represented the country with the largest proportion of recommended minimum number of emergency obstetric care services available at over 50%, as compared to proportions between 25% - 35% for the rest of the region (UNICEF and WHO, 2015). This translated into 8.7% of births in 2009 delivered via csection, over twice the proportion observed for Burundi (World Health Organization, 2018b).

Kenya has a fully costed implementation plan in place, including other health systems and financing, to protect the health of women and their newborns. As of 2015, the country had laws and regulations allowing adolescents to access contraceptives without spousal or parental consent, as well as a national maternal deaths notification system in place. Further, there were several policies in place aimed at preventing complications/mortality, including Kangaroo Care for low-weight or preterm babies, antenatal corticosteroids as routine management of preterm labour, and postnatal home visits within the first week of birth. The density of health workers per 10,000 population was just over 10 in 2013, yet in 2003, the country had met only 54% of the recommended EmONC facilities for the country (UNICEF and WHO, 2015). Together these policies suggest Kenya has prioritized maternal and newborn health, but a gap exists in implementing these into actual deaths prevented, with the third highest lifetime risk of dying in childbirth in the region.



Figure 3.3. Distribution of women of childbearing age in Kenya, 2015. White lines delineate 2014 DHS boundaries, while grey lines delineate administrative II units (as of 2013).

3.4.3 Rwanda

Figure 3.4 shows the distribution of women of childbearing age in Rwanda for the year 2015, using data obtained from WorldPop (www.worldpop.org). Rwanda had 363,000 births in 2015 with a total of 1,300 deaths in 2013 and a lifetime risk of death due to maternity at 1 in 66. Among one of only four countries, Rwanda successfully achieved the MDG 5a target of reducing the maternal mortality ratio by 75%. In 1990, Rwanda's MMR was the highest among the five EAC countries at 1300 per 100,000 births. In 2015, this number fell to less than 300, representing a nearly 78% reduction and an average annual percent reduction of 6% (WHO et al., 2015). In addition to achieving this reduction in MMR, skilled attendance at birth increased substantially, from 26% of

births attended by skilled health personnel in 1992 to over 90% in 2014/2015, according to DHS data. This represented the highest overall skilled birth attendance in the region. However, these estimates were still subject to gaps in socioeconomic status, with skilled birth attendance coverage among poorer households around 60%. Despite progress in increasing skilled birth attendance and reducing MMR, obtaining recommended antenatal care and postnatal remained low, at 44% and 42%, respectively (UNICEF and WHO, 2015). Lastly, Rwanda had the highest proportion of births delivered by c-section in the region at 13.0% between 2010 and 2015 (World Health Organization, 2018b).

While not having laws or regulations allowing adolescents to access contraceptives without parental or spousal consent, Rwanda did have a fully costed maternal, newborn, and child health implementation plan in place, a routine maternal deaths notification as of 2015, and policies ensuring postnatal home visits within a week of birth, use of antenatal corticosteroids in preterm management, and Kangaroo Care for preterm/low-weight births. The density of health workers per 10,000 was slightly less than Kenya, however, with 7.5 as of 2010, and meeting only 35% of the recommended EmONC facilities in 2007 (UNICEF and WHO, 2015). While there is progress to be made in terms of strengthening policies and health systems, Rwanda remained the only country within the EAC community to achieve the MDG5a target, suggesting they have prioritized maternal and newborn health over the previous years.



Figure 3.4. Distribution of women of childbearing age in Rwanda, 2015. White lines delineate 2015 DHS boundaries, while grey lines delineate administrative II units.

Chapter 3

3.4.4 Tanzania

Figure 3.5 shows the distribution of women of childbearing age in Tanzania for the year 2015, using data obtained from WorldPop (www.worldpop.org). Tanzania saw a total of 7,900 maternal deaths in 2013 among over 2 million births in 2015, with a lifetime risk of maternal death at 1 in 44. In 1990, the country recorded an MMR at just under 1,000 deaths per 100,000, which fell to the just under 400 in 2015. This therefore represented a reduction of approximately 60% with an average annual reduction rate of 3.7%, marking the country as 'making progress' towards achieving MDG 5a (WHO et al., 2015). Wide gaps were again observed among skilled birth attendance between the wealthiest and poorest households using the 2010 DHS, with only 1 in 3 of the poorest women having skilled attendance at delivery, as compared to over 90% of the richest. Under half of women in the country obtained 4+ ANC visits, with less than 1 in 3 obtaining a postnatal visit 2 days after birth (UNICEF and WHO, 2015). Lastly, only 5.9% of births between 2010 and 2016 were delivered via c-section (World Health Organization, 2018b).

Tanzania's health policies and systems are similar to that of Kenya, with a fully costed national implementation plan for maternal, newborn and child health, as well as laws protecting adolescent accessibility to contraceptives without other's consent. While Tanzania did not have routine antenatal corticosteroid management for preterm labour, the country did ensure Kangaroo Care for preterm/low-weight births, and postnatal home visits within a week of birth. The density of health workers, however, was unacceptably low at only 4.7 doctors, nurses and midwives per 10,000 population, and the country met only 21% of the minimum recommended number of EmONC facilities as of 2005 (UNICEF and WHO, 2015). This suggests that while Tanzania has in place several encouraging policies, health systems and financing could be strengthened.

Chapter 3



Figure 3.5. Distribution of women of childbearing age in Tanzania, 2015. White lines delineate 2015 DHS boundaries, while grey lines delineate administrative II units.

3.4.5 Uganda

Figure 3.6 shows the distribution of women of childbearing age in Uganda for the year 2015, using data obtained from WorldPop (<u>www.worldpop.org</u>). Uganda had 1,665,000 births in 2015 with a total of 5,900 deaths in 2013 and a lifetime risk of maternal death at 1 in 44, similar to Tanzania. Also consistent with Tanzania, Uganda was marked as 'making progress' towards achieving MDG 5a, with a reduction of just over 50% in MMR from 687 deaths per 100,000 in 1990 to 343 deaths in 2015, representing an average decline of almost 3%. Consistent with the entirety of the EAC region, skilled attendance presented some of the widest inequalities in maternal health indicators, with around 40% of women in the poorest household receiving skilled birth attendance, as compared to almost 95% of the richest households. In 2011, just under half of women surveyed in the DHS interview had obtained 4+ ANC visits, while 1 in 3 had obtained a postnatal visit 2 days after delivery (UNICEF and WHO, 2015). Lastly, Uganda had similar rates of delivery via c-section as Tanzania and Burundi, with only 5.3% of births between 2006 and 2011 delivered via c-section (World Health Organization, 2018b).

Uganda's policies for promoting maternal and newborn health varied, with a fully costed national implementation plan for maternal, newborn and child health, but only partial regulations protecting adolescents' ability to access contraceptives without parental or spousal consent. Further, while postnatal home visits and Kangaroo Care for preterm/low-weight babies were routine, there were no laws as of 2015 promoting use of antenatal corticosteroids for preterm management. However, the density of health workers was highest in Uganda as of 2005, with just over 14 doctors, nurses and midwives per 10,000 population, and 34% of the recommended minimum number of EmONC

facilities available in 2008 (UNICEF and WHO, 2015). This suggests that the health workers and infrastructure exist within the country, but policies and implementation could be strengthened.



Figure 3.6. Distribution of women of childbearing age in Uganda, 2015. White lines delineate 2016 DHS boundaries, while grey lines delineate administrative II units.

3.5 Study methods

To estimate spatially and temporally disaggregated probabilities of utilisation of MNH services, statistical methods are needed which can estimate parameters of sub-populations at geographically small areas while accounting for sample error and study design (Ebener et al., 2015). Several approaches such as small area estimation (SAE), mixed effects modelling, and Bayesian inference exist to accomplish this, each with their own caveats. Within MNH research, SAE models provide a powerful tool for researchers to link measures of interest at small geographic areas (such as census enumeration areas) with covariates of interest to model estimates at these small areas (Ebener et al., 2015). Two broad SAE methodologies include area level techniques (Fay and Herriot, 1979), where covariates are available at the areal level, and nested-error models (Battese et al., 1988), where covariate information is available at the individual level and linked to primary measures at the areal level (Ebener et al., 2015). These methods have been used within MNH research to link demographic and fertility data from household surveys (such as the DHS) to census data, where prevalence of contraceptive use could be derived at geographic areas smaller than the DHS regions where estimates were reported (Johnson et al., 2012). The applications of these approaches can be limited, however, by methodological considerations and data limitations (Ebener et al., 2015), especially in scenarios with zero-inflated data (Chandra and Sud, 2012), or in situations where auxiliary data such as census data are difficult to obtain or proprietary.

In this work, I employed a hierarchical mixed effects modelling approach, followed by a structured additive regression model within a Bayesian framework. Notably, hierarchical mixed effects modelling and Bayesian models are able to produce geospatially interpolated estimates at geographical administrative units smaller than the boundaries at which the data were sampled (Gething et al., 2015), while also accounting explicitly for spatial correlation and utilising openly available data. These methods therefore make them ideal for inclusion in this work to address the research objectives outlined in Section 3.2.

Specifically, mixed effects modelling approaches have been utilised previously in the literature when working with DHS data, and are ideal for accounting for the multistage sampling design of the data (Magadi and Desta, 2011; Snijders and Bosker, 1999). I further expand the multilevel approach used in Chapter 4 to include a latent Gaussian regression model within a Bayesian framework in the third chapter. Specifically, I use the Integrated Nested Laplace Approximation (INLA) regression technique in Chapters 5 and 6 to control for spatially and temporally correlated effects using the DHS data. This technique has similarly been utilised with the DHS literature to predict small area outcomes, while also quantifying associated uncertainty (Achia, 2014; Gething et al., 2015; Mtambo et al., 2015; Neal et al., 2016; Niragire et al., 2015).

3.5.1 Hierarchical mixed effects modelling

In Chapter 4, I model the probability of receiving 4+ ANC visits before delivery, SBA during delivery, and receiving PNC within 48 hours of delivery, using DHS data throughout the Burundi, Kenya, Rwanda, Tanzania and Uganda. I visualise these probabilities firstly at a high-resolution (300 x 300 m grid cells), and then aggregate to more policy relevant scales (administrative level 2, typically districts) to highlight spatial heterogeneity and sub-national inequalities in MNH care utilisation throughout the region of interest. By producing estimates at a high spatial resolution such as 300m, these probabilities can be aggregated and visualised at any policy-relevant level later on, including administrative boundaries, health catchment boundaries, or census enumeration areas. These estimates therefore represent substantive methodological contributions (at the 300m scale) to the literature, as well as policy contributions (at the administrative level 2 boundary).

Because of the importance of physical accessibility for care utilisation, my primary explanatory variable of interest is geographic inaccessibility to the nearest health facility. Towards this, I explore MNH utilisation within this context by creating a gridded travel impedance surface reflecting overall ease of traversing each 300 x 300 m square in the study region, given the topography and transport infrastructure of the region. Because consolidation of the various input data sources (e.g., land cover, elevation, water bodies, etc.) necessitate identical cell size, data were aggregated to the coarsest resolution available, thereby driving a 300m resolution. I subsequently use this impedance surface to inform a cost-distance analysis, generating a geographic inaccessible) to 7

(highly inaccessible) and represent ease of access to the nearest health facility among a dataset of 9,314 facilities. I include this inaccessibility score as a covariate in a hierarchical mixed effects logistic regression model by overlaying DHS cluster locations on this generated accessibility surface and extracting scores for each cluster location. Finally, I use the resulting model of best fit to visualise probabilities of obtaining MNH care for each 300 x 300 m square throughout the region. Random effects included in the model consist of DHS regions (N = 61) within five EAC countries, while fixed covariates include cluster-level (N = 3,311) effects of accessibility and rurality, and individual-level (N = 25,325) effects of wealth, education, and an interaction between age and total children delivered.

3.5.2 Bayesian modelling

In Chapters 5 and 6, I extend the hierarchical spatial methodologies outlined in Chapter 4 by incorporating temporal change and controlling for spatial effects using a Bayesian inference framework. Specifically, I extend the previous methodologies among ANC, SBA, and PNC indicators to include multiple time points using geo-located DHS surveys extending backwards in time as far as 1999. To create these surfaces, I utilise a structured latent Gaussian additive model, which has been used in the literature previously to control for spatially and temporally correlated effects which are common to DHS datasets (Achia, 2014; Gething et al., 2015; Mtambo et al., 2015; Neal et al., 2016; Niragire et al., 2015). I report both absolute and relative inequality measures, including visualization absolute difference in sub-national coverage estimates per administrative II unit, using the first and last time points available. I also report relative inequality measures by plotting ratios and regression coefficients of best versus worst performing administrative units for each time point available within the study countries, reflecting countries where progress has been made in lessening inequalities and countries with little to no change. Model results and associated posterior parameters such as inter-quartile range are reported, with aims of highlighting sub-national areas in the EAC region where spatial inequalities have narrowed, widened, or remained the same.

Finally, in Chapter 6, I again employ a Bayesian hierarchical modelling framework, utilising a stochastic partial differential equation (SPDE) spatial regression approach in order to predict outcomes at a continuous spatial resolution, implemented using the Integrated Nested Laplace Approximation (INLA) technique within the R-INLA package (Rue et al., 2009). This approach was suitable for these analyses as these spatial processes are generally well captured by a Gaussian field with Matérn correlation (Blangiardo and Cameletti, 2015). Similar models have been used in previous research combining DHS data and geospatial covariates to predict high-resolution childhood vaccination coverage by disaggregating aerial surveillance data (Bosco et al., 2017; C. Edson Utazi et al., 2018b; C. E. Utazi et al., 2018), and specifically allow for a statistical link between the areal DHS data and high-resolution spatial covariates and random effects (C. E. Utazi

et al., 2018). I estimate prevalence of delivery via c-section at the 5km, 50km, and 100km scales in order to quantify the trade-off between model outcomes and associated uncertainty at increasing spatial resolution. I explore the posterior distribution of the 95% credible intervals at each spatial resolution, and visual outcomes and uncertainty at these spatial resolutions.

3.6 Scientific contribution

3.6.1 Knowledge gaps

With the close of the MDG agenda in 2015, Ebener and colleagues sought to outline the state of the field of maternal and newborn health geography, synthesizing what methods and techniques had been utilised in the field at the time (Ebener et al., 2015). They further established a comprehensive framework for analytic methods and outlining what work needed to be done moving into the future. Performing a "rapid" literature review, they identified 33 studies that were geographical in scope and analysing MNH outcomes, consolidating these studies into a table outlining the geographical scope, study aims and objectives, analytic methods and frameworks used, and region and country of study interest. Overall, they found a substantial increase in the number of studies employing spatial methods within the field of maternal and newborn health since 2010, most commonly studying geographic access to health services. Based on this literature review, they summarized three primary analytic objectives observed in the literatures: "1) thematic mapping (creation of maps to convey information about a topic or theme); 2) spatial analyses (extraction or creation of new information from spatial data); and, 3) spatial modelling (spatial analysis that includes the use of a mathematical model to simulate natural or anthropogenic phenomena)" (Ebener et al., 2015). Despite the encouraging trend towards increased use of GIS techniques and spatial methods within MNH, they note that little within-country capacity exists to continue supporting these analytic demands over the years, preventing crucial monitoring over time of disaggregated data and health inequalities.

This state of the art paper was soon followed up by a scoping literature review performed by Makanga and colleagues, evaluating the use of GIS techniques within the field of MNH and identifying knowledge gaps and opportunities from 70 studies (Makanga et al., 2016). Echoing findings from Ebener et al., they noted two key themes emerged surrounding modelling access to maternal services and identifying risks associated with adverse maternal outcomes. Specifically, they noted that the majority of articles modelling access to maternal services focused on "geographic access to care on the basis of the spatial distribution of [primary] health facilities", with few articles examining access to tertiary level of care. Further, they noted that the majority of studies examining access to care neglected low- and middle-income countries, among which disproportionately bear a high burden of adverse maternal health outcomes. Lastly, they identified a key gap in knowledge needing prioritization over the coming years is incorporation of seasonal

Chapter 3

variation in spatial modelling of access to care, where many studies do not have any temporal effect (Makanga et al., 2016).

Finally, Gabrysch and Campbell performed a systematic literature review to identify determinants of accessing skilled birth attendance during delivery within low- and middle-income countries (Gabrysch and Campbell, 2009a). Among the overall themes emerging from this review was the effect of physical accessibility. Similarly discussing the effect of accessing care and distance, they also noted that the effect of distance of health facilities may be intensified when combined with lack of transportation options or poor road infrastructure. Further, they found the effect of distance is also impacted by the perception of quality, with women willing to travel further distances for health facilities offering higher perceived quality of services. Despite the role of distance and transport on service utilisation, they noted a lack of high quality, comparable geographical data limited further analysis of the role of physical accessibility on maternal health outcomes, suggesting many studies have failed to address this important need (Gabrysch and Campbell, 2009a).

3.6.2 Contribution to the literature

Knowledge gaps persist in identification of inequalities in health and access to health services using high quality, comparable disaggregated data. Identifying these disparities in high-resolution over both space and time will be critical in ensuring SDG targets are achieved by the year 2030 (Requejo and Bhutta, 2015; WHO, 2015a). Comparable spatial and temporal disaggregation of maternal child health is an important first step in identifying where and why disparities are occurring within countries, as well as monitoring their progress over time. Establishing scientifically rigorous methodologies for achieving the spatial and temporal disaggregation of health data will therefore prove applicable for not just the poorest countries in the world, but for all countries in which inequalities persist. Further, working closely with in-country partners who have a quantifiable stake in improving the health of their constituents, as well as an in-depth understanding of the social and cultural context in which these disparities occur, will be key to ensuring sustainable solutions are found that will have lasting impact well beyond the SDG era.

Here, this work makes novel substantive and methodological contributions to the body of maternal and newborn health epidemiology literature through spatial and temporal disaggregation of inequalities in MNH care at high-resolution and policy relevant scales across multiple countries in East Africa. While previous studies have modelled travel time or distance to facilities, fewer have examined utilisation of MNH services as an emergent property of geographic accessibility. Among those studies which have explored the impact of geographic accessibility on choice or utilisation of MNH services, analyses have predominantly been performed on the national scale (Gabrysch et al., 2011; Pilkington et al., 2012). Among those studies which have performed sub-national analysis, analyses have been predominantly limited to a single country (Målqvist et al., 2010). No prior

studies have quantified high-resolution spatial inequalities in utilisation of MNH services among multiple countries within a geographic region, nor highlighted how these spatial inequalities have changed over time on a sub-national scale, in both absolute and relative terms.

This work aims to fill in these gaps when such analyses are particularly needed as the world progresses through the SDG era. To ensure sustainability and community engagement in line with SDG targets, results from Chapters 4 and 5 are contextualized with input and collaboration from relevant stakeholders, such as policy makers through the intergovernmental East African Community (EAC), allowing for both academic contribution and real-world impact. The outputs of this research therefore work synergistically to add to the academic body of literature, while informing policy efforts and promoting policy relevance through collaboration with intergovernmental partners in the study region to ensure no one is left behind moving into the SDG era.

However, the production of MNH estimates at high spatial resolutions has important methodological limitations which should be fully considered, lest results are over-interpreted (Boerma et al., 2018). Towards this, the analyses comprising Chapter 6 of this work make substantial contributions to the academic literature by exploring the theoretical trade-off between increasing spatial resolution in model inputs, and resulting model uncertainty in estimates. While the production of global estimates at high spatial resolutions has been on the rise, no studies have commented on the optimal spatial resolution at which to report outcomes, or specifically examined the effect of increasing spatial resolution on modelled estimates. By quantifying this trade-off, this work fills an important knowledge gap, and further presents a novel method by which to communicate associated uncertainty to policy makers in a more intuitive way.

3.6.3 Impact

These analyses were conceptualized and implemented with feedback and collaboration through policy makers at the East African Community, an intergovernmental organization representing a coalition of member states in Burundi, Kenya, Rwanda, South Sudan, Tanzania and Uganda. Before analyses began, CWR travelled to EAC Headquarters based in Arusha, Tanzania, to discuss policy needs and knowledge gaps with the Principal Health Systems and Policy Analysis Officer working with the EAC Secretariat within the Reproductive Maternal Newborn Child and Adolescent Health (RMNCAH) group. Throughout the period of analysis, I consulted officers at the EAC for guidance on ensuring analyses and results were within the scope of ongoing EAC activities and needs. Based on this collaboration, results from Chapter 4 have been used within regional policy briefs funded by the UNFPA's East and Southern Africa Regional Office representing the state of SRMNAH workforce in the region. Lastly, to ensure these analyses have ongoing and sustainable impact, I travelled to EAC Headquarters to conduct a within-region GIS Workshop, with over 25 delegates representing all 5 member states (South Sudan was not a

member state at the time of the workshop). Based on this workshop, future GIS needs and objectives were outlined by the EAC Secretariat, including increased GIS infrastructure, data, and analytic capacity.

Lastly, the analyses comprising Chapter 4 of this work were recently internally assessed for inclusion in the 2021 Research Excellence Framework assessment for the Department of Geography and Environmental Science at the University of Southampton, receiving 3 stars out of 4 in 'quality that is world-leading in terms of originality, significance and rigour'. This work therefore directly contributed to the department's standing in assessing the quality of research in UK higher education institutions, helping to increase the university's prestige and ranking.

3.6.4 Intellectual contribution

CWR conceptualized the study framework and methods used throughout this thesis, and performed all analysis, preparations, and writing of the final thesis and associated manuscripts. CWR also generated all study visualizations and graphs throughout this work. AJT and ZM supervised CWR throughout study conceptualization and application, while other post-doctoral researchers and postgraduate students in the WorldPop research group provided feedback on methodological applications and procedures. Other co-authors and collaborators involved in publication of manuscripts comprising this work, along with their specific contributions, are listed individually within the 'Intellectual Contribution' section for each corresponding chapter.

Chapter 4: Equality in maternal and newborn health: Modelling geographic disparities in utilisation of care in five East African countries

4.1 Abstract

Geographic accessibility to health facilities represents a fundamental barrier to utilisation of maternal and newborn health (MNH) services, driving historically hidden spatial pockets of localized inequalities. Here, I examine utilisation of MNH care as an emergent property of accessibility, highlighting high-resolution spatial heterogeneity and sub-national inequalities in receiving care before, during, and after delivery throughout five East African countries. Specifically, I calculate a geographic inaccessibility score to the nearest health facility at 300 x 300 m using a dataset of 9,314 facilities throughout Burundi, Kenya, Rwanda, Tanzania and Uganda. Using DHS data, I utilise hierarchical mixed effects logistic regression to examine the odds of: 1) skilled birth attendance, 2) receiving 4+ antenatal care visits at time of delivery, and 3) receiving a postnatal health check-up within 48 hours of delivery. I apply model results onto the accessibility surface to visualise the probabilities of obtaining MNH care at both high-resolution and subnational levels after adjusting for live births in 2015.

Across all outcomes, decreasing wealth and education levels were associated with lower odds of obtaining MNH care. Increasing geographic inaccessibility scores were associated with the strongest effect in lowering odds of obtaining care observed across outcomes, with the widest disparities observed among skilled birth attendance. Specifically, for each increase in the inaccessibility score to the nearest health facility, the odds of having skilled birth attendance at delivery was reduced by over 75% (0.24; CI: 0.19 - 0.3), while the odds of receiving antenatal care decreased by nearly 25% (0.74; CI: 0.61 - 0.89) and 40% for obtaining postnatal care (0.58; CI: 0.45 - 0.75). Overall, these results suggest decreasing accessibility to the nearest health facility significantly deterred utilisation of all maternal health care services. These results demonstrate how spatial approaches can inform policy efforts and promote evidence-based decision-making, and are particularly pertinent as the world shifts into the Sustainable Goals Development era, where subnational applications will become increasingly useful in identifying and reducing persistent inequalities.

4.2 Introduction

Worldwide maternal deaths have been cut nearly in half over the past two and a half decades, largely due to a committed global effort to improve the lives and wellbeing of the world's most

vulnerable populations (Alkema et al., 2016a). Despite substantial progress among even the most disadvantaged subgroups, pregnancy remains risky for many of the world's women, and inequalities persist across the socioeconomic spectrum (WHO, 2015a). These risks are borne unequally both between and within countries, with pockets of relatively remote women in rural and poor communities bearing a disproportionate burden of morbidity and mortality (Alkema et al., 2016a). Historically, however, MNH policy has largely relied on aggregate, national-level statistics, which often mask these underlying spatial pockets of sub-national inequalities (Bhutta ZA and Reddy K, 2012). The United Nations therefore announced new SDGs to reduce these disparities and promote health and well-being for all, with specific targets to improve maternal and newborn health (MNH) outcomes among all women and children alike, and a particular emphasis on sub-national monitoring (United Nations General Assembly, 2015). As achieving these SDG targets will necessarily require effective use of limited resources, academics and policymakers alike have begun encouraging spatial disaggregation of health data (Countdown Core Group, 2014). Advances in computational geostatistical techniques and the increasing availability of geolocated data have increased the ability to produce these fine spatial resolution maps, critical for uncovering historically overlooked disparities. By monitoring health within a spatial framework, policy makers can better focus resources and intervention efforts amongst the most disadvantaged and marginalized populations, ensuring advancement of SDG targets in reducing inequalities among all (ICSU, ISSC, 2015).

Ultimately, achieving these targets and accelerating progress towards reducing adverse MNH outcomes requires understanding use of pregnancy-related services before, during, and after delivery, in addition to understanding the underlying disparities that drive observed utilisation patterns (Kerber et al., 2007). Specifically, the WHO has identified several critical determinants in reducing preventable pregnancy-related morbidity and mortality as: 1) access to antenatal care during pregnancy, 2) skilled birth attendance during delivery, and 3) postnatal care in the days and weeks following birth (World Health Organization, 2016b). Care seeking behaviour throughout the duration of a woman's pregnancy, however, is a complicated decision fraught with delays and barriers, including individual beliefs, societal norms, monetary barriers, and geographic access to necessary services (De Allegri et al., 2011).

Several studies have identified geographic accessibility specifically as a fundamental barrier in obtaining MNH care among developing nations, driving persistently high maternal and neonatal mortality rates (Ebener et al., 2015; Gething et al., 2012). This physical accessibility is driven by various geographic factors such as distance to the nearest health facility, topography of the local landscape, household transportation capacity, and road network infrastructure, all synergistically determining the extent and duration to which a woman can seek and obtain care (Jones-Webb and King-Schultz, 2008; Målqvist et al., 2010). Studies have therefore called for a nuanced understanding of accessibility as central to assessing maternal and newborn health and guiding

policy interventions (De Allegri et al., 2011; Ebener et al., 2015; Gething et al., 2012; United Nations, 2013).

The use of spatial statistics has become an increasingly recognised methodology to identify these hidden gaps, as geographic and spatial dynamics largely drive physical accessibility. These techniques can be used to extend an understanding of accessibility to produce policy relevant, high-resolution maps that can be used to focus resources where pregnancy is often riskiest (Ebener et al., 2015). Despite this, current spatial analyses of maternal and newborn health remain limited, and spatially explicit data currently available (such as geo-referenced DHS and Service Provision Assessment (SPA) surveys) remain underutilised (Gabrysch and Campbell, 2009a; Larmarange et al., 2011). While previous studies have examined the impact of geographic accessibility on a handful of MNH outcomes, few studies have explored accessibility as a determinant across the spectrum of pregnancy, particularly at a disaggregated level.

Here, I examined several MNH outcomes in the context of geographic accessibility to map the likelihood of receiving care before, during, and after delivery. Specifically, I utilised a georeferenced dataset of nearly 10,000 health facilities to generate high-resolution surfaces and administratively relevant maps reflecting the probability of a woman obtaining critical MNH care throughout the five East African Community partner states. By mapping disaggregated MNH outcomes as a function of geographic accessibility, outputs of these analyses have important policy implications in focusing future intervention efforts and allocation of resources, as well as providing a baseline from which to monitor future progress in the SDG era.

4.3 Methods

4.3.1 Overview

In this study, I modelled the probability of a woman receiving 4+ antenatal care (ANC) visits before delivery, skilled birth attendance (SBA) during delivery, and receiving postnatal care (PNC) within 48 hours of delivery, using DHS data throughout Burundi, Kenya, Rwanda, Tanzania and Uganda. I visualised these probabilities at both high-resolution (300 x 300 m grid cells) and policy relevant (administrative level 2, typically districts) scales to highlight spatial heterogeneity and sub-national inequalities in MNH care utilisation throughout the region of interest. Because of the importance of physical accessibility for care utilisation, my primary explanatory variable of interest was geographic inaccessibility to the nearest health facility.

To explore MNH utilisation in the context of geographic inaccessibility, I created a gridded travel impedance surface reflecting overall ease of traversing each 300 x 300 m square in the study region, given the topography and transport infrastructure of the region. I subsequently used this impedance surface to inform a cost-distance analysis, generating a geographic inaccessibility score

for each 300 m grid cell. These scores ranged from 0 (highly accessible) to 7 (highly inaccessible) and represented ease of access to the nearest health facility among a dataset of 9,314 facilities. I included this inaccessibility score as a covariate in a hierarchical mixed effects logistic regression model by overlaying DHS cluster locations on this generated accessibility surface and extracting scores for each cluster location. Finally, I used the resulting model of best fit to visualise probabilities of obtaining MNH care for each 300 x 300 m square throughout the region. Random effects included in the model consisted of DHS regions (N = 61) within the five countries, while fixed covariates included cluster-level (N = 3,311) effects of accessibility and rurality, and individual-level (N = 25,325) effects of wealth, education, and an interaction between age and total children delivered.

4.3.2 Data

4.3.2.1 Assembling individual-level MNH data

To explore variations in utilisation of MNH care across central East Africa, I obtained data from the most recent standard DHS for each country: Kenya (2014), Tanzania (2010), Uganda (2011), Rwanda (2010), and Burundi (2010) (Institut de Statistiques et d'Études Économiques du Burundi (ISTEEBU), Ministère de la Santé Publique et de la Lutte contre le Sida Burundi (MSPLS), and ICF International Inc, 2012; Kenya National Bureau of Statistics (KNBS) and ICF Macro, 2010; National Bureau of Statistics (NBS) Tanzania and ICF Macro, 2011; National Institute of Statistics of Rwanda (NISR), Ministry of Health (MOH) Rwanda, and ICF International Inc, 2012; Uganda Bureau of Statistics (UBOS) and ICF International Inc, 2012). Data were combined and processed using SAS v. 9.4 software, (SAS Institute Inc., 2013) with a total of 72,952 respondents between the five countries. For these analyses, I included women with a birth in the preceding five years, resulting in a total of 36,460 women. I obtained corresponding GPS coordinates for cluster locations via the DHS (for detailed methods, see http://dhsprogram.com/What-We-Do/GPS-Data-Collection.cfm) and mapped these using ArcGIS 10.2.2 software (Environmental Systems Research Institute, 2014).

To maintain participant confidentiality, the DHS randomly displaces GPS locations, with displacement diameters varying by urban (up to 2 km) and rural (up to 5 km, with 1% up to 10km) location (Burgert et al., 2013). To minimize displacement bias, I therefore drew corresponding buffers (circles placed around point locations with a specified diameter) of 2 km and 5 km around cluster locations to be used in later analyses, according to DHS guidelines (Perez-Heydrich et al., 2015). Only women with associated cluster locations were consequently included in further analyses, with 36,178 women among 3,311 clusters (see Figure A.1). Finally, to allow for subsequent model validation, I trained the model using 70% of the total sample with the remaining 30% set aside for model validation, resulting in 25,325 women used in the final model (see Appendix A for model validation details).

The University of Southampton Ethics and Research Governance approved secondary analysis of these data (ethics approval number 16918). Survey data used in these analyses are freely available via the DHS website, and participant confidentiality is outlined further at http://dhsprogram.com/What-We-Do/Protecting-the-Privacy-of-DHS-Survey-Respondents.cfm.

4.3.2.2 Assembling a database of health facilities

With input and collaboration from the intergovernmental East African Community (EAC) Secretariat, I obtained health facility data from ministries of health on over 19,000 mapped facility locations throughout the five EAC partner countries. These data included information for each health facility on operational status, private/public ownership, and type, and initially included dispensaries, maternity homes, district hospitals, health centres, and health posts. In the presented analyses, I included national, district, or regional hospitals, maternity homes, and health centres, and excluded dispensary facilities, as these facilities do not reliably offer comprehensive MNH care, including inpatient care critical for skilled birth attendance and postnatal care (Muga et al., 2005). After excluding dispensary facilities and cleaning the dataset to correct for duplicate entries or incorrect coordinates, I used a remaining 9,314 facilities in my cost-distance analyses. Finally, I imported this final list of health facilities into ArcGIS software and geo-located facility locations using latitude and longitude coordinates within the dataset to create a shapefile of health facility locations throughout the study countries (see Figure 4.1b).



Figure 4.1 Mapping inaccessibility to the nearest health facility. A) Impedance surface representing the difficulty in traversing a given 300 x 300 m cell. B) Health facility locations used as destinations. C) Accessibility surface generated via cost-distance analysis using inputs A) and B), representing the least accumulative geographic "cost" value for a 300 x 300 m cell in accessing the nearest health facility.

Chapter 4

4.3.3 Creating an impedance surface and inaccessibility score

I combined these health facilities with a gridded impedance surface representing difficulty of travel through a given cell to generate a geographic accessibility surface that reflected the most efficient, or least "cost" accumulative, pathway to the nearest health facility. I created this impedance surface for the study region by combining data on primary, secondary and tertiary road networks, permanent water bodies and river networks via DIVA-GIS (freely available at <u>www.diva-gis.org</u>), land cover data from the European Space Agency's 2009 GlobCover initiative, and elevation data from the Advanced Spaceborn Thermal Emission and Reflection Radiometer-Global Digital Elevation Model (ASTER-GDEM) Version 2 (*ASTER Global Digital Elevation Model Version 2*, 2011; *Global Land Cover Map (GlobCover)*, 2009). These surfaces were rasterized, or gridded, and consolidated to create a single surface where travel speeds could be assigned to each cell. Because consolidation of these surfaces necessitates identical cell size, I aggregated data to the resolution of the coarsest surface, thereby driving a 300 x 300 m resolution throughout further analysis.

To assign travel speeds to each 300 m cell, I built upon methods outlined previously by Alegana et al. 2012 (Alegana et al., 2012). While previous studies have modelled likely mode of transport used to access health facilities using sparse survey data and small area estimation approaches, (Gething et al., 2012) similar data were not available for all study countries used in these analyses. I therefore assumed mechanised transport on primary and secondary road networks, walking for offroad and tertiary networks, and that major water bodies were not traversable. For primary and secondary road networks, I assigned driving speeds of 80 and 60 km/hour, respectively. While maximum travel speeds for all five countries vary considerably, these speeds have been used in the literature previously, and were conservative estimates among the five countries (Alegana et al., 2012; Bureau for Industrial Cooperation, 2011). For cells not containing road networks, I classified land cover into broad categories such as tree, shrub or herbaceous cover, water body, and cultivated/managed or bare area. I then assigned these categories associated walking speeds, ranging from 2 km/hour (for desert area) to 5 km/hour (for cultivated or built areas). I inferred slope from elevation data using the Slope tool in ArcGIS software, and incorporated this into walking speeds based on Tobler's equation, which adjusts for increased walking speed on downslopes and decreased walking speed on up-slopes (Tobler, 1993). Permanent major rivers and other large water bodies represented a barrier to movement in these analyses, and were therefore designated a walking speed of 0 km/hour. For a detailed list of land cover classification and associated travel speeds, see Alegana et al. 2012 (Alegana et al., 2012). Finally, these discrete travel speed definitions were used to create standardized, ranked 'impedance' indices across the study region ranging from 1 to 7, where 1 represented the fastest travel speed (80 km/hr) and 7 represented the slowest speed (0 km/hr). The resulting impedance surface was then used in the cost-distance analyses (see Figure 4.1a).

Using this impedance surface, I created an overall inaccessibility score by performing a least accumulative cost-distance analysis using the Cost Distance tool in ArcGIS software. Briefly, this tool calculates the total geographic "cost"—a relative estimate of accessibility to a "source", in this case health facilities—for each 300 m square throughout the study region, given an input impedance (or "cost") surface. For each cell within the gridded raster, the pathway representing the least "cost" between a defined origin and defined destination is calculated; in general, "cost" may represent distance, geography, economic cost, or anything else the user defines. In my analyses, however, "cost" is defined by the created impedance surface, representing landscape, the built environment, and road infrastructure. The analyses performed here are similar, therefore, to Euclidean distance analyses (which simply measure shortest distance from origin to destination), with the added benefit of calculating the shortest weighted distance, incorporating factors such as bodies of water, landscape type, and use of road networks. This therefore measures not only distance, but also accessibility, between an origin (each 300m cell within the study area) and destination (health facility). Here, each square's calculated inaccessibility score, ranging from 0 (highly accessible) to 7 (highly inaccessible), represents the sum of the impedance values of traversed cells when traveling from that square to the facility with the lowest overall travel "cost". Therefore, higher scores on this index represent greater geographic "cost" and greater difficulty in reaching the nearest health facility.

Figure 4.1 outlines the inputs and output of this analysis, representing a) the impedance surface as described above (i.e., origin), b) facility locations (i.e., destination), and c) the resulting accessibility surface from the cost distance analysis. Using this accessibility surface, I subsequently overlaid buffered DHS cluster locations (as outlined in Section 3.3.2.1), and extracted the mean score throughout the buffers to include as an explanatory variable in exploring probabilities of obtaining MNH care. I chose to extract the mean score because DHS clusters can be located anywhere within the displacement radius, and therefore choosing one value (be it the minimum or maximum value within the buffer) is probabilistically not likely to reflect the actual underlying value of the true cluster location, thereby biasing results. To quantify this phenomenon common to geographical studies, Perez-Heydrich and colleagues performed a sensitivity study examining the ancillary raster extraction on DHS cluster locations, finding that simple point extraction does not bias results for continuous rasters, analyses extracting data from categorical raster surfaces are recommended to use buffer means (Perez-Heydrich et al., 2015). I therefore employed buffer means to mitigate this potential displacement bias, as the accessibility surface was a categorical raster ranging from 0 to 7. Lastly, because the impedance surface incorporates information on travel speed, the resulting accessibility surface units may be translated to travel time to the most accessible health facility. However, as the accessibility surface was used as an input covariate into the hierarchical model, this further translation was redundant and beyond the scope of this study.

4.3.4 Analysis of MNH outcomes

4.3.4.1 Hierarchical mixed effects modelling of MNH indicators

I used the inaccessibility score associated with DHS cluster locations to model the MNH outcomes of interest in the DHS data. The primary outcomes examined in these analyses included skilled birth attendance, antenatal care, and postnatal care, chosen through feedback from policy makers at the EAC organisation based on their programmatic relevance and impact. I modelled these outcomes using hierarchical mixed effects logistic regression using the 'lme4' package in R software (Bates et al., 2014, p. 4). Such multilevel analyses have been used previously in the literature with DHS data to account for the inherent nesting structure and multistage sampling design of the data (Magadi and Desta, 2011; Snijders and Bosker, 1999). I employed a random intercepts model for each outcome, allowing the intercept to vary among DHS regions (n = 61) throughout the study area:

$$y_{ij} = \beta_0 + \beta_1 x_{ij} + \beta_2 x_{ij} \dots + \beta_k x_{ij} + u_j + e_{ij}$$

where *y* is the dependent variable of interest, β_1 through β_k represent the array of independent variables of interest, *x* is the random effect of interest (in this case, DHS region), *u* is the group level effect, and *e* is the individual level error. In these models, I assumed u_j and e_{ij} are normally distributed with means 0 and variances σ_u^2 and σ_e^2 .

Figure 4.2 outlines the hierarchical levels existing in these analyses: the individual level, the cluster level, and the regional level. By using hierarchical, mixed effects-based model inference, spatial variation as a result of region-specific contextual factors (such as road infrastructure and health financing) can be captured in the model which otherwise would have been unaccounted for, and the inherent nested sampling structure employed through the DHS may be controlled for, regardless of significance of the random effect itself (Barr et al., 2013; Snijders and Bosker, 1999).





In these analyses, I defined skilled birth attendance as assistance during delivery by any doctor, nurse, or midwife (as defined by country-specific DHS observations), while antenatal care was defined based on WHO guidelines as 4+ antenatal visits during pregnancy, and postnatal care was defined as the respondent's first check-up occurring within 48 hours of delivery. For ANC and PNC, the DHS captures information on the most recent live birth, while SBA is recorded among all births in the preceding 5 years. Therefore, ANC and PNC represent care received for the most recent live birth a woman had, while SBA represents the proportion of all births a women had in the preceding 5 years which had a skilled attendant present. By examining SBA among all births in the preceding 5 years, I included all available data for the 37,309 total births occurring among the 25,325 women used in these analyses. Figure 4.2 outlines specific covariates used in the models, with individual level fixed effect covariates including DHS wealth quintile, highest level of education obtained, and an interaction between number of children delivered and respondent age. Lastly, DHS regions (N=61) within the five countries were included in the model as a random effect to account for unexplained spatial variation and region-specific differences.

4.3.4.2 High-resolution mapping of MNH care utilisation

I applied the results of the best fit model across the previously described accessibility surface to visualise the probability of a given birth receiving MNH care for each 300 x 300 m grid square. However, underlying population distributions and human settlements are highly heterogeneous and therefore variable in birth rates, limiting direct applicability of these probabilities. Therefore, to present a more accurate reflection of actual births at risk of not receiving these critical MNH services, I sought to incorporate information on the current number of live births occurring sub-

nationally for the year 2015, reflecting probability of a given birth obtaining MNH care in a more policy-relevant framework.

4.3.4.3 Estimating births-adjusted MNH care utilisation sub-nationally

The distribution of live births in the study region was obtained via the WorldPop project at a 100 m spatial resolution, freely available at www.worldpop.org (see Figure A.2) (UNFPA et al., 2014). Briefly, these data incorporate population demographics and satellite imagery data such as settlements, land cover, night-time lights, and sub-national age structure to model the distribution of women of childbearing age. Sub-national age-specific fertility rates, UN population projections, and estimates of abortions, stillbirths, and miscarriages are then used to model live births and pregnancies. Detailed methodology is outlined further in Tatem et al. 2014 (Tatem et al., 2014).

Using ArcGIS software, I adjusted the birth surfaces from 100 m to 300 m spatial resolution to match the resolution of the probability surface, multiplied the births and probability surfaces, and summed these values to the administrative unit level 2 to reflect actual number of births at-risk for each outcome. I then calculated the births-adjusted probability for each administrative level 2 unit throughout the region by dividing these at-risk births with total births in the administrative unit, as obtained from WorldPop. By incorporating data on live births, I present maps reflecting actual fertility rates, accounting for age structure, urban/rural differences, stillbirths, miscarriages, etc.

4.4 Results

Figure 4.3 represents the distribution of skilled birth attendance, no antenatal care visits, and postnatal care within 48 hours of delivery for the study area by DHS region, for three years preceding the survey. These thematic maps represent weighted estimates by DHS region, as compiled through the DHS's STATcompiler feature, with specific estimates outlined in Table 4.1 (ICF International, 2015). DHS regions varied by indicator, with skilled birth attendance and postnatal care ranging from just under 25% skilled attendance at delivery in Pemba North, Tanzania to over 90% in Bujumbura Mairie, Burundi, and under 10% of women receiving a postnatal care check-up within 48 hours of delivery in Mwanza, Tanzania to over 70% in the Central DHS region in Kenya. Those receiving no antenatal care visits at all were far less than the previous two outcomes, with many DHS regions reporting 0% of women with no ANC visits, to a high of nearly 25% of women receiving no ANC visits in the North Eastern region of Kenya. In general, the North Eastern region of Kenya represented a consistently 'at-risk' region, with 23.8% of women having no antenatal visits, 35.3% of women having skilled birth attendance at delivery, and only 14.9% of women receiving a postnatal care check-up within 48 hours of delivery (Figure 4.3).

DHS Year	DHS Region	No ANC	SBA	PNC	DHS Year	DHS Region	No ANC	SBA	PNC
Burundi				Lindi	0.0	55.2	31.8		
	Bujumbura								
2010	Mairie	1.7	92.9	44.4		Mtwara	0.0	64.6	32.7
	North	0.5	60.3	28.3		Ruvuma	0.7	77.1	56.7
	Centre-East	0.9	60.4	25.0		Iringa	3.8	77.6	54.3
	West	1.2	70.6	35.1		Mbeya	4.9	42.1	25.6
	South	0.9	66.2	29.5		Singida	1.6	47.8	29.3
	TOTAL	0.9	65.0	29.8		Tabora	5.9	46.3	28.6
Kenya						Rukwa	2.0	29.5	13.6
2014	Central	1.9	90.2	71.9	2010	Kigoma	0.0	32.5	14.8
	Coast	2.2	61.1	49.5		Shinyanga	3.4	36.6	22.3
	Eastern	2.8	66	61.1		Kagera	0.0	56	18.4
	Nairobi	1.6	91.1	71.6		Mwanza	0.5	40.6	8.6
	North				2010				
	Eastern	23.8	35.3	14.9		Mara	1.8	30.9	12.6
	Nyanza	2.2	68.3	61.0		Manyara	4.5	42.3	25.8
	Rift Valley	6.4	53.5	45.9		Zanzibar North	0.2	43.6	13.6
						Zanzibar	0.2	1510	10.0
	Western	2.1	54.6	34.6		South	0.0	68.4	43
	TOTAL	4.0	64.5	52.9		Town West	0.0	77.0	41.7
Rwanda						Pemba North	0.6	24.3	12.6
2010						Pemba			
	Kigali	1.1	86.0	24.8		South	1.1	42.0	25.2
	South	2.0	75.9	23.8		TOTAL	1.9	49.9	25.8
	West	1.5	77.4	13.2	Uganda				
	North	1.3	73.1	17.9		Kampala	1.9	92.2	61.2
	East	1.9	77.1	13.6		Central 1	10.8	60.6	39.4
	TOTAL	1.7	77.1	17.6		Central 2	3.6	70.5	39.3
Tanzania						East			
	10				2011	Central	8.1	67.7	24.5
2010	Dodoma	2.8	46.4	29.9		Eastern	5.3	58.1	31.2
	Arusha	3.2	46.1	25.5		North	1.1	57.4	27.8
	Kilimanjaro	0.0	87.9	43.7		Karamoja	2.0	31.1	26.8
	Tanga	1.1	42	25.9		West Nile	2.0	62.0	40.5
	Morogoro	0.9	58.3	20.2		Western	2.5	52.6	24.1
	Pwani	0.0	74.7	35.8		South West	0.8	46.8	18.8
	Dar es Salaam	0.0	88.1	43.0		TOTAL	4.1	59.5	33.0

Table 4.1 MNH outcomes within the study area disaggregated by sub-national DHS region and national total, gathered through STATcompiler (ICF International, 2015)



Figure 4.3 Prevalence of skilled attendance at delivery, no antenatal care, and postnatal care provided within 48 hours of delivery, by DHS region

4.4.1 Gridded accessibility surface

Figure 4.1c presents results of the cost-distance analysis, and represents geographic accessibility to the nearest health facility for a given 300 x 300 m square. Unsurprisingly, health facility distribution predominantly drives patterns in the accessibility surface, with remote areas lacking facilities representing the most inaccessible throughout the region, regardless of landscape topography. This result visually highlights the spatial patterns in accessibility emerging from the heterogeneous placement of major health facilities. Of note, the area of lowest accessibility observed in northern Tanzania and southern Uganda is a result of Lake Victoria, represented by red in Figure 4.1a. However, because neither health facilities nor DHS data are located in this area, results remain unaffected by this artefact.

4.4.2 Explanatory variables of MNH care utilisation

Table 4.2 presents modelled odds ratios and 95% confidence intervals for each MNH outcome. Across all outcomes, increasing wealth and education was associated with greater odds of obtaining MNH care, with the most disparities between groups observed among skilled birth attendance. Specifically, those in the highest wealth quintile had over 4 times odds of delivering with a SBA as compared to those in the lowest wealth quintile (4.71; CI: 4.22 - 5.25), while those with higher education had over 7 times the odds of having a SBA present at delivery (7.63; CI: 5.91 - 9.84). In contrast, those in the richest quintile had only a 1.5 to 2 times increased odds of obtaining ANC (1.52; CI: 1.36 - 1.70) and PNC (1.89; CI: 1.67 - 2.14) as compared to those in the poorest quintile, while those with the highest education had a 2 to 3 times increased odds of obtaining ANC (2.82; CI: 2.35 - 3.39) and PNC (1.92; CI: 1.61 - 2.28) as compared to those with no education.

	SBA	ANC	PNC			
Fixed Effects	OR (95% CI)					
Age						
15 to 20	Ref	Ref	Ref			
20 to 30	0.86 (0.69, 1.08)	1.28 (1.01, 1.61)	1.09 (0.84, 1.43)			
30 to 40	0.54 (0.42, 0.69)	1.38 (1.07, 1.77)	1.15 (0.86, 1.53)			
40 to 50	0.43 (0.29, 0.62)	1.09 (0.74, 1.6)	0.91 (0.59, 1.41)			
Total # children delivered	0.64 (0.57, 0.72)	0.85 (0.73, 1)	0.87 (0.73, 1.04)			
Wealth quintile						
Poorest	Ref	Ref	Ref			
Poorer	1.41 (1.32, 1.51)	1.10 (1.02, 1.20)	1.28 (1.15, 1.41)			
Middle	1.78 (1.65, 1.92)	1.18 (1.08, 1.29)	1.43 (1.29, 1.59)			
Richer	2.50 (2.31, 2.71)	1.36 (1.24, 1.49)	1.69 (1.52, 1.88)			
Richest	4.71 (4.22, 5.25)	1.52 (1.36, 1.70)	1.89 (1.67, 2.14)			
Education						
No education	Ref	Ref	Ref			
Primary	1.61 (1.51, 1.72)	1.04 (0.96, 1.12)	1.33 (1.21, 1.45)			
Secondary	3.04 (2.75, 3.37)	1.29 (1.16, 1.43)	1.63 (1.44, 1.84)			
Higher	7.63 (5.91, 9.84)	2.82 (2.35, 3.39)	1.92 (1.61, 2.28)			
Residence						
Urban	Ref	Ref	Ref			
Rural	0.69 (0.64, 0.75)	1.00 (0.92, 1.08)	0.90 (0.82, 0.98)			
Accessibility to nearest facility	0.24 (0.19, 0.30)	0.74 (0.61, 0.89)	0.58 (0.45, 0.75)			
Age x Total # Children Delivered						
15 to 20	Ref	Ref	Ref			
20 to 30	1.19 (1.05, 1.34)	1.05 (0.90, 1.22)	1.01 (0.84, 1.21)			
30 to 40	1.39 (1.23, 1.57)	1.08 (0.93, 1.26)	1.06 (0.88, 1.27)			
40 to 50	1.44 (1.27, 1.64)	1.15 (0.98, 1.35)	1.10 (0.92, 1.33)			
Random Effects	SBA	ANC	PNC			
	Variance (SD)					
DHS Region	0.1923 (0.4385)	0.1422 (0.3770)	0.2248 (0.4741)			

Table 4.2 Hierarchical mixed effects logistic regression model odds ratios of MNH outcomes among female DHS respondents in five East African countries (N = 25,325)

Living in rural areas was also associated with decreased odds of obtaining MNH care across outcomes. Specifically, those living in rural areas were 30% less likely to deliver with a skilled birth attendant present (0.69; CI: 0.64 - 0.75) and 10% less likely to obtain PNC after delivery (0.90; CI: 0.82 - 0.98) as compared to those living in urban areas. However, while living in rural areas was also associated with decreased odds of receiving 4+ ANC visits, this was not significant (1.00; CI: 0.92 - 1.08). Lastly, I found age and total number of children delivered were highly correlated, in line with previous studies (Magadi et al., 2007). Because both age and parity are important indicators in determining service use with sometimes opposing directionality, I included an interaction term in the model to control for the relationship between number of children delivered and respondent's age. This interaction term improved model fit and was highly significant in explaining skilled birth attendance (p<0.0001), with older women experiencing a greater increase in odds of SBA per child delivered as compared to younger women, and is in line

with previous findings (Gabrysch and Campbell, 2009a). While similar patterns were observed among ANC and PNC outcomes, these effects were not significant.

Finally, increasing geographic inaccessibility to the nearest health facility was associated with the greatest reduction in odds of utilising MNH care, particularly among skilled birth attendance. Specifically, for each unit increase in inaccessibility to the nearest health facility, the odds of having a skilled birth attendant present at delivery was reduced by over 75% (0.24; CI: 0.19 - 0.30), while odds of receiving 4+ ANC visits decreased by nearly 25% (0.74; CI: 0.61 - 0.89) and 40% for obtaining PNC (0.58; CI: 0.45 - 0.75). Due to how I calculated inaccessibility scores (as outlined in Methods), these scores do not represent a directly interpretable index of travel time/difficulty, but quantify distance in grid cells to the nearest health facility, scaled by the relative speed of traversing each grid cell type. Overall, these results suggest decreasing accessibility to the nearest health facility significantly deterred utilisation of all maternal health care services.

4.4.3 High-resolution mapping of MNH care utilisation

I applied coefficients from the resulting model of best fit to the previously described accessibility surface to generate high-resolution maps reflecting probability of obtaining MNH care as an emergent effect of accessibility. Figure 4.4 presents boxplots for these distributions, while Figure 4.5 presents high-resolution probability surfaces among all MNH outcomes. The probability of having a skilled birth attendant during delivery exhibited the greatest amount of variability across the study region (Figure 4.4), with an average probability of 58%, ranging between nearly 0% and 75% for a given 300 x 300 m square. The ranges of receiving ANC and PNC were lower than SBA, with a mean probability of 37% (ranging from 9% to 40%) and 18% (ranging from <1% to 22%), respectively.



Figure 4.4 Boxplots of modelled probabilities of skilled birth attendance (SBA), antenatal care (ANC), and postnatal care (PNC).

Figure 4.5 shows the probability of MNH outcomes at 300 x 300 m, representing spatial heterogeneity in rates of obtaining care as a function of accessibility. As previously discussed, the ranges of probabilities varied widely between outcomes, making direct comparison between probability surfaces difficult due to the varying scales used between subfigures. Instead, this figure shows variations in the emergent spatial patterns resulting from geographic accessibility between outcomes, with the most drastic effect of space observed among SBA, as represented by more contained pockets of higher SBA probability surrounding health facility locations. This spatial effect becomes coarser for ANC and PNC utilisation, with relatively similar patterns observed between the two, despite varying probability distributions (Figure 4.5).

Chapter 4



Figure 4.5 Modelled probability surfaces representing the spatial effect of accessibility at 300 x 300 m. a) Delivery with a skilled birth attendant (SBA) present, b) Four or more antenatal care (ANC) visits at time of delivery, and c) Postnatal care (PNC) received within 48 hours of delivery.

4.4.4 Sub-national mapping of MNH care utilisation

To facilitate policy relevance, I aggregated high-resolution probability surfaces to an administratively relevant scale, and adjusted these values using the surface of births per 300 m grid square, as previously described above. Figure 4.6 therefore represents the probability of: a) having a skilled attendant present during delivery, b) obtaining 4+ ANC visits by time of delivery, and c) the woman receiving PNC within 48 hours of delivery, respectively. Because I adjusted probabilities to reflect actual births at-risk, these maps represent the mean probability of obtaining MNH for a given birth within each geographic unit, accounting for where live births are most likely to occur.



Figure 4.6 Births adjusted probability maps representing the probability of obtaining MNH care for a given birth at the administrative II unit. A) Delivery with a skilled birth attendant (SBA) present, B) Four or more antenatal care (ANC) visits at time of delivery, and C) Postnatal care (PNC) received within 48 hours of delivery.

Similar to the high-resolution surfaces, the observed ranges of probabilities varied between countries, with Kenya and Tanzania predominantly driving the scale of heterogeneity across outcomes. Regionally, the lowest probabilities of receiving MNH care occurred throughout northern Kenya and central Tanzania across outcomes. These probabilities generally tended to be higher in urban versus rural districts across countries and outcomes. Conversely, my model results showed Rwanda and Burundi to have consistently higher probabilities of a given birth receiving MNH care, as compared to Uganda, Tanzania and Kenya. Further, the range of these probabilities between administrative units varied less, suggesting less inequality between districts.

4.4.5 Model validation

To test the regression model's prediction power for utilisation of skilled birth attendance and antenatal/postnatal care, I performed a receiver operating characteristics (ROC) analysis. This analysis compares predicted prevalence to observed, and is used widely within diagnostic evaluation to validate regression models (Brooker et al., 2002). This method has been widely used within epidemiology and ecology, with recent applications in validating regression models within a spatial context (Clements et al., 2006a; Noor et al., 2008). The statistic used to evaluate ROC curves is known as the area under the curve (AUC), or a plot of sensitivity (or the ability to detect true positives) by 1-1 – specificity (or the ability to detect true negatives) (Clements et al., 2006a). The AUC statistic therefore represents the discriminatory power of the model in predicting true prevalence of a threshold, representing a trade-off between sensitivity and specificity (Fawcett,

2006). In general, higher AUC values represent increasing discriminatory ability of the model, or the ability to correctly classify predicted observations as receiving or not receiving a given service, with 0.9 representing a well-fitting model, 0.7 moderately-fitting model and 0.5 representing no better fit than what would be expected through random sampling (Clements et al., 2006a).

Figure 4.7 outlines the results of the ROC analysis, and provides metrics of the AUC-ROC, with the skilled birth attendance model performing relatively better than the ANC and PNC models. With an AUC value of 0.65, this model falls around values expected for a moderately well-fitting model, indicating a 65% probability that a woman observed with SBA had a higher estimated probability of SBA than a woman without SBA, for a random pair of women with and without the outcome (Clements et al., 2006a). The models representing antenatal care and postnatal care, however, were substantially less well-fitting, with AUC values of 0.55 and 0.57, respectively. While these models are better than what would be expected by random chance, these results suggest different factors may be driving SBA utilisation, as compared to ANC and PNC, and that the latter two outcomes should be explored in further detail to better predict their use.



Figure 4.7 ROC curves for skilled birth attendance (SBA), antenatal care (ANC), and postnatal care (PNC) models, with associated area under the curve (AUC) metrics.

4.5 Discussion

Spatial inequalities in utilisation of MNH care continue to persist among low- and middle-income countries, particularly among skilled birth attendance and antenatal care coverage, and targeting

spatial pockets of low utilisation will be critical for future health interventions (WHO, 2015a). A nuanced understanding of how geographic accessibility influences uptake of MNH care at a very fine spatial resolution will be key to identify and reduce these inequalities and alleviate coverage gaps. Here, I have highlighted the emergent spatial patterns of MNH care resulting from geographic accessibility at a high-resolution scale, and presented probabilities of obtaining MNH care for a given birth at the sub-national level. The spatial patterns revealed have important policy implications for informing allocation of future intervention efforts to target the most disadvantaged women and riskiest pregnancies. As these analyses highlight areas of low geographic accessibility, the presented results could help target not only health-related resources, but also future development more generally, including road networks and other transportation-related infrastructure.

Overall, I found that disparities exist in obtaining MNH care across wealth, education, and levels of rurality, and that decreased accessibility to the nearest health facility resulted in the widest disparities in obtaining care across the spectrum of pregnancy (Table 4.2). These findings are in line with previous studies, further establishing that barriers in deciding how, when, and even whether to seek care are greatest for uneducated, remote women in poverty (Barros et al., 2012; De Allegri et al., 2011; Gething et al., 2012). Regionally, I found that Kenya and Tanzania had the strongest patterns of spatial heterogeneity in the observed outcomes and generally lower probabilities of obtaining all types of care, with the lowest probabilities observed throughout rural districts in northern Kenya and central Tanzania. Conversely, I found Rwanda and Burundi to have generally higher probabilities of obtaining care, as compared to Tanzania, Kenya, and Uganda. This trend could be due in part to the relative density of facilities available, while more remote areas of Kenya and Tanzania had comparatively less facilities and primary or secondary road networks. This pattern also occurred sub-nationally, as I observed higher probabilities of obtaining care in urban versus rural districts, indicating infrastructure density is important in increasing MNH care coverage.

These findings suggest the allocation of funds focused on supporting increased infrastructure, such as buildings, equipment, and health workers, should appropriately reflect the demography and epidemiology of the area. These analyses could help to direct the flow of such resources, by highlighting areas and populations where care utilisation rates are lowest. While such infrastructure can improve geographic impedance and therefore accessibility, placement of primary health facilities offering comprehensive obstetric care or maternity waiting homes in these particularly resource-poor areas will ultimately be critical in reducing adverse MNH outcomes such as maternal and neonatal mortality (Gabrysch et al., 2011). Specifically, my results highlight northern Turkana, Samburu, Marsabit, Wajir, and Mandera districts in Kenya and central Rufiji and Liwale districts in Tanzania as key targets for increased coverage of facilities offering maternity care and construction of better road networks and transportation services. However, because these analyses did not

capture information on use of lower tier facilities, referral networks, or community health workers linked to larger health facilities, MNH care utilisation in these areas may artificially be low. Therefore, to promote a more accurate representation of facility coverage in these areas, future data collection efforts should focus on areas of low geographic accessibility, capturing services provided and quality of services.

Future research should explicitly compare health systems throughout these countries, particularly in Rwanda and Burundi, to understand why MNH care performance is comparatively better in some countries as opposed to others. In particular, previous studies have noted that while MNH outcomes such as neonatal mortality are positively correlated with distance to closest health facility, health facility placement may already be saturated in countries that have prioritised improving facility coverage (Målqvist et al., 2010). In such cases, development of existing infrastructure, including continued support and education opportunities for community health workers, improved road networks and public transportation, and increased availability of comprehensive obstetric facilities or maternity waiting homes may ultimately prove effective in alleviating geographic accessibility as a barrier to accessing maternal health services. Finally, to promote a more nuanced and accurate understanding of geographic accessibility, future data collection efforts should aim to capture information on specific health facility used during pregnancy and birth, mode of transportation used to access these facilities, as well as self-reported travel time to these facilities.

4.5.1 Limitations

These analyses are subject to several limitations. Firstly, by excluding dispensaries from the analyses, it is possible that I incorrectly identify some women as having more difficulty in obtaining MNH care than is actually the case. However, I chose to be overly conservative in my modelling efforts by excluding these facilities, to avoid incorrectly assuming that a woman would have access to MNH services that are not actually there. Secondly, I use self-reported MNH outcomes and definitions, which may vary from country to country and may be subject to recall bias. This limitation explicitly impacts skilled birth attendance analyses, as some women may be unaware of what type of attendant was present at delivery, and countries may further define "skilled attendants" differently. While I limited my analyses of skilled attendance only to doctors, nurses and midwives, it is possible countries may define nurses and midwives variably. Future studies should ensure that definitions can be standardized on a multi-national level, and examine impacts of these definitions on model results. Thirdly, I assumed in my analyses that women would obtain care by travelling to the nearest health facility to their house. This often does not occur in reality, however, and is influenced by a host of individual and systematic factors, which may include, among others, the use of multiple facilities to obtain care via referral networks, the quality of care provided at the facility, and individual perception of the facility (De Allegri et al., 2011). Future work should more explicitly examine respondents' actual use of health facilities, and
explore key factors explaining where a woman decides to obtain care. Finally, my analyses were limited temporally by survey availability. Specifically, with data ranging over a 4-year span throughout the study countries, and gathering information on births in the previous 5 years, it is possible, that these results may no longer reflect the current state of maternal and newborn health in some countries. Future analyses should examine change in utilisation of MNH care over time, and use more recent datasets where available.

4.6 Conclusions

Inequalities in obtaining MNH care continue to persist, despite progress in increasing coverage and availability amongst the most vulnerable subgroups in the world. Spatial disaggregation of MNH data and a nuanced understanding of the geographic processes driving these disparities will be critical to continue this progress and accelerate achievement of SDG goals to reduce health disparities among all. Here, I spatially modelled the probability of several MNH outcomes at both high-resolution and policy relevant scales to highlight spatial patterns in accessibility and subnational inequalities in MNH care utilisation throughout central East Africa. I found that disparities exist across the socioeconomic spectrum, with the widest disparities observed in geographic accessibility to health facilities, particularly among skilled birth attendance. The results of these analyses demonstrate how spatial approaches can be used to measure and identify spatial pockets of historically overlooked inequalities, thereby strategically informing policy efforts and promoting evidence-based decision-making. These findings are particularly pertinent to the East African Community in its efforts to accelerate progress in women's, children's and adolescent's health and equity within the framework of the SDGs.

4.7 Acknowledgements

This research was a collaborative effort with the EAC Secretariat through the EAC Open Health Initiative (OHI) agenda of strengthening accountability for results and resources for women and children's health, funded through the Norwegian Development Agency (NORAD) and ICS Integrare. CWR is supported by the University of Southampton's Economic and Social Research Council's Doctoral Training Centre. AJT was supported by funding from NIH/NIAID (U19AI089674), the Bill & Melinda Gates Foundation (OPP1106427, 1032350, OPP1134076), the Clinton Health Access Initiative, the RAPIDD program of the Science & Technology Directorate, Department of Homeland Security and the Fogarty International Center, National Institutes of Health, and a Wellcome Trust Sustaining Health Grant (106866/Z/15/Z). CRB was supported by the United States Agency for International Development (USAID) through the DHS Program (AID-OAA-C-13-00095). The funders had no role in study design, data collection and analysis, decision to publish, and preparation of the manuscript.

4.8 Intellectual contribution

The author list for this published work is as follows: Ruktanonchai CW, Ruktanonchai NW, Pezzulo C, Alegana VA, Bosco C, Nove A, Lopes S, Bytyqi A, Ayiko R, Charles A, Lambert N, Msechu E, Matthews Z, Tatem AJ. CWR conceived of the study framework and methods for this study, performed all data management and analysis, generated all study visualizations and results interpretations, and wrote and prepared the manuscript for journal submission. NWR, CB, CP, and VA contributed insight and feedback into the statistical and geographical analysis for this study. AN, SL, RA, and AC provided feedback on broad implications of study findings. ZM and AJT supervised CWR during production of this manuscript, and oversaw that methods and analysis were being executed within a sound and scientifically rigorous framework. RA, AC, NL, EM, and EK were involved with providing resources for this study, including country-specific datasets. All authors reviewed the manuscript and provided feedback before final publication.

Chapter 5: Temporal trends in spatial inequalities of maternal and newborn health services among four East African countries, 1999 – 2015

5.1 Abstract

Sub-Saharan Africa continues to account for the highest regional maternal mortality ratio (MMR) in the world, at just under 550 maternal deaths per 100,000 live births in 2015, compared to a global rate of 216 deaths. Spatial inequalities in access to life-saving maternal and newborn health (MNH) services persist within sub-Saharan Africa, however, with varied improvement over the past two decades. While previous research within the East African Community (EAC) region has examined utilisation of MNH care as an emergent property of geographic accessibility, no research has examined how these spatial inequalities have evolved over time at similar spatial scales.

Here, I analyse temporal trends of spatial inequalities in utilisation of antenatal care (ANC), skilled birth attendance (SBA), and postnatal care (PNC) among four East African countries. Specifically, I use Bayesian spatial statistics to generate district-level estimates of these services for several time points using Demographic and Health Surveys data in Kenya, Tanzania, Rwanda, and Uganda. I examine temporal trends of both absolute and relative indices over time, including the absolute difference between estimates, as well as change in performance ratios of the best-to-worst performing districts per country.

Across all countries, I find the greatest spatial equality in ANC, while SBA and PNC tended to have greater spatial variability. In particular, Rwanda represents the only country to consistently increase coverage and reduce spatial inequalities across all services. Conversely, Tanzania has noticeable reductions in ANC coverage throughout most of the country, with some areas experiencing as much as a 55% reduction. Encouragingly, however, I find that performance gaps between districts have generally decreased or remained stably low across all countries, suggesting countries are making improvements to reduce spatial inequalities in these services.

I find that while the region is generally making progress in reducing spatial gaps across districts, improvement in PNC coverage has stagnated, and should be monitored closely over the coming decades. This study is the first to report temporal trends in district-level estimates in MNH services across the EAC region, and these findings establish an important baseline of evidence for the Sustainable Development Goal era.

5.2 Introduction

Substantial progress has been made in reducing global maternal mortality over the past three decades, and while total coverage of maternal health services may have increased over time, inequalities among those utilising these services have persisted (Boerma et al., 2008; Nguhiu et al., 2017). Sub-Saharan Africa continues to account for the highest regional maternal mortality ratio (MMR) in the world, at just under 550 maternal deaths per 100,000 live births in 2015, compared to a global rate of 216 deaths. Further, only four countries achieved the 75% reduction (Cabo Verde, Equatorial Guinea, Eritrea, and Rwanda) set out by Millennium Development Goal (MDG) 5a (WHO et al., 2015). These ratios mask underlying heterogeneity, however, with MMRs ranging from as low as 167 per 100,000 live births in Southern Africa, to as high as 675 deaths per 100,000 live births in Western Africa in 2015 (WHO et al., 2015). The greatest reduction in MMR between 1990 and 2015 occurred within the Eastern Africa sub-region (as defined by the United Nations' (UN) MDG groupings), with a 57% overall change and 3.4% average annual change (WHO, 2015a). However, even within this sub-region, countries falling in the East African Community (EAC) region (comprised in 2015 of Burundi, Kenya, Rwanda, Tanzania, and Uganda, with the addition of South Sudan in 2016) experienced varied improvement in preventing maternal deaths over the past two decades, with Rwanda representing the only country within the region to reach the MDG 5a target.

Reducing health inequalities in low- and middle-income countries has become an increasingly important and quantifiable objective in the post-2015 Sustainable Development Goals (SDG) agenda (Barros and Victora, 2013; WHO, 2015b). While there has been a renewed call for disaggregation of national level indicators, much research in maternal and newborn health (MNH) over the previous decades has focused on disaggregating health data by socioeconomic status such as wealth quintile and education status (Assaf and Pullum, 2016; Boerma et al., 2008; WHO, 2015a). The state of a woman's health depends not only on her education and income, however, but also on where she lives and the progress her country is making in addressing maternal and newborn health issues (WHO, 2015a). Over the coming decades, therefore, spatial and temporal disaggregation in addition to this continued socioeconomic disaggregating data across spatial, temporal and socioeconomic dimensions, health inequalities amongst subgroups may be highlighted, as well as how these inequalities have changed over time (WHO, 2015a).

Preventing the deaths of women and newborns ultimately requires delivery of timely and high quality service packages and interventions across the continuum of care, such as antenatal care, skilled birth attendance, and postnatal care (Kerber et al., 2007). Previous studies have examined temporal trends of child mortality (Golding et al., 2017) and health indicators such as education (Graetz et al., 2018) and child growth (Osgood-Zimmerman et al., 2018) at high spatial resolutions,

but fewer studies have examined temporal trends in utilisation of key MNH services at similar spatial resolutions. Victora and colleagues (Victora et al., 2016) reported progress in maternal, newborn, and child survival through the Millennium Development Goal era, while more recently, child and maternal mortality estimates have been systematically reported at the global, regional and national scales (Alkema et al., 2016a; Black et al., 2010). Assaf and Pullum (Assaf and Pullum, 2016) further reported temporal trends in key MNH services disaggregated by socioeconomic indicators such as wealth and education; these studies, however, represent analysis performed at a provincial or national-level spatial scale, and potentially mask important variation at policy-relevant spatial scales, such as the district level.

Within the EAC region, previous research (C. W. Ruktanonchai et al., 2016) has examined utilisation of MNH services as an emergent property of accessibility, highlighting high-resolution inequalities in receiving care before, during, and after delivery. While these inequalities have been spatially disaggregated, no research has examined how these spatial inequalities have evolved over time at similar spatial scales. Further, estimates at a policy-meaningful level such as the district level remain limited, as data collected in-country tend to be limited by insufficient reporting, sampling bias, or other methodological challenges. Estimates of MNH service utilisation have historically relied on household surveys such as the Demographic and Health Surveys (DHS), but these are not typically representative below the national or regional level due to sample design (with some recent exceptions such as the 2015 Kenya DHS). This necessitates the use of spatial statistics, such as small area estimation techniques and geo-spatial modelling frameworks, to generate predictive estimates at smaller spatial levels, but these approaches also come with methodological considerations. Spatial interpolation methods using a Bayesian framework are one such application of predictive modelling, and represent an ideal opportunity to quantify uncertainty in estimates through posterior distributions (Gething et al., 2015). Briefly, a Bayesian framework generates a distribution of possible estimates, or posterior distribution, in which the "true" value may lie, and allows for reporting of standard distribution statistics, such as mean, median, standard deviation, and credibility interval. This is particularly valuable when reporting DHS data at a spatial scale different from that which the data were collected at, where the range of uncertainty may vary in more rural, less sampled areas (Gething et al., 2015; Neal et al., 2019).

Here, this study examines how prevalence of antenatal care (ANC), skilled birth attendance (SBA), and postnatal care (PNC) use has changed sub-nationally over time within the EAC region, using both absolute and relative measures of inequality. Barros and Victora (Barros and Victora, 2013) argue that reporting temporal trends of both absolute and relative measures of inequalities is essential, as these measures complement each other to provide a more comprehensive reflection of change in inequality over time. This paper therefore aims to generate estimates of ANC, SBA, and PNC ranging between 1999 and 2015 at a higher spatial resolution than has been previously reported. I examine temporal trends of absolute indices by visualising the difference in these

estimates between the first and last time points available. I further aim to examine temporal trends of relative indices by quantifying how the gap between the best-to-worst performing administrative units has changed over time, as well as explore how the effect of space alone has changed in predicting utilisation of these services.

5.3 Methods

5.3.1 Data

To explore sub-national change in ANC, SBA, and PNC over time, I compiled data from DHS for Kenya, Tanzania, Rwanda, and Uganda for several time points available (see Table 5.1) using SAS version 9.4 software (SAS Institute Inc., 2013). To calculate estimates using DHS data at a spatial area smaller than those which are reported through the DHS program, information on spatial location of household surveys are necessary through global positioning system (GPS) coordinates (Gething et al., 2015). The DHS program provides this information for recent surveys at the cluster (an aggregate of households) level, which is then displaced to maintain participant confidentiality. To facilitate spatial interpolation, I therefore included only standard DHS surveys in these analyses with corresponding geo-located cluster data available. Of note, at the time these analyses were performed, Burundi contained a full DHS survey with associated GPS data for only one year, and therefore was not included in these analyses. I further restricted these analyses to women with births in the previous five years to generate estimates of MNH services. Table 5.1 displays the DHS survey characteristics, final sample size, and number of clusters used in these analyses. I mapped cluster locations using ArcGIS software (Environmental Systems Research Institute, 2014) and drew corresponding buffers of 2km and 5km around urban and rural locations (respectively) in order to minimize bias resulting from DHS displacement protocols, in accordance with DHS recommendations outlined by Burgert and colleagues (Burgert et al., 2013).

Country	Citation	Survey Year	Survey Type	Clusters	Sample
Kenya	(Central Bureau of Statistics - CBS/Kenya et al., 2004)	2003	Standard DHS-IV	399	2,969
	(Kenya National Bureau of Statistics - KNBS et al., 2010)	2008/9	Standard DHS-V	397	3,054
	(Kenya National Bureau of Statistics et al., 2015)	2014	Standard DHS-VII	1,585	11,151
Rwanda	(Institut National de la Statistique du Rwanda - INSR and ORC Macro, 2006)	2005	Standard DHS-V	462	4,002
	(National Institute of Statistics of Rwanda - NISR et al., 2012)	2010	Standard DHS-VI	492	4,746
	(National Institute of Statistics of Rwanda et al., 2016)	2014/15	Standard DHS-VII	492	4,467
Tanzania	(National Bureau of Statistics/Tanzania and Macro International, 2000)	1999	Standard DHS-IV	176	1,504
	(National Bureau of Statistics - NBS/Tanzania and ICF Macro, 2011)	2010	Standard DHS-VI	475	4,019
	(Ministry of Health et al., 2016)	2015/16	Standard DHS-VII	608	5,288
Uganda	(Uganda Bureau of Statistics - UBOS and ORC Macro, 2001)	2000/1	Standard DHS-IV	266	2,790
	(Uganda Bureau of Statistics - UBOS and Macro International, 2007)	2006	Standard DHS-V	336	3,420
	(Uganda Bureau of Statistics - UBOS and ICF International, 2012)	2011	Standard DHS-VI	400	3,645

Table 5.1 DHS survey used in study analysis and characteristics.

Finally, to allow for temporal analysis of model outcomes and spatial comparison, clusters at each survey year available were spatially linked to the most recent administrative II unit available for each country using ArcGIS software. Briefly, administrative units are subnational geographic areas used for administrative or political purposes, such as districts, regions, counties and states. In the United States, for example, administrative I units correspond to the state level, while administrative II units correspond to counties (with the exception of two states). Because the word for these geographic areas may vary substantially by country (i.e., district, county, borough, etc.), the administrative II-unit level used in these analyses is hereby referred to as the 'district' level for the purposes of this paper. I spatially linked survey data to the most current district boundaries available for each country, as both DHS and political boundaries in many of the study countries have changed since 1999, preventing direct comparison of change over time. Further, by disaggregating each country at a uniform district level, spatial comparisons across countries could be standardized.

5.3.2 Absolute indices of temporal change

I employed a Bayesian inference framework using the Integrated Nested Laplace Approximation (INLA) package (Rue et al., 2009) in R (R Development Core Team, 2017) to spatially interpolate coverage estimates for ANC, SBA, and PNC at the district level throughout the study countries. Specifically, I used the Besag-York-Mollier (BYM-2) class of models within the INLA package, which have been shown to be particularly useful in mapping disease, as unstructured spatial effects can be added to account for region-specific variation (Niragire et al., 2015; Riebler et al., 2016). The model was therefore defined as

$$logit(p_{ij}) = \beta_0 + \beta_1 x_{ij} + \beta_2 x_{ij} \dots \beta_5 x_{ij} + f_{spat}(admin_i)$$

where logit(p_{ij}) represents the odds of a woman's most recent birth, *i*, in administrative unit, *j*, obtaining the corresponding health service (SBA, ANC, and PNC); $\beta_0 + \beta_1 x_{ij} + \beta_2 x_{ij} \dots \beta_5 x_{ij}$ represents the fixed effects of the model as described below; and $f_{spat}(admin_j)$ represents the structured spatial effect of administrative unit, *j*, as a combination of both the structured and unstructured random effects, defined as

$$f_{spat}(admin_i) = f_{struc}(admin_i) + f_{unstruc}(admin_i)$$

For these analyses, I assumed an uninformative prior distribution on model parameters to allow the data to drive model results, as no previous literature or data exist at this level for each year to inform expectations of the spatial distribution of model outcomes. By assuming uninformative priors across all models at each time point available, this allowed for better comparison of model results. Similar approaches have been used previously (Neal et al., 2019) with adolescent health indicators using DHS data. The model outputs generated by this method represent a posterior distribution of possible estimates for each outcome at the district level. The mean of this distribution can therefore be taken to represent a point estimate for each geographic unit, while also allowing for reporting of standard distribution statistics (such as standard deviation and credibility intervals) which can be used to represent uncertainty surrounding each estimate. To visualize the absolute change over time among these indicators, I compared estimates for each country between the first and last surveys available.

Binary model outcomes included SBA, ANC, and PNC, while fixed model effects included urban/rural residence, education status, wealth quintile, maternal age, and parity, which have been defined in previous literature as important predictors of MNH services (Gabrysch and Campbell, 2009a; Neal et al., 2015; C. W. Ruktanonchai et al., 2016). To maintain comparability across countries, I defined skilled birth attendance as births attended by a doctor, nurse, or auxiliary midwife for the most recent birth available. Antenatal care was defined amongst the most recent birth as 4+ antenatal care visits, while postnatal care was defined as a maternal check-up within 48 hours of the most recent delivery by a health professional (doctor, nurse, or auxiliary midwife). For deliveries occurring at a health facility, I assumed postnatal care was provided by a health professional (as defined above) unless otherwise specified by the data. Lastly, I report model fit through the Deviance Information Criterion (DIC) values, which provide a measure of goodnessof-fit for Bayesian models, while adjusting for model complexity and effective number of parameters, with smaller DIC values representing better fitting models (Spiegelhalter et al., 2002).

5.3.3 Relative indices of temporal change

I examined relative indices of temporal change by quantifying the ratio between best-versus-worst modelled estimates among districts, with larger values representing increased gaps in coverage between districts, and smaller values nearing one representing decreasing spatial inequality. I further examined the temporal trend of spatial effects by reporting univariate logistic odds ratio (OR) using the 'lme4' package in R software (Bates et al., 2014) for each outcome and time point available. Similar approaches have been used by researchers at the DHS program (Assaf and Pullum, 2016) to temporally examine socioeconomic inequalities such as wealth in MNH. These analyses have reported coefficients for the richest quintiles as compared to the poorest, with values overlapping zero representing no statistically significant difference in services as predicted by wealth. I similarly report coefficients for DHS region with the best-performing (or highest coverage) for each outcome, as compared to the worst-performing (or lowest coverage) region, representing temporal trends in spatial inequalities divorced of modelled estimates. Specifically, I defined coverage as the proportion of women in the sample accessing a given service—for example, 90% of women reporting skilled attendance at birth would correspond to 90% coverage for this indicator. DHS regions used for reference and coefficients reported are outlined in Table B.2 (see Appendix B). More information on region boundaries used by the DHS can be found at spatialdata.dhsprogram.com/boundaries. In these analyses, values overlapping zero represent no significant effect of space in predicting odds of MNH outcome use, while increasing values represent a greater effect of space alone in predicting service utilisation.

5.3.4 Model validation

To validate model performance, I employed an out-of-sample validation technique, where 25% of the data were removed for validation purposes, while the remaining 75% were used for model training. I report standard validation statistics, including mean absolute error (MAE), mean square error (MSE), and pseudo-R2. Previous studies (Bosco et al., 2017; Gething et al., 2015) have employed similar statistics when interpolating surfaces using DHS data, and represent information on model precision, model bias, and variance explained, respectively.

5.3.5 Ethics approval

Ethics approval for this study was submitted and approved through the University of Southampton Ethics and Research Governance Council (ethics approval ID 26474). The data used in these analyses were obtained from the Demographic and Health Surveys (DHS) Program, which makes global health and demographic data confidentially and freely available to researchers across the world. More information on how the DHS program conducts the Informed Consent process can be found at <u>https://dhsprogram.com/What-We-Do/Protecting-the-Privacy-of-DHS-Survey-Respondents.cfm</u>.

5.4 Results

5.4.1 Absolute indices of temporal change

Figures 5.1a - d show modelled prevalence estimates of skilled birth attendance (i), antenatal care (ii), and postnatal care (iii) at the district level, and corresponding absolute change over time in these estimates between the first and last time points available. For absolute change over time, areas in blue indicate an increasing prevalence estimates between the first and last time points, while areas in white represent a small, but increasing change over time, and areas in red indicate decreasing prevalence estimates over time. The corresponding absolute change estimates for each districts are labelled and ordered increasingly in figures B.3 - B.6 (see Appendix B), to facilitate visualisation of greatest to least improvement for each district.

Rwanda (Figure 5.1c) represented the only country with exclusively increasing prevalence over time amongst all three services, with as much as 85% increases in SBA for some districts and nearing universal coverage for both SBA and PNC. While more moderate gains were seen for antenatal care, every area within the country saw increases in prevalence, ranging to as high as a 60% increase. Equally encouraging, the greatest increases for Kenya (Figure 5.1a) were primarily localized within the eastern region of Kenya, which has historically shown higher levels of wealth inequality and more adverse maternal health outcomes as compared to the rest of the country, particularly in the northern parts of the region (Assaf and Pullum, 2016). Increases in this area were also seen amongst skilled birth attendance and postnatal care, however these were comparatively more conservative increases.

Conversely, Tanzania (Figure 5.1b) had noticeable reductions in coverage throughout most of the country in utilisation of 4+ antenatal care visits, with some areas experiencing as much as a 55% reduction. This trend was predominantly driven by high coverage in ANC in 1999, which were substantially reduced in the 2010 and 2015 DHS surveys. This trend however, did not hold for skilled birth attendance and postnatal care, with much of the country experiencing improvement amongst these services. While Uganda (Figure 5.1d) experienced improvement in skilled birth

attendance over time, it experienced equally substantial decreases or little to no improvement in antenatal care and postnatal care in Northern and central Uganda. Similar to Tanzania, these trends were predominantly driven by high estimates during the 2000 DHS, decreasing with the 2006 and 2011 surveys.





Figure 5.1 Predicted prevalence and absolute change in i) skilled birth attendance (blue), ii) 4+ antenatal care visits (green), and iii) postnatal check-up within 48 hours (red) among administrative II units, among DHS data in a) Kenya, 2003 – 2014 b) Tanzania, 1999 – 2015, c) Rwanda, 2005 – 2014, and d) Uganda, 2000 – 2011.

Model outcomes, estimates and fit are outlined in Table B.3 (see Appendix B). Figure 5.2 compares model fits across outcomes by country, as measured by the DIC. Of note, the y-axis for DIC values varies by country, as these values should not be compared between countries. In general, SBA and PNC models tended to perform better as compared to ANC models, with the exception of the most recent models in Uganda, where models all performed similarly. DIC values also tended to increase over time, possibly due to increasing sample size (see Table B.3), with the exception of Rwanda. Models in Rwanda tended to perform similarly in 2010, but had more variability in 2005 and 2014. However, this variability only ranged between 30 points, similar to models in other countries.



Figure 5.2 DIC values for skilled birth attendance (blue), 4+ antenatal care visits (green), and postnatal care (red) models, by country, DHS data, 1999 – 2015.

5.4.2 Relative indices of temporal change

Figure 5.3 shows performance ratios of each outcome, reflecting the relative gap within countries between the district with the highest-versus-lowest prevalence estimates over time. Values closer to one therefore represent spatial homogeneity in coverage, while higher values represent more disparate gaps within countries amongst services. Across all countries, the lowest ratios tended to be among ANC, with the exception of Rwanda, while SBA and PNC tended to have greater ratios between the highest and lowest district estimates, suggesting greater spatial heterogeneity. Generally, ratios have typically decreased or remained stably low over time across all countries.

Historically, Rwanda had relatively small ratios that decreased over time, while Kenya had the highest inequalities in SBA, these were substantially reduced over time. Tanzania equally experienced reductions in SBA and PNC, but saw little to no change in ANC over time as it was already relatively low. Uganda also saw substantial improvement in reducing disparities over time for SBA and PNC (despite an increasing ratio for the year 2005), nearly halving the SBA ratio from 25 to 12 within the span of a decade. Regardless, the most recent ratios in Kenya and Tanzania across service utilisation still remained around 5, meaning the best performing region of the country had coverage about 5-times higher than the corresponding lowest region. Further, while Uganda saw improvement across services, ratios amongst SBA and PNC still represent the highest ratios across the region, suggesting there is still need for further improvement.

Chapter 5



Figure 5.3 Ratio of modelled best-to-worst performing administrative II units over time for skilled birth attendance (blue), 4+ antenatal care visits (green), and postnatal care (red), by country, DHS data, 1999 - 2015.

Lastly, Figure 5.4 shows results of the unadjusted logistic odds ratio, plotting coefficients of the DHS region with the highest corresponding coverage as compared to the region with the lowest coverage as reference. Positive, significant coefficients imply a statistically significant effect of space exists, while estimates spanning zero imply no significant difference across regions in likelihood of utilising SBA, ANC, or PNC. Historically, Rwanda saw high levels of inequality in utilising MNH services by space, but univariate logistic odds results suggest the effect of space has been substantially reduced over the decades, representing the country with the lowest odds ratios across services amongst the most recent survey. Of note, the odds of obtaining PNC in 2010 was negative, implying odds in the best performing regions were actually reduced as compared to the worst performing region. Kenya further saw decreasing ORs over time for both ANC and PNC, but saw little improvement for SBA, with Nairobi having 2 to 3 times higher odds of utilising skilled

birth attendance over time as compared to the North Eastern region. Despite this improvement, Kenya's ORs remain amongst the highest across the region, in combination with Tanzania.

In contrast to Rwanda and Kenya's decreasing trends in ORs, Tanzania and Uganda had substantial variability in ORs over time. Encouragingly, the effect of space is relatively low in Uganda amongst ANC and PNC, but has increased over time for SBA. However, the odds of obtaining SBA amongst the DHS region with the highest SBA coverage versus the lowest was just over 1 in 2011, representing the lowest coefficient for SBA outside of Rwanda. Tanzania also saw an increasing effect of space over time for PNC, and further has high ORs along with Kenya.



Figure 5.4 Unadjusted odds ratio of skilled birth attendance (blue), 4+ antenatal care visits (green), and postnatal care (red) over time, with worst performing DHS region as reference, DHS data, 1999 – 2015.

5.4.3 Model validation

Model validation results are show in Table 5.2. Generally, model fit was greatest for all services in Kenya, ranging from 88.4% variance explained for ANC in 2008 to as high as 99.7% in 2014 for

SBA and PNC. Model fit was poorest for ANC in Uganda, with psudeo-R2 estimates of 0.138 and 0.239 for ANC in 2006 and 2014. In general, model fit was poorest for ANC for all years and countries, with the exception of Rwanda after 2010. Model precision and bias were predominantly uniform across countries, but was worst among Uganda models, potentially contributing to the low explained variance.

Kenya				Tanzania					
	MSE	MAE	Psuedo -R ²		MSE	MAE	Psuedo -R ²		
2003				1999					
SBA	0.0022	0.0344	0.949	SBA	0.0099	0.076	0.876		
ANC	0.002	0.0337	0.925	ANC	0.0164	0.101	0.544		
PNC	0.0019	0.0286	0.960	PNC	0.0098	0.0763	0.855		
2008				2010					
SBA	0.0013	0.027	0.973	SBA	0.0049	0.0501	0.929		
ANC	0.0021	0.0357	0.884	ANC	0.0067	0.0580	0.796		
PNC	0.0017	0.030	0.967	PNC	0.0041	0.0455	0.938		
2014				2015					
SBA	0.00009	0.0073	0.997	SBA	0.0043	0.0441	0.901		
ANC	0.0001	0.0095	0.987	ANC	0.006	0.055	0.787		
PNC	0.00008	0.0072	0.997	PNC	0.004	0.0376	0.925		
	Rw	anda		Uganda					
	MSE	MAE	Psuedo -R ²		MSE	MAE	Psuedo -R ²		
	2	005		2000					
SBA	0.005	0.0538	0.80	SBA	0.024	0.115	0.656		
ANC	0.003	0.0404	0.475	ANC	0.04	0.151	0.529		
PNC	0.0043	0.0517	0.816	PNC	0.035	0.140	0.702		
	2	010		2006					
SBA	0.002	0.033	0.887	SBA	0.030	0.127	0.657		
ANC	0.001	0.0300	0.905	ANC	0.054	0.173	0.138		
PNC	0.001	0.0326	0.877	PNC	0.028	0.124	0.684		
2014				2011					
SBA	0.0008	0.024	0.704	SBA	0.025	0.119	0.678		
ANC	0.0023	0.0365	0 795	ANC	0.050	0.171	0.239		
	0.0025	0.0505	0.775	AILC	0.020	011/1	0.20)		

Table 5.2 Model validation results

5.5 Discussion

The vast majority of maternal deaths can be prevented through routine health services, such as antenatal care and skilled birth attendance, or treated through timely interventions and prevention (Bailey et al., 2006). Within developing countries, however, the use of key life-saving interventions can be limited and inequitably distributed, and varies by country-specific contextual issues, such as funding and organization of health care or social and cultural issues (Say and Raine, 2007). Continuing MDG progress in preventing maternal deaths and achieving SDG targets of "ensuring healthy lives and promoting well-being for all at all ages" (WHO, 2015b) will require more resolved spatial, temporal, and demographic information to identify and monitor persistent health inequalities. Examining temporal change in spatial inequalities of maternal health service use, however, requires reporting of both absolute and relative indices, as such measures often interact with each other synergistically and therefore require joint interpretation (Barros and Victora, 2013).

Here, I found that Rwanda was the only country to make substantial progress in both absolute and relative measures, increasing coverage amongst all services and reducing relative inequalities. Importantly, Rwanda had historically low relative inequality in service coverage (Figure 5.3), suggesting that the increases in service coverage seen in Rwanda over the decades were experienced in spatially equitable manner, with most of the country improving together. These results are in line with previous findings, given that Rwanda was the only country in the region to achieve the MDG target 5a (reduce MMR by 75% between 1990 and 2015) (WHO et al., 2015). After experiencing a devastating genocide in the mid-90s, Rwanda radically re-developed their health system, aimed at: 1) coordinating policy with external donors and government aid; 2) implementation of national-level health insurance; and, 3) introduction of a performance-based pay system for health workers (Logie et al., 2008). This commitment to improving health across the country translated into a nearly 78% reduction in MMR throughout the MDG era, as well as a substantial reduction in under-5 mortality (Golding et al., 2017), and may contribute to the findings outlined in this study.

I found that Uganda has experienced both absolute and relative improvement over time in SBA across most of the country, while improvement in PNC and ANC lagged, particularly in the northern region. Previous studies (Ayiko et al., 2009) have similarly found poor outcomes in underfive mortality in this area, attributing the trend partly to the nearly two decades of armed conflict in the region which disrupted health systems and impacted the socio-economic stability across the region. Other studies (Parkhurst et al., 2005) have found that while skilled birth attendance is high, other key MNH metrics such as vaccine coverage lag, and suggest that quality of care in Uganda is insufficient, resulting in delayed emergency treatment and insufficient supplies. Further studies (Roberts et al., 2015) have found minimal increases over time in MNH indicators requiring multiple contacts with the health system, such as 4+ antenatal care visits, consistent with these

findings. This suggests that while Uganda has rapidly increased the number and scale of maternal health interventions across the country, some areas (particularly in the northern regions) have historically and systematically lagged behind and require more deliberate efforts and focused interventions to further close the gap in MNH inequalities (Ayiko et al., 2009).

In Tanzania, I found a notable absolute decrease in coverage of 4+ antenatal care visits between 1999 and 2010, with little change through 2015. Regardless, northern and north-eastern Tanzania experienced a relative reduction over time across all services, suggesting the possibility that these areas are being left behind in improving access to or utilisation of maternal health services. Tanzania achieved MDG targets for child survival, and while previous studies (Hanson et al., 2017) have noted that geographic inequalities in access to primary care at childbirth have been reduced, inequalities persist in actual hospital-based deliveries and antenatal care. These findings are in line with these studies (Armstrong et al., 2016) suggesting maternal survival has lagged behind due to low coverage of maternal health services, with rural women bearing a disproportionate burden of risk. Examining potential bottlenecks to explain these patterns, Armstrong et al. (Armstrong et al., 2016) suggested that Tanzania must make progress in all four "benchmarks" of quality health systems, as the country experienced low density of health workers, poor availability of supplies at health facilities, and low levels of health financing, particularly in the Lake and Western districts. Despite this, I found ratios between districts generally decreased slightly, suggesting some progress is being made in reducing inequalities. Further, I found the odds of region alone in predicting SBA substantially declined and is remaining relatively stable with ANC, yet is noticeably increasing for PNC. These findings suggest that geographic inequalities in coverage of SBA are being reduced within country, but coverage in ANC and PNC must be followed closely and prioritized over the coming decades.

Finally, I found that Kenya experienced substantial improvement in absolute change of 4+ ANC visits over time, particularly with the eastern districts, but coverage in SBA and PNC decreased, particularly in the northern districts. I further found that while relative inequalities between districts are reducing across services, they still remain high compared to the rest of the region. Across the country, I also found that urban areas experienced substantially higher odds of obtaining MNH services (and particularly skilled birth attendance) as compared to rural areas, in line with previous research (Assaf and Pullum, 2016). Nguhiu et al. similarly found that while overall coverage of MNH interventions has steadily increased over time and maternal inequities decreased within Kenya, coverage of individual interventions including antenatal care and skilled birth attendance remained stubbornly low and inequitable, with ANC experiencing the most inequitable coverage (Nguhiu et al., 2017). They suggest that increased overall coverage may be linked to increasing per capita health expenditure within the country, as well as urbanization and expansion of lower level health facilities over the decades, but propose that interventions directed at those services lagging behind should be accordingly prioritised over the coming decades.

5.5.1 Limitations

As this work primarily utilises estimates generated using statistical inference, it is subject to several limitations, including survey and sampling errors inherent to DHS data. Particular to spatial interpolation at levels below the DHS region, these data were collected using methods representative at geographic units which are different from the measures I report, potentially resulting in model uncertainty and errors. However, the DHS program endorses use of geospatial interpolation methods such as Bayesian inference, particularly because of the ability to quantify this uncertainty in posterior distribution estimates (Gething et al., 2015). The DHS data is also subject to temporal biases, as it is collected at varying intervals between countries, and information on maternal services are collected for up to five years previous. This represents a potential time lag in temporal analyses, and the trends reported here may not represent the current picture of MNH service utilisation in these countries. Further, with some of the countries used in these analyses, the most recent DHS available is upwards of five years old, representing an important avenue for future analyses to continue to examine these trends using more recent data. Lastly, the actual hospital or health facility used for the services used was not included in these analyses, as the DHS does not report this information. This could potentially bias model results, as the hospital or health facility may be outside of the respondent's surveyed district. Actual health facility used should be included in future research if available, as well as cross-border movement.

While the scope of this paper is to report temporal trends in spatial inequalities of MNH service use over time, future work should aim to explain why these patterns are occurring to potentially offer insight into policy interventions aimed at maintaining progress to ensure no populations are left behind in accessing MNH services. Future research should examine factors at several socio-ecological levels, including the individual, cultural, and national level, as well as quality of care provided. Spatial context such as travel time to health facilities and actual health facilities used should also be taken into account (Makanga et al., 2016).

5.6 Conclusions

This study is the first to report model-based estimates at the district level for several time points across the EAC region, as well as report temporal trends in both absolute and relative measures of spatial inequalities. I found that relative inequality between districts have generally decreased or remained stably low over time across all countries, suggesting improvements are being made to reduce the gap between the geographic areas with the highest and lowest coverage in services. I further found that Rwanda in particular was successful in reducing relative inequalities over time, as well as increasing absolute coverage across all MNH services. Despite this progress, I found that relative measures of spatial inequalities across the region indicate that the effect of space is becoming more prominent over the decades among PNC in particular, suggesting this should be

monitored closely throughout the SDG era and examined further with new sources of DHS data or other household surveys.

These analyses demonstrate how the use of spatial and temporal disaggregation methods can be used to monitor the evolution of health inequalities over the SDG period. These results highlight the need for continued disaggregation of these data over time, which will be key in improving health amongst all populations across the East African Community. Lastly, in-country uptake and adoption of GIS-enabled analytics facilitating concurrent temporal and spatial analysis of socio-economic, environmental and health systems level determinants will be key to development of precise policy actions for addressing areas with intractable health challenges and ensuring that SDG health targets are met by 2030.

5.7 Acknowledgements

CWR is supported by the University of Southampton's Economic and Social Research Council's Doctoral Training Centre. AJT was supported by funding from NIH/NIAID (U19AI089674), the Bill & Melinda Gates Foundation (OPP1106427, 1032350, OPP1134076), the Clinton Health Access Initiative, the RAPIDD program of the Science & Technology Directorate, Department of Homeland Security and the Fogarty International Center, National Institutes of Health, and a Wellcome Trust Sustaining Health Grant (106866/Z/15/Z). The authors would like to NWR for discussing data visualization techniques for this manuscript, and colleagues at the EAC Secretariat Headquarters for their ongoing support and feedback on these analyses. CWR would also like to thank OP for support and for lending an open ear throughout the data analysis process.

5.8 Intellectual contribution

The author list for this published work is as follows: Ruktanonchai CW, Nilsen K, Alegana VA, Bosco C, Ayiko R, Charles ASK, Matthews Z, Tatem AJ. CWR conceived of the study framework and methods for this study, performed all data management and analysis, generated all study visualizations and results interpretations, and wrote and prepared the manuscript for journal submission. CB, VA and KN contributed insight and feedback into the statistical and geographical analysis for this study. RA and AC provided feedback on broad implications of study findings relevant to the study region. ZM and AJT supervised CWR during production of this manuscript, and oversaw that methods and analysis were executed within a sound and scientifically rigorous framework. All authors reviewed the manuscript and provided feedback before final publication.

Chapter 6: Estimating uncertainty in geospatial modelling at multiple spatial resolutions: the pattern of delivery via caesarean section in Tanzania

6.1 Abstract

Visualising maternal and newborn health (MNH) outcomes at fine spatial resolutions is crucial to ensuring the most vulnerable women and children are not left behind in improving health. Disaggregated data on life-saving MNH interventions remain difficult to obtain, however, necessitating the use of Bayesian geostatistical models to map outcomes at small geographical areas. While these methods have improved model parameter estimates and precision among spatially correlated health outcomes and allowed for the quantification of uncertainty, few studies have examined the trade-off between higher spatial resolution modelling and how associated uncertainty propagates.

Here, I explore the trade-off between model outcomes and associated uncertainty at increasing spatial resolutions by quantifying the posterior distribution of delivery via c-section in Tanzania. Overall, in modelling delivery via c-section at multiple spatial resolutions, I demonstrate poverty to be negatively correlated across spatial resolutions, suggesting important disparities in obtaining life-saving obstetric surgery persist across sociodemographic factors. Lastly, I note that while uncertainty increased with higher spatial resolution input, model precision was best approximated at the highest spatial resolution, suggesting an important policy trade-off between identifying concealed spatial heterogeneities in health indicators.

6.2 Introduction

Achieving the Sustainable Development Goal aims laid out in 2015 necessitates measurement of health outcomes at small geographical areas to ensure 'no one left behind' (C. Edson Utazi et al., 2018b). With recent advancements in the collection and distribution of geo-located household surveys, such as those collected via the Demographic and Health Survey (DHS) program (<u>www.dhsprogram.com</u>), researchers are increasingly utilising methods such as small area estimation and geostatistical additive models to generate high spatial resolution maps of health and development indicators (Blangiardo and Cameletti, 2015; Gething et al., 2015; C. E. Utazi et al., 2018). Such subnational, high-resolution estimates have become useful tools for researchers and policy makers alike in uncovering hidden health inequities which would otherwise be masked by aggregate or national-level health indicators, enabling targeted interventions in settings with limited resources (Pezzulo et al., 2016; C. W. Ruktanonchai et al., 2016, 2018; C. Edson Utazi et al., 2018b; WHO, 2015a).

Visualising health outcomes and associated uncertainty at high spatial resolutions has distinct policy relevance amongst maternal and newborn health (MNH) outcomes (C. Edson Utazi et al., 2018a), as maternal and neonatal mortality both vary geographically and occur relatively rarely. Further, the data associated with maternal mortality are subject to limitations, misclassification, and bias (Ahmed et al., 2014; Ebener et al., 2015), particularly within more rural areas of sub-Saharan Africa where many deaths do not occur at hospitals and may go unrecorded (Say et al., 2014b). As with maternal mortality, data on life-saving MNH interventions such as antenatal care, skilled birth attendance, and delivery via caesarean section (c-section) can be widely obtained at aggregate levels, but remain difficult to measure at subnational levels, especially in the most rural and vulnerable areas of the world. While some work has been done modelling key MNH interventions at subnational scales such as maternal health services, exclusive breastfeeding, childhood vaccinations, and health systems performances, (Bhattacharjee et al., 2019; Mosser et al., 2019; Roberts et al., 2015; C. W. Ruktanonchai et al., 2016, 2018), other vital life-saving interventions which occur less frequently, such as delivery via c-section, have not been modelled previously at high spatial resolutions.

With advancements in computational resources and data availability over recent decades, researchers across disciplines have begun employing Bayesian geostatistical additive models (GAM) to map disease and quantify uncertainty in posterior model outcomes, particularly using hierarchical clustered data such as from the DHS (Blangiardo and Cameletti, 2015; Gething et al., 2015; Magalhães and Clements, 2011; Patil et al., 2011; Schur et al., 2011). The application of these methods is increasingly pertinent, as access to healthcare services is heterogeneously distributed across landscapes, requiring high-resolution spatial data and modelling techniques to identify the most vulnerable populations. However, these methods and associated spatial data carry

limitations and bias, manifesting in uncertainty which should be adequately quantified and communicated to decision makers and non-academic audiences for optimum policy impact (Tatem et al., 2012). While the use of such GAMs to predict high-resolution health outcomes has improved model parameter estimates and precision among spatially correlated and rare adverse health outcomes (Gething et al., 2015; Kirby et al., 2017) and allowed for this quantification of uncertainty (Clements et al., 2006b), no studies have examined the trade-off between predicting health outcomes at higher spatial resolutions and visualising the spatial distribution of associated uncertainty (Goovaerts, 2006).

Here, I estimate prevalence of delivery via c-section in Tanzania, using input covariates at varying levels of spatial coarseness within a Bayesian geostatistical model framework. With these models, I investigate how uncertainty varies with spatial resolution, and how this changing uncertainty can be better visualised and communicated. Specifically, I explore the trade-off between model estimates and associated uncertainty at increasing spatial resolutions through exploration of the posterior distribution of modelled delivery via c-section at multiple spatial resolutions.

6.3 Methods

6.3.1 DHS data

I compiled Demographic and Health Survey (DHS) data from Tanzania for 2015 (Ministry of Health et al., 2016) using SAS version 9.4 software (SAS Institute Inc., 2013), and restricted the sample to women with a birth in the preceding 5 years (n = 7,050 women) with corresponding spatial data, as provided by DHS cluster locations. Briefly, the DHS provides GPS coordinates for clusters of aggregated household survey data in order to facilitate spatial analyses, while also maintaining participant confidentiality. These coordinates are displaced up to 2km in urban areas and 5km in rural areas, with up to 1% of points displaced up to 10km in rural areas (Burgert et al., 2013). Using these geo-located cluster locations, spatial inference occurs at a higher spatial resolution than the geographic region in which the survey is designed to be representative of. This hierarchical sample design therefore necessitates the use of geostatistical models to make inferences at spatial resolutions finer than the DHS region level (Gething et al., 2015). In these analyses, the binary outcome of interest was defined as the number of women who underwent any delivery via c-section for a birth within the preceding five years (regardless of whether it was the most recent), as compared to women who had not experienced delivery via c-section for any preceding birth. To maintain survey representativeness, this was calculated at the DHS level 1 resolution, representing the 30 administrative I regions of Tanzania (Figure 6.1).



Figure 6.1 Delivery by caesarean section at the administrative 1 level using DHS data, Tanzania, 2015

6.3.2 Covariate data

In addition to demographic data gathered through the DHS, I also compiled environmental geospatial covariate data that I extracted to DHS cluster locations (Figure 6.2). Because these locations are displaced, I averaged geospatial covariates to 2km and 5km buffers for urban and rural locations, respectively. While up to 1% of coordinates in rural areas are displaced within a 10km radius, the addition of buffers at the 10km level have been shown to impact very few rural coordinates (Burgert et al., 2013), while unnecessarily introducing bias in environmental covariates, thereby justifying use of a 5km buffer in rural areas. Firstly, I gathered data from the European Commission's Joint Research Centre on accessibility to major cities for the year 2000, representing travel time to the nearest city exceeding a population of 50,000 using land (i.e., roads) or water-based travel (i.e., rivers and lakes) (Nelson, 2008). Next, I included data on annual night light intensity for the year 2013 (an indicator of urbanicity), as generated by the National Oceanic and Atmospheric Administration's (NOAA) National Centers for Environmental Information (NCEI) (US NOAA National Geophysical Data Center and US Air Force Weather Agency, 2014). I also included live births for the year 2015 at the 1 km resolution, as well as Multidimensional Poverty Index (MPI) estimates for the year 2010, as obtained via the WorldPop Project (www.worldpop.org.uk) and outlined by Tatem et al. (A. Tatem et al., 2013; Tatem et al., 2014).

Lastly, I included travel time to the nearest public hospital, as outlined by Ouma et al. (Ouma et al., 2018), calculated through a cost-distance algorithm incorporating a travel impedance surface by assigning travel speed to road networks.



Figure 6.2 Study analysis flowchart

These covariates were chosen as previous studies have shown them to be predictive of MNH outcomes and risks (Bosco et al., 2017; Gabrysch and Campbell, 2009b; Gething et al., 2012; Mpembeni et al., 2007; C. W. Ruktanonchai et al., 2018), representing a suite of geospatial covariates with robust predictive power to examine how uncertainty changes as a function of spatial resolution. Notably, I chose to include only geospatial covariates in this model as I could vary the spatial resolution of these variables, thereby addressing research objectives. Specifically, these datasets were compiled at the 1km spatial resolution, and subsequently averaged at the 5km, 10km, and 100km resolutions to facilitate projecting the fitted model onto gridded surfaces at these levels. These surfaces represent a theoretical exploration of the trade-off between increasing gridded spatial resolution and modelled estimates, and were chosen to clearly illustrate the difference between in the practical range of estimates, as shown in Figure C.8 within Supplementary Information.

6.3.3 Model framework

To explore how uncertainty in posterior modelled c-section delivery estimates varied at multiple spatial resolutions, I employed a Bayesian hierarchical model framework with input covariates at varying spatial coarseness. These models have been used extensively with DHS and other household survey data (Bearak et al., 2018; Hug et al., 2019; Kazembe and Namangale, 2007; Neal et al., 2019; C. Edson Utazi et al., 2018b; C. E. Utazi et al., 2018), as they are able to robustly

account for the multi-stage sampling efforts employed during the data collection process, resulting in hierarchically structured data provided through the DHS. Here, these models account for the nested structure of DHS data by allowing for variation in the *n*th region among individual respondents, as outlined below. These models were fit independently of each other, resulting in 3 models with input covariates and modelled outcomes at the 5km, 50km, and 100km spatial scales. To predict c-section delivery at a continuous spatial resolution, I implemented these models via stochastic partial differential equation (SPDE) spatial regression approach, implemented using the Integrated Nested Laplace Approximation (INLA) technique within the R-INLA package (Rue et al., 2009). This approach was suitable for this analysis as these spatial processes are generally well captured by a Gaussian field with Matérn correlation (Krainski and Castro-camilo, 2018). These models have similarly been used in previous research combining DHS data and geospatial covariates to predict high-resolution childhood vaccination coverage by disaggregating aerial surveillance data (Bosco et al., 2017; C. Edson Utazi et al., 2018b; C. E. Utazi et al., 2018). I employed a similar modelling framework, generally defined as

$$Y_i \sim \text{Binomial}(N_i, p_i), i = 1, ..., n_A, n_A + 1, n_A + n_p$$

where n_A represents subnational DHS regions within Tanzania to maintain survey representation; Y_i represents the number of women delivering via c-section within area, A_i ; p_i represents the probability of a woman delivering via c-section over grid points n_p ; and, N_i represent the number of women surveyed within area, A_i .

In this framework, the areal units and observation grid points are linked using the following equations

$$logit(p_i) = \tilde{x}_i'\beta + |A_i|^{-1} \int n(s)ds + \phi_i, i = 1, ..., n_A$$
$$logit(p_i) = x_i'\beta + \eta(s_i) + \phi_{A_i}, i = n_A + 1, ..., n_A + n_p$$

where x_i and \tilde{x}_i represent covariates for the *i*th area and grid point, respectively. This framework provides a statistical link between the areal data and high-resolution spatial covariates and random effects, allowing for models at two spatial levels. Further details on the model framework are outlined in Utazi et al. (C. E. Utazi et al., 2018). See (Bosco et al., 2017; Krainski and Castrocamilo, 2018; C. E. Utazi et al., 2018) for more detailed information on similar approaches implementing an SPDE approach of Bayesian hierarchical models via R-INLA using DHS data.

6.3.4 Ethics approval

The University of Southampton Ethics and Research Governance Council Ethics approved this study (ethics approval ID 47073). The data used in these analyses were obtained from the

Demographic and Health Surveys (DHS) Program, which makes global health and demographic data confidentially and freely available to researchers across the world. More information on how the DHS program conducts the Informed Consent process can be found at https://dhsprogram.com.

6.4 **Results**

Table 6.1 shows posterior marginal effects for the fixed effects within the 5km, 50km, and 100km models as well as model hyperparameters. Fixed effects estimates with upper and lower 95% credible intervals which do not cross 1 are considered significant. Overall, I found that modelled c-section prevalence negatively correlated strongly with poverty and slightly with night-time lights across all spatial scales, as shown in Figure C.7. Of note, night-time lights were not significant within the model, but presented wide credible intervals for marginal effects, as seen in Table 6.1. While poverty was not significant at the 5km scale, it was significant at the 50km and 100km scales, and showed a consistent pattern at the 5km scale with other spatial resolutions (Figure C.7). Conversely, these estimates were strongly positively associated with travel time to the nearest hospital across scales, although this was not significant within the model.

	5 km			50 km			100 km		
Parameter	Mean	Lower 95% CI	Upper 95% CI	Mean	Lower 95% CI	Upper 95% CI	Mean	Lower 95% CI	Upper 95% CI
Accessibility to									
cities	1.0001	0.9975	1.0027	0.9999	0.9975	1.0022	0.9999	0.9975	1.0022
Night-time lights	0.9603	0.0522	19.7189	0.666	0.0308	14.7623	0.6574	0.0298	14.9301
Live births	1.0471	0.8645	1.2873	0.9915	0.8313	1.1913	0.9938	0.8313	1.197
Poverty	0.0271	0.0005	2.1231	0.0071	0.0001	0.4548	0.0071	0.0001	0.4682
Travel to nearest									
hospital	1.0046	0.9946	1.0143	1.0062	0.9965	1.0163	1.0063	0.9964	1.0164
Hyperparameters									
	Mean	SD	95% CI	Mean	SD	95% CI	Mean	SD	95% CI
Θ_1	-0.9259	0.6329	(-2.1791, 0.3104)	0.229	0.942	(-1.636, 2.071)	0.213	0.945	(-1.646, 2.070)
Θ_2	0.1348	0.4988	(-0.8380, 1.1226)	0.041	0.724	(-1.376, 1.472)	0.052	0.718	(-1.352, 1.474)
Precision	1816.2045	1813.6598	(121.78, 6626.24)	5.290	5.794	(0.647, 20.476)	5.814	6.901	(0.648, 23.539)
λ	0.5020	0.2699	(0.0497, 0.9491)	0.354	0.243	(0.031, 0.876)	0.354	0.243	(0.030, 0.876)
DIC	155.67		152.4		152.61				
P _D	17.65		17.70		17.81				
Marginal	-118.52		-120.84		-120.89				
likelihood									

Table 6.1 Marginal effects of the fixed effects and hyperparameters of the posterior c-section models at 5km, 50km, and 100km

The large precision estimate for the 5km model as shown in Table 6.1 suggests the spatial process was estimated well with a Gaussian field, while smaller estimates among the 50km and 100km models suggest this was not the case. Regardless, the deviance information criterion (DIC) estimates for the latter two models were slightly improved over the 5km model (Table 6.1). Briefly, these DIC estimates represent a measure of model comparison, trading off between model complexity and model goodness of fit, and performing well among Bayesian models in particular (Spiegelhalter David J. et al., 2002). Smaller DIC values represent models with better fit, given model complexity, suggesting these models perform better as compared to the other models.

Figure 6.3 shows the distributions of the posterior 95% credible intervals for each model as violin plots. These plots show similar summary statistics as boxplots, while also providing information on probability densities, where thinner sections represent a lower probability of a given value occurring. These estimates approximate the trade-off between spatial resolution and uncertainty, representing the density of the width between the posterior upper and lower 95% credible interval for each grid cell for the 5km, 50km, and 100km surfaces. Here, all models had CI widths ranging from near zero to 1, but CI width became more narrowly distributed and approached zero with decreasing spatial resolution. Mean density at the 100km and 50km scales were 0.13 (\pm 0.14) and 0.14 (\pm 0.13), respectively, while mean density at the 5km scale was 0.21 (\pm 0.16).





Figure 6.4 visualises prevalence of delivery via c-section and associated uncertainty at the 5km, 50km, and 100km spatial resolution. These maps show spatial patterns typical of c-section

deliveries, with higher prevalence observed in cities such as Arusha and Dar Es Salaam, and lower prevalence in areas with high inaccessibility to a health facility or amongst more impoverished women (Table 6.1). Overall, the mean estimated prevalence of obtaining a c-section at delivery at the 5km resolution was $8.7\% (\pm 6.2\%)$, while the mean uncertainty as measured by the posterior distribution was $20.9\% (\pm 15.7\%)$. Mean estimated prevalence was slightly lower at the 50km and 100km resolutions, measuring 7.9% ($\pm 4.9\%$) and 7.8% ($\pm 4.8\%$), respectively, while mean uncertainty was 14.2% ($\pm 12.7\%$) and 13.1% ($\pm 14.3\%$). Areas of higher c-section utilisation were associated with higher uncertainty and observed around major urban areas, notably Dar Es Salaam, Arusha and Moshi, and Dodoma. This trend was observed across spatial resolutions, and in accordance with DHS data. Amongst DHS regions, Dar Es Salaam had the highest prevalence of delivery via c-section at 17%, compared to a national average of 5.9% (Figure 6.1).



Figure 6.4 Modelled c-section prevalence estimates (left) and associated 95% credible interval (right) at 5km (top), 50km (middle), and 100km (bottom), Tanzania, 2015

6.5 Discussion

Overall, I found increasing model uncertainty associated with increasing spatial resolution, as quantified by increasing 95% credible interval widths (Figure 6.3). This is unsurprising, given that increasing spatial resolution comes with problems of increasingly sparse events, zero inflation, and missing data (Richardson Sylvia et al., 2004). Further, both c-section prevalence as well as model uncertainty tended to be higher in urban areas, reflecting greater variance within the data. This uncertainty could be due to increased data availability within large population centres, or potentially due to GPS displacement error within urban areas. While Bayesian hierarchical modelling techniques have developed to help account for these limitations through rigorous quantification of uncertainty (Gething et al., 2015), these findings suggest an important and often overlooked trade-off persists between modelling these high-resolution health indicators and corresponding policy relevance if these estimates tend to be highly uncertain. Despite increasing uncertainty which accompanied increasing spatial resolution, I found that the 5km model was the most precise, as evidenced by a high precision estimate (Table 6.1). These findings correspond to findings reported in the Tanzania DHS report (Ministry of Health et al., 2016) where rates of csection utilisation are higher in urban areas. Other studies similarly suggest that while the prevalence of caesarean section is increasing globally (Betrán et al., 2007; Vogel et al., 2015), women in more rural areas that cannot access a health facility quickly have a lower chance of undergoing the procedure in emergency circumstances (Mishra and Ramanathan, 2002; Neuman et al., 2014).

Within the models, I further found that poverty was negatively correlated with undergoing delivery via c-section across spatial resolutions, and was significant at lower spatial resolutions (50km and 100km). These findings are again in line with reported DHS findings suggesting women in the highest wealth quintile were eight times more likely to undergo a c-section as compared to those in the lowest quintile (Ministry of Health et al., 2016). Because health insurance coverage is generally low in Tanzania and relies heavily on payment at point-of-service (Mtei et al., 2012), these findings may suggest that more impoverished women are either unable to afford c-section surgeries when needed, or are generally accessing health care less frequently across the continuum of pregnancy and childbirth.

Researchers are increasingly quantifying health and development indicators at the district level and high spatial resolutions, with aims of achieving Sustainable Development Goals to ensure "no one left behind" (United Nations, 2018; WHO, 2015a). While identifying these previously hidden pockets of vulnerable and marginalised populations is vital to improving the health and well-being of all, the geostatistical methods used to accomplish these goals have inherent uncertainty and bias, which should be communicated effectively to policy makers and other non-academic audiences. While studies have recognized the importance in quantifying this uncertainty (Gething et al., 2015;
Goovaerts, 2006; Richardson Sylvia et al., 2004), no studies have explored how increasing spatial resolution impacts model estimates and uncertainty. This study is therefore the first to map high-resolution estimates of c-section prevalence in Tanzania, and examine the trade-off between increasing spatial resolution and associated model uncertainty in these estimates. These results of this study imply an important trade-off between identifying concealed spatial heterogeneities and accuracy of estimates, which should be optimally communicated in policy relevant settings.

6.5.1 Limitations

This methodological study was exploratory in nature, examining the trade-off between increasing spatial resolution and model uncertainty, and is therefore subject to a variety of limitations. Firstly, I used a suite of standard covariates to explore the impact of spatial resolution alone on model uncertainty, and therefore rigorous covariate selection and model validation efforts were not undertaken during these analyses. As such, the results of these models may not be generalizable to Tanzania, nor to other study countries, and should be used for illustrative purposes only. Further, the spatial resolutions chosen for these analyses represent a theoretical exploration of the impact of increasing resolution on modelled estimates, and are unlikely to represent estimates at resolutions at which policy decisions are made. Future work may explore the impact of increasing spatial resolution on modelled estimates at policy relevant administrative boundaries, as opposed to the rasters employed in these analyses. Secondly, the suite of covariates and DHS data used are subject to their own biases and limitations—for example, the DHS collects data on births within the previous five years, so the estimates presented here may not reflect the current situation within Tanzania. Further, DHS data are not routinely collected registration data, and therefore do not capture information on c-sections performed on women who have subsequently died. This potentially represents biased information, as only women obtaining and surviving the procedure are interviewed. Additionally, travel to the nearest public hospital does not account for individuals who may bypass the nearest facility in favour of a facility with higher quality of care. Lastly, the geospatial covariates used have associated error and misclassification bias, particularly night-time lights which may suffer from light refraction errors, for example. Future work should aim to include more recent data on actual health facility used, where possible, and explore the impact of misclassification bias inherent to these environmental covariates.

6.6 Conclusions

Researchers are increasingly applying Bayesian hierarchical modelling techniques to visualise high-resolution spatial patterns of health indicators. These techniques offer powerful and rigorous methods to quantify and visualise model uncertainty, but few studies have explored how to communicate this uncertainty in policy-relevant settings. Here, I explored how model uncertainty changes with increasing spatial resolution, and found that while uncertainty increases with higher

spatial resolutions, model precision was best approximated at the highest spatial resolution, suggesting an important policy trade-off between identifying concealed spatial heterogeneities in health indicators. In modelling delivery via c-section at multiple spatial resolutions, I demonstrate poverty to be negatively correlated across spatial resolutions, suggesting important disparities in obtaining life-saving obstetric surgery persist across sociodemographic factors. This work is the first study to explore modelled c-section estimates and uncertainty at varying spatial resolutions, and has potential policy implications in terms of visualising spatial patterns of obstetric surgery, as well as focusing maternal and newborn health data collection efforts within Tanzania.

6.7 Acknowledgments

This work forms part of the output of WorldPop (<u>www.worldpop.org</u>) at the University of Southampton. The authors would like to acknowledge the support provided by the Economic and Social Research Council's (ESRC) Doctoral Training Programme, which funds CWR.

6.8 Intellectual contribution

The author list for this published work is as follows: Ruktanonchai CW, Nieves JJ, Ruktanonchai NW, Nilsen K, Steele JE, Matthews Z, Tatem AJ. CWR conceived of the study framework and methods for this study, performed all data management and analysis, generated all study visualizations and results interpretations, and wrote and prepared the manuscript for journal submission. JJN and NWR contributed insight and feedback on the conceptual study design, as well as statistical feedback on analysis and model results. KN and JES provided input on how the study results fit into the broader fields of maternal health and geography, respectively. Lastly, ZM and AJT supervised CWR during production of this manuscript, and oversaw that methods and analysis were executed within a sound and scientifically rigorous framework. All authors reviewed the manuscript and provided feedback before final submission to the journal.

Chapter 7: Conclusions

To ensure 'no one left behind' in improving the health of the most vulnerable women and children, the disaggregation of MNH outcomes at high spatial resolutions will be crucial over the coming decades (Molla et al., 2019). Despite this, the use of spatial statistics and geospatial methods in MNH research remains limited (Ebener et al., 2015). This thesis aimed to broadly address this knowledge gap by: 1) producing methods to estimate utilisation of key MNH services, highlighting high spatial resolution and policy-relevant inequalities; 2) tracking how these spatial inequalities in utilisation of MNH services have changed over time, identifying sub-national regions which have made progress in narrowing inequalities over time, as well as regions where progress still needs to be made; and, 3) exploring the trade-off between increasing spatial resolution of modelled estimates and associated uncertainty in model outcomes. In this section, I discuss the key findings of this work, followed by contributions this work has made to academic literature, as well as limitations of these analyses and future work. Lastly, I outline the policy and development implications of this work, followed by concluding remarks.

7.1 Key findings

Table 7.1 outlines a synthesis of key research findings of this thesis, by overall study objective and subsequent research question addressed in each analysis. As each study objective was addressed by a single paper, these are designated as 'Paper 1', 'Paper 2', and 'Paper 3'. Table 7.1 further outlines the key gaps in knowledge related to each study objective, as well as the data and methods used in each analysis. Addressing Study Objective 1 in Paper 1 (Chapter 4), I found that lower wealth and education levels within the East Africa region were associated with less odds of obtaining MNH care, while higher geographic inaccessibility scores were associated with the strongest effect in lowering odds of obtaining different types of care, with the widest disparities seen for skilled birth attendance. Kenya and Tanzania showed generally lower probabilities of obtaining key MNH services, while Rwanda and Burundi had generally higher odds of women utilising these services, possibly due to relatively higher density of health facilities.

Addressing Study Objective 2 in Paper 2 (Chapter 5), Rwanda represented the only country to increase coverage over time amongst all MNH services examined, with as much as 85% increases in SBA for some districts and nearing universal coverage for both SBA and PNC. Conversely, Tanzania experienced noticeable reductions in coverage throughout most of the country in utilisation of ANC, with some areas seeing as much as a 55% reduction, driven in part by exceedingly high estimates in earlier survey years. Across all countries, the relative gap in coverage within countries tended to be the most equitable for ANC coverage, while SBA and PNC tended to have greater gaps between the highest and lowest district estimates within countries, suggesting

Chapter 7

greater spatial disparities. Over time, however, the region has generally made progress in reducing spatial disparities within countries, yet improvement in PNC coverage has stagnated, and must continue to be monitored over the coming decades.

Addressing Study Objective 3 in Paper 3 (Chapter 6), increasing model uncertainty was generally associated with increasing spatial resolution, as quantified by increasing 95% credible interval widths. Despite this, the model showing the greatest precision was the model with the highest spatial resolution at 5km. Further, both c-section prevalence as well as model uncertainty tended to be higher in urban areas, potentially reflecting greater variance within the data, most likely due to more data availability within large population centres. These findings suggest an important and often overlooked trade-off persists between modelling these high-resolution health indicators and corresponding policy relevance if these estimates tend to be highly uncertain. Lastly, these methods can be applied to other regions throughout the world, and represent an integration between geolocated survey data and geospatial covariates that can reliably map MNH coverage at high spatial resolutions and across time. By applying these methods, researchers and policy makers can track spatial inequalities in key health coverage indicators throughout the SDG timeframe.

Table 7.1 Key research findings, by study objective

Study	Research questions	Gaps in knowledge	Data & methods	Key findings
objectives				
Paper 1: Produce high- resolution and policy relevant probabilities of utilisation of MNH services as predicted by geographic accessibility, highlighting spatial heterogeneity and sub- national inequalities	 What are the primary drivers of utilisation of antenatal care, skilled birth attendance, and postnatal care within the East African Community study region? Does geographic accessibility to a health facility predict utilisation of these services, and does this vary by service utilised (i.e., ANC, SBA, versus PNC)? Provided current health facility locations throughout the study region, where are women most likely to utilise these MNH services at both a high-resolution (300 x 300 m) scale and policy- relevant scale (administrative level II)? 	 Use of spatial statistics to identify previously hidden health inequalities is key to ensuring 'no-one left behind' in SDG era (WHO, 2015a) Spatial analysis of MNH outcomes remains limited (Ebener et al., 2015; Molla et al., 2019, 2017) No previous studies have explored geographic accessibility as a determinant across the spectrum of pregnancy at a high spatial resolution 	 Hierarchical mixed effects model to estimate odds of: 4+ antenatal care visits (ANC), skilled birth attendance (SBA), postnatal check-up within 48 hours (PNC) Fixed model effects included: urban/rural residence; education status; wealth quintile; maternal age; parity, and interaction term between age and parity Primary explanatory variable: Geographic accessibility to nearest health facility Most recent DHS data used, N = 25,325 women with birth in preceding 5 years 9,314 facilities used to create geographic accessibility surface (primary explanatory variable) 	 Across all outcomes, decreasing wealth and education levels were associated with lower odds of obtaining MNH care Increasing geographic inaccessibility scores were associated with the lowest odds of obtaining care observed across outcomes, with the widest disparities observed among skilled birth attendance Kenya and Tanzania had generally lower probabilities of obtaining all MNH services, with the lowest probabilities observed throughout rural districts in northern Kenya and central Tanzania Conversely, Rwanda and Burundi had generally higher probabilities of obtaining care were observed in urban versus rural districts, further indicating infrastructure density is important factor in predicting MNH care utilisation The integration of geo-located survey data with geospatial covariates in statistical models can reliably map coverage of MNH service utilisation at fine spatial scales

Study	Research questions	Gaps in knowledge	Data & methods	Key findings
objectives				
Paper 2: Track how spatial inequalities in utilisation of MNH services have changed over time, identifying sub- national regions which have made progress in narrowing inequalities over time, as well as regions where progress still needs to be made	 3. How has coverage in SBA, ANC, and PNC within study countries changed absolutely over time, by administrative unit II? Which administrative units have had the most improvement in coverage? Which have had the least improvement? 4. How does the change in coverage among these administrative units compare relative to each other? a. How wide is the change-in-coverage gap between administrative units within a country? b. Has this change-in-coverage gap improved over time, gotten worse, or stayed the same? 	 Spatial inequalities in access to life-saving MNH services persist within sub-Saharan Africa Previous studies have systematically reported child and maternal mortality at the global, regional, and national level (Alkema et al., 2016b; Black et al., 2010), and health indicators such child growth (Osgood-Zimmerman et al., 2018) at high spatial resolutions, but fewer studies have examined temporal trends in utilisation of key MNH services at similar spatial resolutions While Paper 1 examined utilisation of MNH services as an emergent property of accessibility at high spatial resolutions, no research has examined how these spatial inequalities have evolved over time at similar spatial scales 	 Compiled data from DHS for Kenya, Tanzania, Rwanda, and Uganda for several time points available (see Table 5.1) Employed a Bayesian model framework using the Integrated Nested Laplace Approximation (INLA) package in R software to spatially interpolate coverage estimates for ANC, SBA, and PNC at the district level Fixed model effects included: urban/rural residence; education status; wealth quintile; maternal age; parity Absolute inequality: compared estimates for each country between the first and last surveys available Relative inequality: quantified the ratio between best-versus- worst modelled estimates among districts 	 Rwanda represented the only country with exclusively increasing coverage over time amongst all three services, with as much as 85% increases in SBA for some districts and nearing universal coverage for both SBA and PNC Conversely, Tanzania had noticeable reductions in coverage throughout most of the country in utilisation of ANC, with some areas experiencing as much as a 55% reduction Across all countries, the relative gap between countries tended to be smallest among ANC (with the exception of Rwanda), while SBA and PNC tended to have greater gaps between the highest and lowest district estimates, suggesting greater spatial heterogeneity Historically, Rwanda had relatively small ratios which decreased over time, while Kenya had the highest inequalities in SBA, these were substantially reduced over time While the region is generally making progress in reducing spatial gaps across districts, improvement in PNC coverage has stagnated These methods can be applied over time to monitor progress in reducing absolute and relative inequalities in MNH service utilisation

Study	Research questions	Gaps in knowledge	Data & methods	Key findings
objectives				
Paper 3: Explore the trade-off between increasing spatial resolution of modelled estimates of delivery via caesarean section in Tanzania, and associated uncertainty in model outcomes	 What is the trade-off between increasing spatial resolution and model fit? How does the overall posterior distribution of the 95% credible interval change with increasing spatial resolution? How does uncertainty in estimates of prevalence of delivery via c-section propagate spatially across the landscape at varying resolution sizes? Are there areas with consistently high or low estimates of delivery via c-section across spatial resolutions? Are there areas with consistently high or low uncertainty across spatial resolutions? 	 Advancements in computational resources and data availability have allowed researchers across disciplines to employ Bayesian geostatistical additive models (GAM) to map disease and quantify uncertainty in posterior model outcomes, particularly using hierarchical clustered data such as from the DHS (Blangiardo and Cameletti, 2015; Gething et al., 2015; Magalhães and Clements, 2011; Patil et al., 2011; Schur et al., 2011) GAMs have improved model parameter estimates and precision among spatially correlated and rare adverse health outcomes (Gething et al., 2015; Kirby et al., 2017) No studies have explored how increasing spatial resolution impacts model estimates and uncertainty (Goovaerts, 2006) 	 DHS data for Tanzania, 2015, N = 7,050 women with birth in preceding 5 years Employed a Bayesian hierarchical model framework with input covariates at the 5km, 50km, and 100km spatial scales Implemented via stochastic partial differential equation (SPDE) spatial regression approach, using the Integrated Nested Laplace Approximation (INLA) technique Binary outcome of interest: woman underwent any delivery via c-section for a birth within the preceding five years Environmental covariates: accessibility to major cities, night- time lights, live births, Multidimensional Poverty Index estimates, and travel time to nearest hospital 	 Overall, increasing model uncertainty was associated with increasing spatial resolution, as quantified by increasing 95% credible interval widths Despite increasing uncertainty which accompanied increasing spatial resolution, 5km model had greatest model precision Findings suggest an important and often overlooked trade-off persists between modelling these high-resolution health indicators and corresponding policy relevance if these estimates tend to be highly uncertain Both c-section prevalence as well as model uncertainty tended to be higher in urban areas, reflecting greater variance within the data, most likely due to more data availability within large population centres

Chapter 7

7.2 Thesis contributions

This work explores the spatial inequalities in accessibility to key MNH services in the East Africa region, how these inequalities have evolved over time, and the trade-off between modelled estimates at high spatial resolutions and model uncertainty. In doing so, this work makes several novel contributions to the scientific literature.

The analyses comprising Chapter 4 represented the first of their kind to model the probability of obtaining antenatal care, skilled birth attendance, and postnatal care at high spatial resolutions in the East African region. Further, while previous studies have examined geographic accessibility to health facilities in the context of obtaining emergency care, no studies have reported the probability of obtaining routine MNH services, as predicted by geography. There has consequently been a call for using geospatial analysis and GIS tools to explicitly account for the effect of geographic variation in key MNH outcomes and service utilisation (Makanga et al., 2016). These analyses were therefore also the first study to model these outcomes as an emergent property of geographic accessibility using a geo-located database of nearly 10,000 health facilities throughout the region. Lastly, the geo-located database assembled for this study also represented the first of its kind to be openly available to researchers with detailed information on location, operational status, public/private ownership, and facility type. These results demonstrate how novel spatial approaches can be utilised to inform policy efforts and promote evidence-based decision-making, making substantial contributions to both the academic literature in addition to policy contributions.

While the analyses outlined in Chapter 4 examined spatial inequalities in MNH service utilisation, no research has similarly looked at how these spatial inequalities have evolved over time, particularly at high spatial resolutions. Previous research has examined inequalities in key MNH outcomes, including ANC, SBA, and PNC, disaggregated by socio-economic factors such as wealth quintile and education (WHO, 2015a) and geographical factors such as DHS region (Assaf and Pullum, 2016). However, no research has disaggregated these inequalities over time at policyrelevant, high spatial resolutions such as the administrative level II unit. The analyses comprising Chapter 5 therefore represent the first to report temporal trends in spatial inequalities among these services within the region at policy-relevant scales, particularly in the context of both absolute and relative health inequalities. Further, while previous work has reported temporal trends in inequalities as measured by logit regression coefficients such as wealth quintile (Assaf and Pullum, 2016), these analyses were the first to report logit regression coefficients of the effect of DHS region, representing temporal trends in spatial inequalities, divorced of modelled estimates. By reporting trends in district-level MNH service utilisation throughout these study countries, this work contributed to the academic literature, and further established a baseline of evidence moving forward into the Sustainable Development Goal era.

Lastly, while the use of Bayesian geostatistical approaches have allowed for the estimation of MNH service utilisation at high spatial resolutions, no previous research has examined the optimal spatial resolution to model outcomes or the trade-off between increasing spatial resolution in model inputs and resulting model uncertainty. The analyses comprising Chapter 6 of this work were the first to report on model uncertainty resulting from increasing spatial resolution inputs. The results of this work suggest an important trade-off between identifying concealed spatial heterogeneities and accuracy of estimates, which should be communicated in policy relevant settings.

7.3 Limitations

The analyses comprising this work are broadly subject to limitations, as outlined within each corresponding chapter. Notably, DHS data were used throughout all analyses in this work, and are subject to a variety of limitations. Firstly, statistical inference was made at administrative units smaller than the geographical units at which these data are representative, potentially resulting in model uncertainty and sampling error. However, the DHS has provided guidance on how to perform such inference in a robust and statistically valid way (Gething et al., 2015), endorsing the use of Bayesian geostatistical methods which can quantify the resulting uncertainty. Secondly, women with births in the preceding five years were used throughout these analyses, potentially representing a time lag between data collection and analysis. Therefore, these analyses may not represent the current state of MNH service utilisation, instead representing trends up to five years previous. As these limitations are inherent to DHS data, they potentially impact results reported throughout all three analyses.

In addition to the above broad limitations, each analysis is also subject to its own limitations specific to the methods used. Firstly, within the analyses comprising Chapter 4, some health facilities were not included in the geographic accessibility analysis. These facilities were excluded as they were deemed unlikely to offer maternal and newborn health services, but it is possible that some facilities did indeed offer services and were therefore inaccurately excluded. In this scenario, geographic accessibility would be inaccurately decreased for some participants, introducing error into model results. Further, I assumed accessibility to the nearest health facility in these analyses, which is very often not the reality of the situation, as many women bypass the nearest health facility in lieu of facilities offering higher quality of care or as a result of an emergency referral (De Allegri et al., 2011; Hulton et al., 2016). While actual health facility used would be the ideal input into these geographic accessibility analyses, these data were not available for these study countries. Secondly, the outcomes used in this analysis varied by country, specifically in regards to skilled birth attendance. While women were asked to recall whether a skilled attendant was present at delivery, it is possible women were not aware of the qualifications of their attendant, potentially introducing recall bias into these analyses. Further, even though these analyses use the WHO

definition of skilled birth attendance as trained 'doctors, nurses, and midwives', qualifications and titles may vary between countries, further resulting in error (Adegoke et al., 2011).

Within the analyses comprising Chapter 5, the time lag inherent to DHS data may have a particular impact on temporal trend analyses. Further, while the most recent DHS data were used in these study countries, many countries did not have data available for the past 5 years, and indeed Burundi had only one time point available and could therefore not be included in these analyses. More recent data representing births within a timelier window is necessary to confirm the trends reported. Lastly, similar to Chapter 4, actual health facility used was not used in these analyses, as these data were not available. In cases where women may be referred to a district hospital or may travel long distances to give birth elsewhere (such as when staying with family, etc.), error is potentially introduced into the models, as where MNH services were utilised does not necessarily reflect the district where the respondent was surveyed at the time of data collection.

Lastly, the suite of covariates used in the analyses comprising Chapter 6 were not rigorously tested as predictors of delivery via c-section, as the scope of the study was to explore the trade-off between increasing spatial resolution and model uncertainty. While these covariates have been used in previous studies, it is possible they do not predict delivery via c-section rigorously, and any model results should therefore be interpreted cautiously, as results may not be generalizable to the rest of the study region because model validation efforts were not undertaken. Further, the spatial resolutions chosen for these analyses represent a theoretical exploration of the impact of increasing resolution on modelled estimates, and are unlikely to represent estimates at resolutions at which policy decisions are made. Future work may explore the impact of increasing spatial resolution on modelled estimates at policy relevant administrative boundaries, as opposed to the rasters employed in these analyses. Secondly, similar to previous chapters, time to health facility represented time to the nearest health facility, potentially overlooking scenarios in which a woman bypasses the nearest health facility due to referral, higher quality of care, etc. Additionally, travel estimates are also subject to bias themselves, as mobility patterns vary by season, and information on road quality is not captured, excluding smaller footpaths which may be used in transit. Lastly, other environmental covariates such as night-time lights have associated misclassification bias due to light refraction, etc., potentially introducing further error into the models.

7.4 Future work

Future work should address the broad limitations inherent to these studies, as outlined above, with aims of strengthening data quality and analysis. Firstly, future work should include information on actual health facility used where possible, and examine how this compares to assumptions made regarding nearest health facility. Along these lines, mobility data reflecting actual movement patterns among pregnant women should be used, as collected by mobile phone technology, for

example. Previous work suggests that women living in high poverty, rural areas may be historically under-represented in mHealth program interventions that utilize mobile phone ownership to target and access hard to reach, remote populations (Wesolowski et al., 2012). While some work has explored mobility patterns among more broad populations using mobile phone technology such as call data records (CDRs), this work tends to employ anonymized datasets for security and privacy reasons (González et al., 2008; Lai et al., 2019). More recently, Google Location History (GLH) data has been explored as a potential novel source to link mobile phone usage patterns and data on sociodemographic and health surveys (N. W. Ruktanonchai et al., 2018), but no work has explored mobility patterns among pregnant women specifically, particularly within the context of health facility accessibility. Future work should therefore aim to collect individual level health and demographic data on pregnant or recently pregnant women, in combination with actual mobility patterns to health facilities used throughout the spectrum of pregnancy.

Secondly, while nationally representative surveys such as DHS and MICS surveys offer researchers an opportunity to follow health over time, these data are collected infrequently every 3 to 5 years and are therefore subject to time lag, decreasing their suitability for SDG monitoring. Data collection efforts can be further hampered by political instability and conflict, as evidenced by a lack of DHS data in Burundi, preventing its inclusion in analyses performed in Chapter 5. Lastly, these surveys frequently evolve over time, potentially limiting their comparison over time, as well. To overcome these issues, routine health management information systems (HMIS) and national registry data regularly collected by ministry of health programmes offer a potential way forward.

A highly functioning HMIS will collect information on health systems resources, such as personnel, financing, and infrastructure, that can then be used to measure progress in key health indicators, compile a variety of data sources including census data and nationally representative survey data, manage data to aggregate forms such as health catchments and administrative units, and disseminate data products to key stakeholders (Health Metrics Network, 2008). While many countries struggle to compile a functional HMIS due to competing priorities and limited resources, other countries such as Kenya, Sierra Leone, South Africa, Malawi, and Zanzibar have successfully integrated their HMIS into an online, open-source platform such as the DHIS2.

The aim of DHIS2 implementation is ultimately to "generate, analyse, and disseminate health information to facilitate effective policy formulation, management, planning, budgeting, implementation, monitoring and evaluation of health services and program interventions in the health sector" (Karuri et al., 2014). Ideally, this system should be implemented with local and national support, and include a collaborative network of scientists and researchers who can disseminate analysis to stakeholders appropriately. Further, to ensure successful consolidation of routinely collected health data, quality and standardization of data must occur, as evidenced by successful implementation in Sierra Leone and South Africa (Karuri et al., 2014). Without ensuring data quality, researchers are often left to utilise surrogate methods to account for missing and

unreliable data, such as aggregating to operational administrative units and utilising Bayesian inference (Alegana et al., 2016). As the world continues to move closer a 2030 deadline, it is vital that future efforts ensure that otherwise fragmented surveillance data and irregularly collected survey data are consolidated and standardized into similar DHIS2 platforms, allowing for monitoring and evaluation of key SDG indicators over time.

Lastly, future work should prioritise surveillance and geo-coding of maternal deaths and strengthening mortality statistics. In order to strengthen these systems, maternal deaths occurring outside of health care centres and hospitals must be recorded, geographic coordinates must be collected, and data must be checked to ensure quality control. These data should be captured by routine health systems data, and available to ministries of health and technical experts, as discussed above. Once these data are available, routine visualisation and monitoring and surveillance efforts can occur to better understand where inequalities in maternal deaths persist, helping to focus interventions. Until countries are able to visualise where maternal deaths are occurring, policy makers are left to best estimate where resources and intervention efforts are best applied.

7.5 Policy and development implications

The analyses comprising this thesis have several policy and development implications, as evidenced by its inclusion in region-specific policy briefs. Ensuring that SDG targets are met by the year 2030 necessitates ensuring that the hidden pockets of the most vulnerable women and children are identified and addressed. Towards this, no single intervention is sufficient in preventing unnecessary deaths and ill health among these populations, but instead universal health coverage and access to high quality of care is essential over the coming years. However, in the absence of universal health coverage indices, key intervention indicators will be crucial in monitoring health coverage and resulting inequalities (World Health Organization and International Bank for Reconstruction and Development/The World Bank, 2017). It is therefore crucial that researchers and governments monitor the change in coverage over time in these indicators at sub-national levels to ensure that progress is made not just at national levels, which can hide important disparities within countries, but good health is achieved amongst all women and children, leaving no one behind.

These analyses demonstrate how spatial approaches can be used to inform monitoring and evaluation efforts and promote evidence-based policy making decisions. These applications, however, must be adopted and maintained by experts within the region in order to ensure analyses are relevant and sustainable over the course of decades (Tanser and le Sueur, 2002). It is therefore crucial to ensure ongoing capacity strengthening and knowledge exchange efforts are maintained. Towards this, the analyses comprising Chapter 4 of this work were informed by key interventions monitored by the intergovernmental East Africa Community's RMNCAH Scorecard Framework, a

tool for tracking and communicating progress in key RMNCAH interventions to key stakeholders in the region. More importantly, the results of these analyses were communicated back to collaborators through in-country workshops, such as *The Workshop and Technical Exchange on the use of advanced Geographic Information System (GIS) enabled mapping and validation of the 2nd annual EAC RMNCAH Scorecard*, which I hosted in 2016. By building and maintaining these collaborative networks and capacity strengthening efforts, the spatial approaches outlined in this body of work may be sustained within country by local experts with the appropriate technical and cultural knowledge to monitor and evaluate progress. This is further evidenced within the regionspecific policy brief, *The State of the World's Midwifery: Analysis of the Sexual, Reproductive, Maternal, Newborn and Adolescent Health Workforce in East & Southern Africa.* Here, these analyses demonstrated the utility of GIS tools in highlighting sub-national inequalities for use in decision making. By collaborating with policy makers at EAC throughout the entirety of this work, strong collaborative networks were built, facilitating and complementing ongoing in-country efforts to monitor and strengthen reproductive, maternal, newborn, adolescent, and child health efforts.

The results of these analyses have further policy implications that should be communicated back to stakeholders and decision makers within the region. The results of Chapter 4 suggest that allocating funds for increased infrastructure such as health facilities, health workforce, and road networks, as well as upgrading existing infrastructure, may help in alleviating some of the inequalities observed within the region. Specifically, funds can be allocated towards efforts to improve geographic accessibility to health facilities, such as paving road networks, increasing transportation services, and building bridges in areas likely to flood in the rainy season. However, increasing accessibility alone is insufficient in increasing service utilisation, particularly among services such as skilled birth attendance where unplanned emergencies may increase barriers to utilising these services. Ultimately, investing in infrastructure such as increased density and higher quality of health facilities and health workforce will be crucial in reducing adverse MNH outcomes amongst resource-poor settings. By highlighting sub-national and high spatial resolution inequalities within the region, policy makers can better allocate resources and funds to areas of the region most at-risk of low service utilisation, ultimately ensuring no gaps in health coverage across the continuum of pregnancy and delivery.

The analyses comprising Chapter 5 of this work aimed to monitor the progress of coverage of these services over time, highlighting areas within the region where progress in increasing coverage was being made, as well as areas where work was still needed. This represented the first study to examine change in spatial inequalities over time at these spatial resolutions, where previous studies have used aggregate outcomes masking underlying spatial variance. Basing policy on such aggregations is problematic, is not resource efficient, and misses those most in need. The results of these analyses therefore have important policy implications, and suggest that progress in reducing

disparities in service utilisation has varied across the region, with some countries performing better than others in reducing coverage gaps.

In particular, Rwanda has made exceptional progress in ensuring high universal coverage across all MNH services studied, as well as ensuring progress is made across the country in a spatially equitable manner. Further, as the only country within the region to achieve MDG targets of reducing maternal mortality by 75% between 1990 and 2015, the country represents a potential case study in how to prioritize and improve maternal and child health. Specifically, Rwanda has radically redeveloped their health system, aiming to implement tiered national health insurance schemes available to even residents unable to afford insurance, performance-based pay schemes to incentivize high quality health workforce, and coordination between government aid and external donors (Logie et al., 2008). While these policies may not be suitable for other countries within the region, they demonstrate a commitment to improving RMNCAH over the previous decades, suggesting that aggressive investment in women and children's health and human rights at the national level is necessary to reducing preventable deaths and adverse health outcomes.

Lastly, the analyses comprising Chapter 6 of this study suggest caution in interpreting results without fully comprehending the associated uncertainty in model estimates. While estimates of MNH service utilisation and health outcomes is increasingly necessary in identifying hidden spatial inequalities, the results of this chapter suggest uncertainty also increases with increasing spatial resolution. This represents an important trade-off between increasing spatial resolution and accuracy of model estimates, suggesting that researchers and policy makers should collaborate on the optimal geographical unit to predict health outcomes and service utilisation. This further has important policy implications in the context of communicating uncertainty to policy makers, which is not an intuitive or approachable task, even amongst researchers. While the ultimate decision of how, when and where to allocate resources and funds lies with policy makers, researchers have a responsibility to communicate the limitations of such analyses to these decision makers, else limited resources are at risk of being inappropriately and inefficiently distributed.

The collective results comprising this work suggest that collaboration and buy-in among policy makers is crucial to ensuring the appropriate research questions are addressed and results are interpreted in a culturally competent way. It is not the role nor right of the researcher to make distinct policy recommendations, particularly among countries where the researcher lacks context. Instead, policy makers and key stakeholders should be consulted throughout all phases of the research development, from inception to study objectives to funding to interpretation and dissemination. Without the participation and engagement of the representatives most affected by these health challenges, research will continue to be isolated, unsustainable, and patronizing. Examples of knowledge exchange activities could include capacity strengthening workshops aimed at increasing technical knowledge, participatory focus groups between researchers and key stakeholders, joint funding endeavours shared between researchers and policy institutions, and

increased coordination and communication amongst in-country health sectors and academic institutions.

7.6 Conclusions

Progress in maternal and newborn health (MNH) outcomes has historically been monitored at aggregate levels such as the global, regional and national level, yet these measures show vast inequalities sub-nationally. While previous work has disaggregated MNH outcomes by socioeconomic factor such as wealth, quintile, place of residence, etc., (Assaf and Pullum, 2016; WHO, 2015a) less work has aimed to visualise MNH outcomes and service utilisation at high spatial resolutions and policy-relevant scales such as the district level, particularly within a framework explicitly accounting for geographic variation. This work utilised geospatial approaches such as cost-distance analysis, mixed effects hierarchical modelling, and Bayesian geostatistical models to visualise disparities in key MNH services throughout the East Africa region at multiple spatial resolutions. Further, this work examined how these spatial disparities have evolved over time within these countries, using both absolute and relative inequality indices, with aims of highlighting where improvement has been made in closing the gaps in these spatial disparities, and where progress still needs to be made. Lastly, this work examined the trade-off between reported model outcomes at increasing spatial resolutions and associated model uncertainty. The analyses contained in this thesis have been published in rigorous academic journals, as well as featured in regional policy briefs, making substantial contributions to the academic literature and policy spheres. The findings contained in this work demonstrate the utility of spatial approaches in identifying previously hidden MNH inequalities, which can inform and direct policy efforts and evidence-based decision-making. These approaches will become increasingly critical in the post-2015 Sustainable Development Goal era, where new approaches in identifying and reducing persistent inequalities will be key to ensure no one is left behind.

Appendices

Appendices

Appendix A Chapter 4 Appendix

A.1 Chapter 4 Supplementary Figures

Figure A.1 Geo-located DHS clusters (N = 3,311) by number of DHS respondents (N = 36,178) and urban versus rural location in five East African countries.



Figure A.2 Estimated live births in five East African countries. Birth estimates were generated by the WorldPop project (www.worldpop.org) and are shown for the year 2015 at 100 x 100 m resolution.

Appendix B Chapter 5 Appendix

B.1 Chapter 5 Supplementary Tables

Table B.2 Unadjusted logistic regression results, with worst performing DHS region as reference, as compared to best performing DHS region.

Outcome	Year	Worst performing Region (Reference)	Best performing region (Coefficient)	Model coefficient (±SD)		
Kenya						
SBA		North Eastern	Nairobi	3.33 (0.22)		
ANC	2003	North Eastern	Nairobi	2.80 (0.21)		
PNC		North Eastern	Nairobi	3.38 (0.23)		
SBA		Western	Nairobi	2.65 (0.19)		
ANC	2008	North Eastern	Nairobi	1.78 (0.18)		
PNC		North Eastern	Nairobi	2.34 (0.29)		
SBA		North Eastern	Nairobi	2.76 (0.18)		
ANC	2014	North Eastern	Nairobi	1.55 (0.13)		
PNC		North Eastern	Central	1.77 (0.32)		
Outcome	Year	Worst performing Region (Reference)	Best performing region	Model coefficient (±SD)		
		Tar	nzania			
SBA		Kigoma	Dar Es Salaam	2.37 (0.49)		
ANC	1999	Kigoma	Dar Es Salaam	3.22 (0.6)		
PNC		Kigoma	Dar Es Salaam	1.27 (0.5)		
SBA		Pemba North	Dar Es Salaam	2.82 (0.27)		
ANC	2010	Pemba North	Dar Es Salaam	2.09 (0.23)		
PNC		Mwanza	Ruvuma	1.05 (0.51)		
SBA		Pemba South	Dar Es Salaam	1.62 (0.21)		
ANC	2015	Pemba North	Dar Es Salaam	2.09 (0.21)		

Outcome	Year	Worst performing Region (Reference)	Best performing region	Model coefficient (±SD)		
	Rwanda					
SBA		Gikongoro	Kigali	2.53 (0.17)		
ANC	2005	Kigali Ngali	Cyangugu	1.03 (0.22)		
PNC		Gikongoro	Kigali	2.49 (0.17)		
SBA		North	South	0.15 (0.08)		
ANC	2010	Kigali	South	0.08 (0.09)		
PNC		Kigali	South	-0.65 (0.22)		
SBA		North	South	-0.13 (0.16)		
ANC	2014	Kigali	South	0.49 (0.09)		
PNC		Kigali	South	0.37 (0.18)		
Outcome	Year	Worst performing Region (Reference)	Best performing region	Model coefficient (±SD)		
		Ug	anda			
SBA		Northern	Central	-0.72 (0.12)		
ANC	2000	Northern	Central	0.38 (0.11)		
PNC		Northern	Central	1.78 (0.12)		
SBA		West Nile	East Central	0.91 (0.14)		
ANC	2006	North	Central 1	0.14 (0.13)		
PNC		West Nile	East Central	-0.36 (0.32)		
SBA		Southwest	East Central	1.03 (0.13)		
ANC	2011	Central 1	Karamoja	0.23 (0.14)		
PNC		Southwest	East Central	0.08 (0.21)		

	SBA	ANC	PNC	
Fixed Effects	Mean Posterior Estimate (SD)			
	Kenya			
	2003			
Intercept	7.39 (2.33)	-1.83 (2.01)	4.49 (2.57)	
Rural/Urban	-0.04 (0.45)	-0.12 (0.39)	0.23 (0.5)	
Education	-1.02 (0.44)	-1.1 (0.42)	-1.4 (0.48)	
Wealth	-1.29 (0.57)	-0.45 (0.5)	-1.4 (0.62)	
Age	-0.13 (0.07)	0.07 (0.06)	-0.07 (0.08)	
Births	-2.01 (0.7)	0.24 (0.63)	-1.25 (0.77)	
DIC	277.06	275.07	277.76	
MLL	-141.21	-139.50	-145.49	
	2008			
Intercept	5.14 (2.53)	0.18 (1.93)	4.47 (2.29)	
Rural/Urban	-0.51 (0.65)	-0.52 (0.46)	-0.48 (0.58)	
Education	-0.32 (0.62)	-0.34 (0.42)	-0.86 (0.55)	
Wealth	-1.96 (0.7)	-0.85 (0.54)	-1.82 (0.63)	
Age	-0.05 (0.08)	0.03 (0.06)	-0.04 (0.07)	
Births	-1.59 (0.93)	-0.19 (0.65)	-1.44 (0.85)	
DIC	282.26	288.76	281.75	
MLL	-148.14	-144.44	-143.96	
	2014			
Intercept	0.95 (3.7)	3.61 (2.56)	0.92 (3.83)	
Rural/Urban	-0.16 (0.7)	-0.55 (0.46)	-0.14 (0.72)	
Education	-0.67 (0.58)	-0.22 (0.42)	-0.91 (0.61)	
Wealth	-2.26 (0.78)	-0.43 (0.51)	-2.29 (0.81)	
Age	0.09 (0.12)	-0.09 (0.08)	0.08 (0.13)	
Births	-1.16 (1.09)	-0.11 (0.72)	-0.98 (1.14)	
DIC	357.55	361.77	356.08	
MLL	-205.56	-197.33	-205.71	
	SBA	ANC	PNC	
Fixed Effects	Mean	n Posterior Estimate	(SD)	
	Tanzania			
	1999			
Intercept	3.97 (1.48)	2.18 (1.13)	4.25 (1.39)	
Rural/Urban	-0.96 (0.32)	0.01 (0.24)	-0.61 (0.31)	
Education	-1.48 (0.54)	-0.56 (0.4)	-1.55 (0.5)	
Wealth	-1.05 (0.42)	-0.37 (0.33)	-1.21 (0.4)	
Age	-0.02 (0.05)	-0.001 (0.04)	-0.01 (0.05)	
Births	-1.27 (0.41)	-0.83 (0.32)	-1.47 (0.39)	
DIC	422.80	422.09	414.53	
MLL	-163.80	-151.68	-156.03	
	2010			
Intercept	6.94 (1.31)	1.44 (0.9)	7.55 (1.39)	
Rural/Urban	-0.87 (0.34)	-0.43 (0.23)	-0.93 (0.36)	
Education	-1.81 (0.56)	-0.27 (0.39)	-2.04 (0.59)	
Wealth	-1 (0.44)	-0.5 (0.31)	-0.93 (0.46)	
Age	-0.08 (0.04)	0.04 (0.03)	-0.1 (0.04)	
Births	-1.92 (0.38)	-1.48 (0.27)	-1.74 (0.4)	
DIC	644.04	652.56	638.43	

Table B.3 Model fit, mean posterior estimates and hyperparameters of modelled MNH outcomes.

Appendix B

	207.00	200.00	200.45
MLL	-297.88	-280.60	-299.45
	2015		
Intercept	4.41 (1.21)	2.07 (0.95)	7.25 (1.47)
Rural/Urban	-0.46 (0.28)	-0.24 (0.21)	-0.68 (0.34)
Education	-2.52 (0.56)	0 (0.44)	-3.3 (0.66)
Wealth	-0.72 (0.35)	-0.53 (0.28)	-0.64 (0.42)
Age	0 (0.04)	0.01 (0.03)	-0.03 (0.05)
Births	-1.69 (0.35)	-1.29 (0.28)	-2.59 (0.42)
DIC	682.95	715.99	641.95
MLL	-312.88	-318.71	-298.86
	Rwanda		
	2005		
Intercept	1.32 (2.15)	-5.84 (2.68)	0.32 (2.21)
Rural/Urban	-1.43 (0.34)	-0.29 (0.4)	-1.57 (0.35)
Education	-1.9 (0.92)	-2.92 (1.17)	-1.57 (0.94)
Wealth	0 (31.62)	0 (31.62)	0 (31.62)
Age	0.02 (0.05)	0.05 (0.07)	0.05 (0.05)
Births	-0.63 (0.75)	2.13 (0.91)	-0.59 (0.76)
DIC	169.60	144.47	170.54
MLL	-79.95	-65.33	-80.75
	2010		
Intercent	0.96 (3.07)	-34(339)	2.94 (3.06)
Rural/Urban	-0.7 (0.63)	-1 22 (0.66)	-0.03(0.63)
Education	-1 66 (1 19)	-1.05(1.33)	-2.02(1.18)
Wealth	-1.00(1.17) -0.98(0.95)	1.05(1.55) 1 76(0.99)	-2.02(1.10) -1.24(0.96)
Age	-0.98(0.93)	1.70(0.77) 0.14(0.09)	-1.24(0.90)
Birtha	0.05(0.00)	0.14(0.0)	-0.01(0.00)
DIC	181.60	-0.09 (1.1)	-0.77 (1.01)
	101.09	162.96	160.2
MLL	-95.76	-90.71	-92.80
Testamand	2014	2(1(2(0)))	0.04 (2.07)
Intercept	-0.82 (4.49)	-3.64 (2.66)	0.84 (3.97)
Rural/Orban	-0.67 (1)	-0.45 (0.57)	1.11 (0.87)
Education	-4.75 (2.42)	-2./3 (1.42)	-1.18 (2.15)
Wealth	-0.33 (1.45)	1.48 (0.89)	-1.57 (1.33)
Age	0.11 (0.13)	0.05 (0.07)	0.06 (0.11)
Births	1.11 (1.85)	1.52 (1.06)	-0.31 (1.6)
DIC	137.26	179.75	148.89
MLL	-64.22	-90.14	-72.30
	SBA	ANC	PNC
Fixed Effects	Mear	n Posterior Estimate	(SD)
	Uganda		
	2000		
Intercept	-0.7 (1.49)	0.91 (1.11)	4.61 (1.28)
Rural/Urban	-1.18 (0.37)	-0.87 (0.29)	-1.72 (0.34)
Education	-0.91 (0.67)	-0.61 (0.49)	-1.25 (0.55)
Wealth	0.48 (0.3)	-0.15 (0.23)	0.07 (0.25)
Age	-0.02 (0.05)	0 (0.03)	-0.08 (0.04)
Births	0.48 (0.44)	-0.13 (0.33)	-0.47 (0.37)
DIC	355.28	394.81	374.86
MLL	-115.51	-129.15	-121.04
	2006		
Intercept	2.59 (1.17)	0.54 (0.88)	2.55 (1.19)

Rural/Urban	-1.39 (0.53)	-0.6 (0.32)	-1.01 (0.54)
Education	-0.99 (0.54)	-0.06 (0.38)	-0.96 (0.56)
Wealth	-1.07 (0.42)	-0.07 (0.29)	-1.3 (0.44)
Age	0.01 (0.04)	0.02 (0.03)	0.02 (0.04)
Births	-0.81 (0.37)	-0.43 (0.3)	-0.99 (0.38)
DIC	420.75	444.63	417.22
MLL	-142.35	-144.13	-142.13
	2011		
Intercept	1.86 (1.18)	0.46 (0.93)	1.63 (1.06)
Rural/Urban	-1.47 (0.4)	-0.45 (0.29)	-0.55 (0.34)
Education	-0.85 (0.52)	0.96 (0.4)	-0.43 (0.48)
Wealth	-1.39 (0.49)	-0.54 (0.36)	-1.26 (0.43)
Age	0.04 (0.03)	-0.01 (0.03)	0 (0.03)
Births	-0.36 (0.35)	0.1 (0.3)	-0.4 (0.33)
DIC	453.70	483.55	476.54
MLL	-161.03	-166.45	-169.32
Hypernarometer	SBA	ANC	PNC
i i y per par ameter	DD 11		Inte
(Spatial Precision)	Mea	n Posterior Estimate	(SD)
(Spatial Precision)	Mea. Kenya	n Posterior Estimate	(SD)
(Spatial Precision) 2003	<i>Mea.</i> <i>Kenya</i> 5.1 (1.8)	n Posterior Estimate 6.2 (2.2)	(SD) 3.6 (1.1)
(Spatial Precision) 2003 2008	Mea. Kenya 5.1 (1.8) 3.5 (1.1)	6.2 (2.2) 8.0 (3.2)	(SD) 3.6 (1.1) 4.8 (1.7)
Ityperparameter (Spatial Precision) 2003 2008 2014	Kenya 5.1 (1.8) 3.5 (1.1) 6.9 (6.4)	6.2 (2.2) 8.0 (3.2) 14.1 (9.2)	(SD) 3.6 (1.1) 4.8 (1.7) 6.9 (4.9)
Ityperparameter (Spatial Precision) 2003 2008 2014	Mea. Kenya 5.1 (1.8) 3.5 (1.1) 6.9 (6.4) Tanzania	6.2 (2.2) 8.0 (3.2) 14.1 (9.2)	(SD) 3.6 (1.1) 4.8 (1.7) 6.9 (4.9)
Ityperparameter (Spatial Precision) 2003 2008 2014 1999	Mea. Kenya 5.1 (1.8) 3.5 (1.1) 6.9 (6.4) Tanzania 2.0 (0.6)	6.2 (2.2) 8.0 (3.2) 14.1 (9.2) 5.8 (2.6)	(SD) 3.6 (1.1) 4.8 (1.7) 6.9 (4.9) 2.5 (0.8)
Ityperparameter (Spatial Precision) 2003 2008 2014 1999 2010	Mea. Kenya 5.1 (1.8) 3.5 (1.1) 6.9 (6.4) Tanzania 2.0 (0.6) 2.0 (0.4)	6.2 (2.2) 8.0 (3.2) 14.1 (9.2) 5.8 (2.6) 5.6 (1.5)	(SD) (SD) $(3.6 (1.1))$ $(4.8 (1.7))$ $(6.9 (4.9))$ $(2.5 (0.8))$ $(1.8 (0.4))$
Ityperparameter (Spatial Precision) 2003 2008 2014 1999 2010 2015	Mea. Kenya 5.1 (1.8) 3.5 (1.1) 6.9 (6.4) Tanzania 2.0 (0.6) 2.0 (0.4) 3.1 (0.6)	6.2 (2.2) 8.0 (3.2) 14.1 (9.2) 5.8 (2.6) 5.6 (1.5) 5.6 (1.3)	(SD) (SD) (SD) (SD) (SD) (1.1) $(4.8 (1.7)$ $(6.9 (4.9)$ $(2.5 (0.8)$ $(1.8 (0.4)$ $(2.1 (0.4)$
Ityperparameter (Spatial Precision) 2003 2008 2014 1999 2010 2015	Mea. Kenya 5.1 (1.8) 3.5 (1.1) 6.9 (6.4) Tanzania 2.0 (0.6) 2.0 (0.4) 3.1 (0.6) Rwanda	6.2 (2.2) 8.0 (3.2) 14.1 (9.2) 5.8 (2.6) 5.6 (1.5) 5.6 (1.3)	3.6 (1.1) 4.8 (1.7) 6.9 (4.9) 2.5 (0.8) 1.8 (0.4) 2.1 (0.4)
Ityperparameter (Spatial Precision) 2003 2008 2014 1999 2010 2015 2005	Mea. Kenya 5.1 (1.8) 3.5 (1.1) 6.9 (6.4) Tanzania 2.0 (0.6) 2.0 (0.4) 3.1 (0.6) Rwanda 36.7 (53.5)	n Posterior Estimate 6.2 (2.2) 8.0 (3.2) 14.1 (9.2) 5.8 (2.6) 5.6 (1.5) 5.6 (1.3) 111.4 (424.9)	3.6 (1.1) 4.8 (1.7) 6.9 (4.9) 2.5 (0.8) 1.8 (0.4) 2.1 (0.4) 25.9 (29.8)
Ityperparameter (Spatial Precision) 2003 2008 2014 1999 2010 2015 2005 2010	Mea. Kenya 5.1 (1.8) 3.5 (1.1) 6.9 (6.4) Tanzania 2.0 (0.6) 2.0 (0.4) 3.1 (0.6) Rwanda 36.7 (53.5) 8.29 (4.6)	6.2 (2.2) 8.0 (3.2) 14.1 (9.2) 5.8 (2.6) 5.6 (1.5) 5.6 (1.3) 111.4 (424.9) 5.9 (2.7)	(SD) (SD) (SD) (SD) (SD) $(1,1)$ $(4,8 (1,7)$ $(6,9 (4,9))$ $(2,5 (0,8))$ $(1,8 (0,4))$ $(2,1 (0,4))$ $(25.9 (29.8))$ $(25.9 (29.8))$ $(8.6 (4,7))$
Tryperparameter (Spatial Precision) 2003 2008 2014	Mea. Kenya 5.1 (1.8) 3.5 (1.1) 6.9 (6.4) Tanzania 2.0 (0.6) 2.0 (0.4) 3.1 (0.6) Rwanda 36.7 (53.5) 8.29 (4.6) 6.9 (6.4)	6.2 (2.2) 8.0 (3.2) 14.1 (9.2) 5.8 (2.6) 5.6 (1.5) 5.6 (1.3) 111.4 (424.9) 5.9 (2.7) 14.1 (9.2)	(SD) (SD)
Ityperparameter (Spatial Precision) 2003 2008 2014 1999 2010 2015 2005 2010 2014	Mea. Kenya 5.1 (1.8) 3.5 (1.1) 6.9 (6.4) Tanzania 2.0 (0.6) 2.0 (0.4) 3.1 (0.6) Rwanda 36.7 (53.5) 8.29 (4.6) 6.9 (6.4) Uganda	6.2 (2.2) 8.0 (3.2) 14.1 (9.2) 5.8 (2.6) 5.6 (1.5) 5.6 (1.3) 111.4 (424.9) 5.9 (2.7) 14.1 (9.2)	(SD) (SD) (SD) (SD) (SD) $(1,1)$ $(4,8 (1,7)$ $(6,9 (4.9)$ $(2,5 (0.8))$ $(1,8 (0.4))$ $(2,1 (0.4))$ $(25.9 (29.8))$ $(25.9 (29.8))$ $(3.6 (4.7))$ $(6.9 (4.9))$
Ityperparameter (Spatial Precision) 2003 2008 2014 1999 2010 2015 2005 2010 2010 2014 2005 2010 2014 2005 2010 2010 2010 2010 2010 2010 2010 2010 2010 2010	Mea. Kenya 5.1 (1.8) 3.5 (1.1) 6.9 (6.4) Tanzania 2.0 (0.6) 2.0 (0.4) 3.1 (0.6) Rwanda 36.7 (53.5) 8.29 (4.6) 6.9 (6.4) Uganda 1.2 (0.4)	6.2 (2.2) 8.0 (3.2) 14.1 (9.2) 5.8 (2.6) 5.6 (1.5) 5.6 (1.3) 111.4 (424.9) 5.9 (2.7) 14.1 (9.2)	3.6 (1.1) 4.8 (1.7) 6.9 (4.9) 2.5 (0.8) 1.8 (0.4) 2.1 (0.4) 25.9 (29.8) 8.6 (4.7) 6.9 (4.9) 1.9 (0.7)
Ityperparameter (Spatial Precision) 2003 2008 2014 1999 2010 2015 2005 2010 2014 2005 2010 2014 2005 2010 2010 2010 2010 2010 2010 2010 2000 2000 2006	Mea. Kenya 5.1 (1.8) 3.5 (1.1) 6.9 (6.4) Tanzania 2.0 (0.6) 2.0 (0.4) 3.1 (0.6) Rwanda 36.7 (53.5) 8.29 (4.6) 6.9 (6.4) Uganda 1.2 (0.4) 2.6 (0.9)	n Posterior Estimate 6.2 (2.2) 8.0 (3.2) 14.1 (9.2) 5.8 (2.6) 5.6 (1.5) 5.6 (1.3) 111.4 (424.9) 5.9 (2.7) 14.1 (9.2) 2.8 (1.1) 26.0 (33.8)	(SD) (SD)

B.2 Chapter 5 Supplementary Figures









Figure B.3 Absolute change in a) 4+ antenatal care visits (green), b) postnatal care check-up within 48 hours (red), and c) skilled birth attendance (blue), Kenya DHS data, 2003 – 2014, ordered by administrative II unit.





154



Figure B.4 Absolute change a) 4+ antenatal care visits (green), b) postnatal care check-up within 48 hours (red), and c) skilled birth attendance (blue), Tanzania DHS data, 1999 – 2015, ordered by administrative II unit.





157



Figure B.5 Absolute change a) 4+ antenatal care visits (green), b) postnatal care check-up within 48 hours (red), and c) skilled birth attendance (blue), Rwanda DHS data, 2005 – 2014, ordered by administrative II unit.

Figure B.6 Absolute change in a) 4+ antenatal care visits (green), b) postnatal care check-up within 48 hours (red), and c) skilled birth attendance (blue), Uganda DHS data, 2000 – 2011, ordered by administrative II unit.

Appendix C Chapter 6 Appendix

C.1 Chapter 6 Supplementary Figures


Figure C.7 Marginal effect density plots at 5km (top), 50km (middle), and 100km (bottom)



Figure C.8. Model variance, model range, and model random effects at the 5km (top), 10km (middle) and 100km (bottom) scales.

Bibliography

- Achia, T.N., 2014. Spatial modelling and mapping of female genital mutilation in Kenya. BMC Public Health 14, 276. https://doi.org/10.1186/1471-2458-14-276
- Adegoke, A.A., Hofman, J.J., Kongnyuy, E.J., van den Broek, N., 2011. Monitoring and evaluation of skilled birth attendance: A proposed new framework. Midwifery, The Maternity Workforce 27, 350–359. https://doi.org/10.1016/j.midw.2011.03.006
- Ahmed, S., Li, Q., Scrafford, C., Pullum, T., 2014. An Assessment of DHS Maternal Mortality Data and Estimates (DHS Methodological Report No. MR13). ICF International, Rockville, Maryland, USA.
- Akunga, D., Menya, D., Kabue, M., 2014. Determinants of Postnatal Care Use in Kenya. Afr. Popul. Stud. 28, 1447–1459. https://doi.org/10.11564/28-3-638
- Alegana, V.A., Atkinson, P.M., Lourenço, C., Ruktanonchai, N.W., Bosco, C., Erbach-Schoenberg,
 E. zu, Didier, B., Pindolia, D., Menach, A.L., Katokele, S., Uusiku, P., Tatem, A.J., 2016.
 Advances in mapping malaria for elimination: fine resolution modelling of Plasmodium
 falciparum incidence. Sci. Rep. 6, 1–14. https://doi.org/10.1038/srep29628
- Alegana, V.A., Atkinson, P.M., Pezzulo, C., Sorichetta, A., Weiss, D., Bird, T., Erbach-Schoenberg, E., Tatem, A.J., 2015. Fine resolution mapping of population age-structures for health and development applications. J. R. Soc. Interface 12, 20150073. https://doi.org/10.1098/rsif.2015.0073
- Alegana, V.A., Wright, J.A., Pentrina, U., Noor, A.M., Snow, R.W., Atkinson, P.M., 2012. Spatial modelling of healthcare utilisation for treatment of fever in Namibia. Int. J. Health Geogr. 11, 6. https://doi.org/10.1186/1476-072X-11-6
- Alkema, L., Chou, D., Hogan, D., Zhang, S., Moller, A.-B., Gemmill, A., Fat, D.M., Boerma, T., Temmerman, M., Mathers, C., Say, L., 2016a. Global, regional, and national levels and trends in maternal mortality between 1990 and 2015, with scenario-based projections to 2030: a systematic analysis by the UN Maternal Mortality Estimation Inter-Agency Group. The Lancet 387, 462–474. https://doi.org/10.1016/S0140-6736(15)00838-7
- Alkema, L., Chou, D., Hogan, D., Zhang, S., Moller, A.-B., Gemmill, A., Fat, D.M., Boerma, T., Temmerman, M., Mathers, C., Say, L., 2016b. Global, regional, and national levels and trends in maternal mortality between 1990 and 2015, with scenario-based projections to 2030: a systematic analysis by the UN Maternal Mortality Estimation Inter-Agency Group. The Lancet 387, 462–474. https://doi.org/10.1016/S0140-6736(15)00838-7
- Althabe, F., Sosa, C., Belizán, J.M., Gibbons, L., Jacquerioz, F., Bergel, E., 2006. Cesarean Section Rates and Maternal and Neonatal Mortality in Low-, Medium-, and High-Income Countries: An Ecological Study. Birth 33, 270–277. https://doi.org/10.1111/j.1523-536X.2006.00118.x
- Armstrong, C.E., Martínez-Álvarez, M., Singh, N.S., John, T., Afnan-Holmes, H., Grundy, C., Ruktanochai, C.W., Borghi, J., Magoma, M., Msemo, G., Matthews, Z., Mtei, G., Lawn, J.E., 2016. Subnational variation for care at birth in Tanzania: is this explained by place, people, money or drugs? BMC Public Health 16 Suppl 2, 795. https://doi.org/10.1186/s12889-016-3404-3
- Assaf, S., Pullum, T., 2016. Levels and Trends in Maternal and Child Health Disparities by Wealth and Region in Eleven Countries with DHS Surveys (No. DHS Comparative Report No 42). ICF International, Rockville, Maryland, USA.

ASTER Global Digital Elevation Model Version 2, 2011.

- Auchincloss, A.H., Gebreab, S.Y., Mair, C., Roux, A.V.D., 2012. A Review of Spatial Methods in Epidemiology, 2000–2010. Annu. Rev. Public Health 33, 107–122. https://doi.org/10.1146/annurev-publhealth-031811-124655
- Ayiko, R., Antai, D., Kulane, A., 2009. Trends and determinants of under-five mortality in Uganda. East Afr. J. Public Health 6, 136–140.
- Bailey, P., Paxton, A., Lobis, S., Fry, D., 2006. The availability of life-saving obstetric services in developing countries: An in-depth look at the signal functions for emergency obstetric care. Int. J. Gynecol. Obstet. 93, 285–291. https://doi.org/10.1016/j.ijgo.2006.01.028
- Balk, D.L., Deichmann, U., Yetman, G., Pozzi, F., Hay, S. I., Nelson, A., 2006. Determining Global Population Distribution: Methods, Applications and Data, in: Hay, Simon I., Graham, A., Rogers, D.J. (Eds.), Advances in Parasitology, Global Mapping of Infectious Diseases: Methods, Examples and Emerging Applications. Academic Press, pp. 119–156. https://doi.org/10.1016/S0065-308X(05)62004-0
- Banerjee, S., 2016. Spatial Data Analysis. Annu. Rev. Public Health 37, 47–60. https://doi.org/10.1146/annurev-publhealth-032315-021711
- Barr, D.J., Levy, R., Scheepers, C., Tily, H.J., 2013. Random effects structure for confirmatory hypothesis testing: Keep it maximal. J. Mem. Lang. 68, 255–278. https://doi.org/10.1016/j.jml.2012.11.001
- Barros, A.J., Ronsmans, C., Axelson, H., Loaiza, E., Bertoldi, A.D., França, G.V., Bryce, J., Boerma, J.T., Victora, C.G., 2012. Equity in maternal, newborn, and child health interventions in Countdown to 2015: a retrospective review of survey data from 54 countries. The Lancet 379, 1225–1233. https://doi.org/10.1016/S0140-6736(12)60113-5
- Barros, A.J.D., Victora, C.G., 2013. Measuring Coverage in MNCH: Determining and Interpreting Inequalities in Coverage of Maternal, Newborn, and Child Health Interventions. PLOS Med. 10, e1001390. https://doi.org/10.1371/journal.pmed.1001390
- Bates, D., Mächler, M., Bolker, B., Walker, S., 2014. Fitting Linear Mixed-Effects Models using lme4. ArXiv14065823 Stat.
- Battese, G.E., Harter, R.M., Fuller, W.A., 1988. An Error-Components Model for Prediction of County Crop Areas Using Survey and Satellite Data. J. Am. Stat. Assoc. 83, 28–36. https://doi.org/10.1080/01621459.1988.10478561
- Bearak, J., Popinchalk, A., Alkema, L., Sedgh, G., 2018. Global, regional, and subregional trends in unintended pregnancy and its outcomes from 1990 to 2014: estimates from a Bayesian hierarchical model. Lancet Glob. Health 6, e380–e389. https://doi.org/10.1016/S2214-109X(18)30029-9
- Betrán, A.P., Merialdi, M., Lauer, J.A., Bing-Shun, W., Thomas, J., Look, P.V., Wagner, M., 2007. Rates of caesarean section: analysis of global, regional and national estimates. Paediatr. Perinat. Epidemiol. 21, 98–113. https://doi.org/10.1111/j.1365-3016.2007.00786.x
- Bhattacharjee, N.V., Schaeffer, L.E., Marczak, L.B., Ross, J.M., Swartz, S.J., Albright, J., Gardner, W.M., Shields, C., Sligar, A., Schipp, M.F., Pickering, B.V., Henry, N.J., Johnson, K.B., Louie, C., Cork, M.A., Steuben, K.M., Lazzar-Atwood, A., Lu, D., Kinyoki, D.K., Osgood-Zimmerman, A., Earl, L., Mosser, J.F., Deshpande, A., Burstein, R., Woyczynski, L.P., Wilson, K.F., VanderHeide, J.D., Wiens, K.E., Reiner, R.C., Piwoz, E.G., Rawat, R., Sartorius, B., Weaver, N.D., Nixon, M.R., Smith, D.L., Kassebaum, N.J., Gakidou, E., Lim, S.S., Mokdad, A.H., Murray, C.J.L., Dwyer-Lindgren, L., Hay, S.I., 2019. Mapping exclusive breastfeeding in Africa between 2000 and 2017. Nat. Med. 1. https://doi.org/10.1038/s41591-019-0525-0

- Bhutta, Z.A., Chopra, M., Axelson, H., Berman, P., Boerma, T., Bryce, J., Bustreo, F., Cavagnero, E., Cometto, G., Daelmans, B., de Francisco, A., Fogstad, H., Gupta, N., Laski, L., Lawn, J., Maliqi, B., Mason, E., Pitt, C., Requejo, J., Starrs, A., Victora, C.G., Wardlaw, T., 2010. Countdown to 2015 decade report (2000–10): taking stock of maternal, newborn, and child survival. The Lancet 375, 2032–2044. https://doi.org/10.1016/S0140-6736(10)60678-2
- Bhutta ZA, Reddy K, 2012. Achieving equity in global health: So near and yet so far. JAMA 307, 2035–2036. https://doi.org/10.1001/jama.2012.4659
- Black, R.E., Cousens, S., Johnson, H.L., Lawn, J.E., Rudan, I., Bassani, D.G., Jha, P., Campbell, H., Walker, C.F., Cibulskis, R., Eisele, T., Liu, L., Mathers, C., 2010. Global, regional, and national causes of child mortality in 2008: a systematic analysis. The Lancet 375, 1969– 1987. https://doi.org/10.1016/S0140-6736(10)60549-1
- Blangiardo, M., Cameletti, M., 2015. Spatial and Spatio-temporal Bayesian Models with R INLA. John Wiley & Sons.
- Bloom, S.S., Lippeveld, T., Wypij, D., 1999. Does antenatal care make a difference to safe delivery? A study in urban Uttar Pradesh, India. Health Policy Plan. 14, 38–48. https://doi.org/10.1093/heapol/14.1.38
- Boerma, J., Bryce, J., Kinfu, Y., Axelson, H., Victora, C., 2008. Mind the gap: equity and trends in coverage of maternal, newborn, and child health services in 54 Countdown countries. Lancet Lond. Engl. 371, 1259–1267. https://doi.org/10.1016/S0140-6736(08)60560-7
- Boerma, T., Victora, C., Abouzahr, C., 2018. Monitoring country progress and achievements by making global predictions: is the tail wagging the dog? Lancet Lond. Engl. 392, 607–609. https://doi.org/10.1016/S0140-6736(18)30586-5
- Boldosser-Boesch, A., Brun, M., Carvajal, L., Chou, D., de Bernis, L., Fogg, K., Hill, K., Jolivet, R., McCallon, B., Moran, A., Say, L., Smith, J., Stanton, M.E., ten Hoope-Bender, P., Wegner, M.N., 2017. Setting maternal mortality targets for the SDGs. Lancet, The Lancet 389, 696–697. http://dx.doi.org/10.1016/S0140-6736(17)30337-9
- Bosco, C., Alegana, V., Bird, T., Pezzulo, C., Bengtsson, L., Sorichetta, A., Steele, J., Hornby, G., Ruktanonchai, C., Ruktanonchai, N., Wetter, E., Tatem, A.J., 2017. Exploring the highresolution mapping of gender-disaggregated development indicators. J. R. Soc. Interface 14, 20160825. https://doi.org/10.1098/rsif.2016.0825
- Boulos, M.N.K., 2004. Towards evidence-based, GIS-driven national spatial health information infrastructure and surveillance services in the United Kingdom. Int. J. Health Geogr. 3, 1. https://doi.org/10.1186/1476-072X-3-1
- Brooker, S., Hay, S.I., Bundy, D.A.P., 2002. Tools from ecology: useful for evaluating infection risk models? Trends Parasitol. 18, 70–74. https://doi.org/10.1016/S1471-4922(01)02223-1
- Bureau for Industrial Cooperation, 2011. Thematic Area 4: Harmonization of Road Safety Laws and Regulations (No. Working Paper No. 4). East African Community.
- Burgert, C.R., Colston, J., Roy, T., Zachary, B., 2013. Geographic displacement procedure and georeferenced data release policy for the Demographic and Health Surveys, DHS Spatial Analysis Reports No. 7. ICF International, Calverton, Maryland, USA.
- Burgert, C.R., Prosnitz, D., 2014. Linking DHS household and SPA facility surveys: Data considerations and geospatial methods.
- Byrne, A., Hodge, A., Jimenez-Soto, E., Morgan, A., 2013. Looking Beyond Supply A Systematic Literature Review of Demand-Side Barriers to Health Service Utilization in the Mountains of Nepal. Asia. Pac. J. Public Health 25, 438–451. https://doi.org/10.1177/1010539513496139

- Central Bureau of Statistics CBS/Kenya, Ministry of Health MOH/Kenya, ORC Macro, 2004. Kenya Demographic and Health Survey 2003. CBS, MOH, and ORC Macro, Calverton, Maryland, USA.
- Chandra, H., Sud, U.C., 2012. Small Area Estimation for Zero-Inflated Data. Commun. Stat. -Simul. Comput. 41, 632–643. https://doi.org/10.1080/03610918.2011.598991
- Clements, A.C.A., Lwambo, N.J.S., Blair, L., Nyandindi, U., Kaatano, G., Kinung'hi, S., Webster, J.P., Fenwick, A., Brooker, S., 2006a. Bayesian spatial analysis and disease mapping: tools to enhance planning and implementation of a schistosomiasis control programme in Tanzania. Trop. Med. Int. Health 11, 490–503. https://doi.org/10.1111/j.1365-3156.2006.01594.x
- Clements, A.C.A., Lwambo, N.J.S., Blair, L., Nyandindi, U., Kaatano, G., Kinung'hi, S., Webster, J.P., Fenwick, A., Brooker, S., 2006b. Bayesian spatial analysis and disease mapping: tools to enhance planning and implementation of a schistosomiasis control programme in Tanzania. Trop. Med. Int. Health 11, 490–503. https://doi.org/10.1111/j.1365-3156.2006.01594.x
- Countdown Core Group, 2014. Fulfilling the Health Agenda for Women and Children: the 2014 Report. The United Nations Children's Fund, New York, NY, USA.
- Dahiru, T., Oche, O.M., 2015. Determinants of antenatal care, institutional delivery and postnatal care services utilization in Nigeria. Pan Afr. Med. J. 21. https://doi.org/10.4314/pamj.v21i1.
- De Allegri, M., Ridde, V., Louis, V.R., Sarker, M., Tiendrebéogo, J., Yé, M., Müller, O., Jahn, A., 2011. Determinants of utilisation of maternal care services after the reduction of user fees: A case study from rural Burkina Faso. Health Policy 99, 210–218. https://doi.org/10.1016/j.healthpol.2010.10.010
- de Montjoye, Y.-A., Hidalgo, C.A., Verleysen, M., Blondel, V.D., 2013. Unique in the Crowd: The privacy bounds of human mobility. Sci. Rep. 3, 1376. https://doi.org/10.1038/srep01376
- Deichmann, U., Balk, D., Yetman, G., 2001. Transforming Population Data for Interdisciplinary Usages: From Census to Grid. NASA Socioeconomic Data and Applications Center (SEDAC), Columbia University, Palisades, NY, USA.
- Dobson, J.E., Bright, E.A., Coleman, P.R., Durfee, R.C., Worley, B.A., 2000. LandScan: A Global Population Database for Estimating Populations at Risk. Photogramm. Eng. Remote Sens. 66, 849–857.
- Ebener, S., Guerra-Arias, M., Campbell, J., Tatem, A.J., Moran, A.C., Amoako Johnson, F., Fogstad, H., Stenberg, K., Neal, S., Bailey, P., Porter, R., Matthews, Z., 2015. The geography of maternal and newborn health: the state of the art. Int. J. Health Geogr. 14, 19. https://doi.org/10.1186/s12942-015-0012-x
- Ensor, T., Cooper, S., 2004. Overcoming barriers to health service access: influencing the demand side. Health Policy Plan. 19, 69–79. https://doi.org/10.1093/heapol/czh009
- Environmental Systems Research Institute, 2014. ArcGIS Desktop: Release 10.2.2. Redlands, CA, USA.
- Fawcett, T., 2006. An introduction to ROC analysis. Pattern Recognit. Lett., ROC Analysis in Pattern Recognition 27, 861–874. https://doi.org/10.1016/j.patrec.2005.10.010
- Fay, R.E., Herriot, R.A., 1979. Estimates of Income for Small Places: An Application of James-Stein Procedures to Census Data. J. Am. Stat. Assoc. 74, 269–277. https://doi.org/10.1080/01621459.1979.10482505

- Finlayson, K., Downe, S., 2013. Why Do Women Not Use Antenatal Services in Low- and Middle-Income Countries? A Meta-Synthesis of Qualitative Studies. PLoS Med. 10. https://doi.org/10.1371/journal.pmed.1001373
- Fort, A., Kothari, M.T., Abderrahim, N., 2006. Postpartum care: levels and determinants in developing countries (No. CR15). Macro International Inc., Calverton, Maryland, USA.
- Gabrysch, S., Campbell, O.M., 2009a. Still too far to walk: Literature review of the determinants of delivery service use. BMC Pregnancy Childbirth 9, 34. https://doi.org/10.1186/1471-2393-9-34
- Gabrysch, S., Campbell, O.M., 2009b. Still too far to walk: Literature review of the determinants of delivery service use. BMC Pregnancy Childbirth 9, 34. https://doi.org/10.1186/1471-2393-9-34
- Gabrysch, S., Cousens, S., Cox, J., Campbell, O.M.R., 2011. The Influence of Distance and Level of Care on Delivery Place in Rural Zambia: A Study of Linked National Data in a Geographic Information System. PLoS Med 8, e1000394. https://doi.org/10.1371/journal.pmed.1000394
- Gething, P., Tatem, A., Bird, T., Burgert, C., 2015. Creating Spatial Interpolation Surfaces with DHS Data (No. SAR 11), Creating Spatial Interpolation Surfaces with DHS Data. ICF International, Rockville, Maryland, USA.
- Gething, P.W., Johnson, F.A., Frempong-Ainguah, F., Nyarko, P., Baschieri, A., Aboagye, P., Falkingham, J., Matthews, Z., Atkinson, P.M., 2012. Geographical access to care at birth in Ghana: a barrier to safe motherhood. BMC Public Health 12, 991. https://doi.org/10.1186/1471-2458-12-991
- Gething, P.W., Noor, A.M., Gikandi, P.W., Ogara, E.A.A., Hay, S.I., Nixon, M.S., Snow, R.W., Atkinson, P.M., 2006. Improving Imperfect Data from Health Management Information Systems in Africa Using Space–Time Geostatistics. PLOS Med. 3, e271. https://doi.org/10.1371/journal.pmed.0030271
- Gitimu, A., Herr, C., Oruko, H., Karijo, E., Gichuki, R., Ofware, P., Lakati, A., Nyagero, J., 2015. Determinants of use of skilled birth attendant at delivery in Makueni, Kenya: a cross sectional study. BMC Pregnancy Childbirth 15, 9. https://doi.org/10.1186/s12884-015-0442-2
- Global Land Cover Map (GlobCover), 2009. . European Space Agency (ESA).
- Goldenberg, R.L., McClure, E.M., 2012. Disparities in interventions for child and maternal mortality. The Lancet 379, 1178–1180. https://doi.org/10.1016/S0140-6736(12)60474-7
- Golding, N., Burstein, R., Longbottom, J., Browne, A.J., Fullman, N., Osgood-Zimmerman, A., Earl, L., Bhatt, S., Cameron, E., Casey, D.C., Dwyer-Lindgren, L., Farag, T.H., Flaxman, A.D., Fraser, M.S., Gething, P.W., Gibson, H.S., Graetz, N., Krause, L.K., Kulikoff, X.R., Lim, S.S., Mappin, B., Morozoff, C., Reiner, R.C., Sligar, A., Smith, D.L., Wang, H., Weiss, D.J., Murray, C.J.L., Moyes, C.L., Hay, S.I., 2017. Mapping under-5 and neonatal mortality in Africa, 2000–15: a baseline analysis for the Sustainable Development Goals. The Lancet 390, 2171–2182. https://doi.org/10.1016/S0140-6736(17)31758-0
- González, M.C., Hidalgo, C.A., Barabási, A.-L., 2008. Understanding individual human mobility patterns. Nature 453, 779–782. https://doi.org/10.1038/nature06958
- Goovaerts, P., 2006. Geostatistical analysis of disease data: visualization and propagation of spatial uncertainty in cancer mortality risk using Poisson kriging and p-field simulation. Int. J. Health Geogr. 5, 7. https://doi.org/10.1186/1476-072X-5-7

- Graetz, N., Friedman, J., Osgood-Zimmerman, A., Burstein, R., Biehl, M.H., Shields, C., Mosser, J.F., Casey, D.C., Deshpande, A., Earl, L., Reiner, R.C., Ray, S.E., Fullman, N., Levine, A.J., Stubbs, R.W., Mayala, B.K., Longbottom, J., Browne, A.J., Bhatt, S., Weiss, D.J., Gething, P.W., Mokdad, A.H., Lim, S.S., Murray, C.J.L., Gakidou, E., Hay, S.I., 2018. Mapping local variation in educational attainment across Africa. Nature 555, 48–53. https://doi.org/10.1038/nature25761
- Gwatkin, D.R., 2005. How much would poor people gain from faster progress towards the Millennium Development Goals for health? The Lancet 365, 813–817. https://doi.org/10.1016/S0140-6736(05)17992-6
- Gwatkin, D.R., Rutstein, S., Johnson, K., Suliman, E., Wagstaff, A., Amouzou, A., 2007. Socioeconomic differences in health, nutrition, and population within developing countries : an overview (No. 48361). The World Bank.
- Hanson, C., Gabrysch, Sabine, Mbaruku, G., Cox, Jonathan, Mkumbo, E., Manzi, F., Schellenberg, J., Ronsmans, Carine, 2017. Access to maternal health services: geographical inequalities, United Republic of Tanzania. Bull. World Health Organ. 95, 810–820. https://doi.org/10.2471/BLT.17.194126
- Hay, S.I., Noor, A.M., Nelson, A., Tatem, A.J., 2005. The accuracy of human population maps for public health application. Trop. Med. Int. Health 10, 1073–1086. https://doi.org/10.1111/j.1365-3156.2005.01487.x
- Hazarika, I., 2011. Factors that Determine the Use of Skilled Care During Delivery in India: Implications for Achievement of MDG-5 Targets. Matern. Child Health J. 15, 1381–1388. https://doi.org/10.1007/s10995-010-0687-3
- Health Metrics Network, 2008. Framework and standards for country health information systems. World Health Organization, Geneva.
- Higgs, G., 2004. A Literature Review of the Use of GIS-Based Measures of Access to Health Care Services. Health Serv. Outcomes Res. Methodol. 5, 119–139.
- Hosseinpoor, A.R., Bergen, N., Magar, V., Hosseinpoor, A.R., Bergen, N., Magar, V., 2015. Monitoring inequality: an emerging priority for health post-2015. Bull. World Health Organ. 93, 591-591A. https://doi.org/10.2471/BLT.15.162081
- Howes, R.E., Reiner Jr., R.C., Battle, K.E., Longbottom, J., Mappin, B., Ordanovich, D., Tatem, A.J., Drakeley, C., Gething, P.W., Zimmerman, P.A., Smith, D.L., Hay, S.I., 2015. Plasmodium vivax Transmission in Africa. PLoS Negl. Trop. Dis. 9. https://doi.org/10.1371/journal.pntd.0004222
- Hug, L., Alexander, M., You, D., Alkema, L., 2019. National, regional, and global levels and trends in neonatal mortality between 1990 and 2017, with scenario-based projections to 2030: a systematic analysis. Lancet Glob. Health 7, e710–e720. https://doi.org/10.1016/S2214-109X(19)30163-9
- Hulton, L., Matthews, Z., Bandali, S., Izge, A., Daroda, R., Stones, W., 2016. Accountability for quality of care: Monitoring all aspects of quality across a framework adapted for action. Int. J. Gynecol. Obstet. 132, 110–116. https://doi.org/10.1016/j.ijgo.2015.11.005
- Hulton, L.A., Matthews, Z., Stones, R.W., 2007. Applying a framework for assessing the quality of maternal health services in urban India. Soc. Sci. Med. 64, 2083–2095. https://doi.org/10.1016/j.socscimed.2007.01.019
- Hulton, L.A., Matthews, Z., Stones, R.W., 2000. A framework for the evaluation of quality of care in maternity services. University of Southampton, Southampton, United Kingdom.

- ICF International, 2015. STATcompiler [WWW Document]. DHS Program STATcompiler. URL http://www.statcompiler.com/en/ (accessed 9.7.17).
- ICSU, ISSC, 2015. Review of the Sustainable Development Goals: The Science Perspective. International Council for Science (ICSU), Paris.
- Institut de Statistiques et d'Études Économiques du Burundi (ISTEEBU), Ministère de la Santé Publique et de la Lutte contre le Sida Burundi (MSPLS), and ICF International Inc, 2012. Burundi Demographic and Health Survey 2010. BUIR61.DTA, BUGE61FL.DBF. ICF International [Distributor], Bujumbura, Burundi.
- Institut National de la Statistique du Rwanda INSR, ORC Macro, 2006. Rwanda Demographic and Health Survey 2005. INSR and ORC Macro, Calverton, Maryland, USA.
- Irani, M., Deering, S., 2015. Challenges affecting access to cesarean delivery and strategies to overcome them in low-income countries. Int. J. Gynecol. Obstet. 131, 30–34. https://doi.org/10.1016/j.ijgo.2015.04.036
- Islam MR, Odland JO, 2011. Determinants of Antenatal and Postnatal Care Visits Among Indigenous People in Bangladesh: A Study of the Mru Community [WWW Document]. Rural Remote Health. URL https://pubmed.ncbi.nlm.nih.gov/21714582/ (accessed 6.16.20).
- Jacobs, B., Ir, P., Bigdeli, M., Annear, P.L., Van Damme, W., 2012. Addressing access barriers to health services: an analytical framework for selecting appropriate interventions in lowincome Asian countries. Health Policy Plan. 27, 288–300. https://doi.org/10.1093/heapol/czr038
- Johnson, F.A., Padmadas, S.S., Chandra, H., Matthews, Z., Madise, N.J., 2012. Estimating unmet need for contraception by district within Ghana: An application of small-area estimation techniques. Popul. Stud. 66, 105–122. https://doi.org/10.1080/00324728.2012.678585
- Jones-Webb, R., King-Schultz, L., 2008. Multi-method Approach to Evaluate Inequities in Prenatal Care Access in Haiti. J. Health Care Poor Underserved 19, 248–257. https://doi.org/10.1353/hpu.2008.0015
- Karuri, J., Waiganjo, P., Orwa, D., Manya, A., 2014. DHIS2: The Tool to Improve Health Data Demand and Use in Kenya. J. Health Inform. Dev. Ctries. 8.
- Kazembe, L.N., Namangale, J.J., 2007. A Bayesian multinomial model to analyse spatial patterns of childhood co-morbidity in Malawi. Eur. J. Epidemiol. 22, 545–556. https://doi.org/10.1007/s10654-007-9145-y
- Kenya National Bureau of Statistics KNBS, National AIDS Control Council/Kenya, National AIDS/STD Control Programme/Kenya, Ministry of Public Health and Sanitation/Kenya, Kenya Medical Research Institute, 2010. Kenya Demographic and Health Survey 2008-09. KNBS and ICF Macro, Calverton, Maryland, USA.
- Kenya National Bureau of Statistics (KNBS) and ICF Macro, 2010. Kenya Demographic and Health Survey 2008-09. KEIR52.DTA, KEGE52FL.DBF. ICF International [Distributor], Calverton, MD, USA.
- Kenya National Bureau of Statistics, Ministry of Health/Kenya, National AIDS Control Council/Kenya, Kenya Medical Research Institute, National Council for Population and Development/Kenya, 2015. Kenya Demographic and Health Survey 2014. Rockville, MD, USA.
- Kerber, K.J., de Graft-Johnson, J.E., Bhutta, Z.A., Okong, P., Starrs, A., Lawn, J.E., 2007. Continuum of care for maternal, newborn, and child health: from slogan to service delivery. The Lancet 370, 1358–1369. https://doi.org/10.1016/S0140-6736(07)61578-5

- Khan, K.S., Wojdyla, D., Say, L., Gülmezoglu, A.M., Van Look, P.F.A., 2006. WHO analysis of causes of maternal death: a systematic review. Lancet Lond. Engl. 367, 1066–1074. https://doi.org/10.1016/S0140-6736(06)68397-9
- Kirby, R.S., Delmelle, E., Eberth, J.M., 2017. Advances in spatial epidemiology and geographic information systems. Ann. Epidemiol., GIS and Spatial Methods in Epidemiology Symposium 27, 1–9. https://doi.org/10.1016/j.annepidem.2016.12.001
- Knight, H.E., Self, A., Kennedy, S.H., 2013. Why Are Women Dying When They Reach Hospital on Time? A Systematic Review of the 'Third Delay.' PLOS ONE 8, e63846. https://doi.org/10.1371/journal.pone.0063846
- Krainski, E.T., Castro-camilo, D., 2018. Advanced Spatial Modeling with Stochastic Partial Differential Equations Using R and INLA. CRC Press, Taylor & Francis Group.
- Kruk, M.E., Prescott, M.R., Galea, S., 2008. Equity of Skilled Birth Attendant Utilization in Developing Countries: Financing and Policy Determinants. Am. J. Public Health 98, 142– 147. https://doi.org/10.2105/AJPH.2006.104265
- Kyei, N.N.A., Campbell, O.M.R., Gabrysch, S., 2012. The Influence of Distance and Level of Service Provision on Antenatal Care Use in Rural Zambia. PLOS ONE 7, e46475. https://doi.org/10.1371/journal.pone.0046475
- Lai, S., Farnham, A., Ruktanonchai, N.W., Tatem, A.J., 2019. Measuring mobility, disease connectivity and individual risk: a review of using mobile phone data and mHealth for travel medicine. J. Travel Med. 26. https://doi.org/10.1093/jtm/taz019
- Larmarange, J., Vallo, R., Yaro, S., Msellati, P., Méda, N., 2011. Methods for mapping regional trends of HIV prevalence from Demographic and Health Surveys (DHS). Cybergeo Eur. J. Geogr. https://doi.org/10.4000/cybergeo.24606
- Linard, C., Gilbert, M., Snow, R.W., Noor, A.M., Tatem, A.J., 2012. Population distribution, settlement patterns and accessibility across Africa in 2010. PloS One 7, e31743. https://doi.org/10.1371/journal.pone.0031743
- Liu, L., Johnson, H.L., Cousens, S., Perin, J., Scott, S., Lawn, J.E., Rudan, I., Campbell, H., Cibulskis, R., Li, M., Mathers, C., Black, R.E., 2012. Global, regional, and national causes of child mortality: an updated systematic analysis for 2010 with time trends since 2000. The Lancet 379, 2151–2161. https://doi.org/10.1016/S0140-6736(12)60560-1
- Liu, L., Oza, S., Hogan, D., Perin, J., Rudan, I., Lawn, J.E., Cousens, S., Mathers, C., Black, R.E., 2015. Global, regional, and national causes of child mortality in 2000–13, with projections to inform post-2015 priorities: an updated systematic analysis. The Lancet 385, 430–440. https://doi.org/10.1016/S0140-6736(14)61698-6
- Logie, D.E., Rowson, M., Ndagije, F., 2008. Innovations in Rwanda's health system: looking to the future. The Lancet 372, 256–261. https://doi.org/10.1016/S0140-6736(08)60962-9
- Magadi, M., Desta, M., 2011. A multilevel analysis of the determinants and cross-national variations of HIV seropositivity in sub-Saharan Africa: Evidence from the DHS. Health Place 17, 1067–1083. https://doi.org/10.1016/j.healthplace.2011.06.004
- Magadi, M.A., Agwanda, A.O., Obare, F.O., 2007. A comparative analysis of the use of maternal health services between teenagers and older mothers in sub-Saharan Africa: Evidence from Demographic and Health Surveys (DHS). Soc. Sci. Med. 64, 1311–1325. https://doi.org/10.1016/j.socscimed.2006.11.004
- Magalhães, R.J.S., Clements, A.C.A., 2011. Mapping the Risk of Anaemia in Preschool-Age Children: The Contribution of Malnutrition, Malaria, and Helminth Infections in West Africa. PLOS Med. 8, e1000438. https://doi.org/10.1371/journal.pmed.1000438

- Makanga, P.T., Schuurman, N., von Dadelszen, P., Firoz, T., 2016. A scoping review of geographic information systems in maternal health. Int. J. Gynecol. Obstet. 134, 13–17. https://doi.org/10.1016/j.ijgo.2015.11.022
- Målqvist, M., Sohel, N., Do, T.T., Eriksson, L., Persson, L.-Å., 2010. Distance decay in delivery care utilisation associated with neonatal mortality. A case referent study in northern Vietnam. BMC Public Health 10, 762. https://doi.org/10.1186/1471-2458-10-762
- Maternal Health and Safe Motherhood Programme, 1996. Mother-Baby Package: Implementing safe motherhood in countries. World Health Organization, Geneva, Switzerland.
- McNamee, P., Ternent, L., Hussein, J., 2009. Barriers in accessing maternal healthcare: evidence from low-and middle-income countries. Expert Rev. Pharmacoecon. Outcomes Res. 9, 41–48. https://doi.org/10.1586/14737167.9.1.41
- Mennis, J., 2003. Generating Surface Models of Population Using Dasymetric Mapping. Prof. Geogr. 55, 31–42. https://doi.org/10.1111/0033-0124.10042
- Ministry of Health, C.D., Ministry of Health MoH/Zanzibar, National Bureau of Statistics -NBS/Tanzania, Office of Chief Government Statistician - OCGS/Zanzibar, ICF, 2016.
 Tanzania Demographic and Health Survey and Malaria Indicator Survey 2015-2016.
 MoHCDGEC, MoH, NBS, OCGS, and ICF, Dar es Salaam, Tanzania.
- Minujin, A., Delamonica, E., 2003. Mind the Gap! Widening Child Mortality Disparities. J. Hum. Dev. 4, 397–418. https://doi.org/10.1080/1464988032000125764
- Mishra, U.S., Ramanathan, M., 2002. Delivery-related complications and determinants of caesarean section rates in India. Health Policy Plan. 17, 90–98. https://doi.org/10.1093/heapol/17.1.90
- Mohan, D., Gupta, S., LeFevre, A., Bazant, E., Killewo, J., Baqui, A.H., 2015. Determinants of postnatal care use at health facilities in rural Tanzania: multilevel analysis of a household survey. BMC Pregnancy Childbirth 15, 282. https://doi.org/10.1186/s12884-015-0717-7
- Molla, Y.B., Nilsen, K., Singh, K., Ruktanonchai, C.W., Schmitz, M.M., Duong, J., Serbanescu, F., Moran, A.C., Matthews, Z., Tatem, A.J., 2019. Best practices in availability, management and use of geospatial data to guide reproductive, maternal, child and adolescent health programmes. BMJ Glob. Health 4, e001406. https://doi.org/10.1136/bmjgh-2019-001406
- Molla, Y.B., Rawlins, B., Makanga, P.T., Cunningham, M., Ávila, J.E.H., Ruktanonchai, C.W., Singh, K., Alford, S., Thompson, M., Dwivedi, V., Moran, A.C., Matthews, Z., 2017. Geographic information system for improving maternal and newborn health: recommendations for policy and programs. BMC Pregnancy Childbirth 17, 26. https://doi.org/10.1186/s12884-016-1199-y
- Mosser, J.F., Gagne-Maynard, W., Rao, P.C., Osgood-Zimmerman, A., Fullman, N., Graetz, N., Burstein, R., Updike, R.L., Liu, P.Y., Ray, S.E., Earl, L., Deshpande, A., Casey, D.C., Dwyer-Lindgren, L., Cromwell, E.A., Pigott, D.M., Shearer, F.M., Larson, H.J., Weiss, D.J., Bhatt, S., Gething, P.W., Murray, C.J.L., Lim, S.S., Reiner, R.C., Hay, S.I., 2019. Mapping diphtheria-pertussis-tetanus vaccine coverage in Africa, 2000–2016: a spatial and temporal modelling study. The Lancet 393, 1843–1855. https://doi.org/10.1016/S0140-6736(19)30226-0
- Mpembeni, R.N., Killewo, J.Z., Leshabari, M.T., Massawe, S.N., Jahn, A., Mushi, D., Mwakipa, H., 2007. Use pattern of maternal health services and determinants of skilled care during delivery in Southern Tanzania: implications for achievement of MDG-5 targets. BMC Pregnancy Childbirth 7, 29. https://doi.org/10.1186/1471-2393-7-29

- Mtambo, O.P.L., Masangwi, S.J., Kazembe, L.N.M., 2015. Spatial quantile regression using INLA with applications to childhood overweight in Malawi. Spat. Spatio-Temporal Epidemiol. 13, 7–14. https://doi.org/10.1016/j.sste.2015.04.001
- Mtei, G., Makawia, S., Ally, M., Kuwawenaruwa, A., Meheus, F., Borghi, J., 2012. Who pays and who benefits from health care? An assessment of equity in health care financing and benefit distribution in Tanzania. Health Policy Plan. 27 Suppl 1, i23-34. https://doi.org/10.1093/heapol/czs018
- Muga, R., Kizito, P., Mbayah, M., Gakuruh, T., 2005. Overview of the Health System in Kenya, in: Kenya Service Provision Assessment Survey 2004. National Coordinating Agency for Population and Development (NCAPD) [Kenya], Ministry of Health (MOH), Central Bureau of Statistics (CBS), ORC Macro, Nairobi, Kenya, pp. 13–26.
- Musa, G.J., Chiang, P.-H., Sylk, T., Bavley, R., Keating, W., Lakew, B., Tsou, H.-C., Hoven, C.W., 2013. Use of GIS Mapping as a Public Health Tool–-From Cholera to Cancer. Health Serv. Insights 6, HSI.S10471. https://doi.org/10.4137/HSI.S10471
- National Bureau of Statistics NBS/Tanzania, ICF Macro, 2011. Tanzania Demographic and Health Survey 2010. NBS/Tanzania and ICF Macro, Dar es Salaam, Tanzania.
- National Bureau of Statistics (NBS) Tanzania and ICF Macro, 2011. Tanzania Demographic and Health Survey 2010. TZIR63.DTA, TZGE61FL.DBF. ICF International [Distributor], Dar es Salaam, Tanzania.
- National Bureau of Statistics/Tanzania, Macro International, 2000. Tanzania Reproductive and Child Health Survey 1999. National Bureau of Statistics/Tanzania and Macro International, Calverton, Maryland, USA.
- National Institute of Statistics of Rwanda NISR, Ministry of Health MOH/Rwanda, ICF International, 2012. Rwanda Demographic and Health Survey 2010. NISR/Rwanda, MOH/Rwanda, and ICF International, Calverton, Maryland, USA.
- National Institute of Statistics of Rwanda, Ministry of Finance and Economic Planning/Rwanda, Ministry of Health/Rwanda, ICF International, 2016. Rwanda Demographic and Health Survey 2014-15. National Institute of Statistics of Rwanda, Ministry of Finance and Economic Planning/Rwanda, Ministry of Health/Rwanda, and ICF International, Kigali, Rwanda.
- National Institute of Statistics of Rwanda (NISR), Ministry of Health (MOH) Rwanda, and ICF International Inc, 2012. Rwanda Demographic and Health Survey 2010. RWIR61.DTA, RWGE61FL.DBF. ICF International [Distributor], Calverton, MD, USA.
- Neal, S., Ruktanonchai, C., Chandra-Mouli, V., Matthews, Z., Tatem, A.J., 2016. Mapping adolescent first births within three east African countries using data from Demographic and Health Surveys: exploring geospatial methods to inform policy. Reprod. Health 13, 98. https://doi.org/10.1186/s12978-016-0205-1
- Neal, S., Ruktanonchai, C.W., Chandra-Mouli, V., Harvey, C., Matthews, Z., Raina, N., Tatem, A., 2019. Using geospatial modelling to estimate the prevalence of adolescent first births in Nepal. BMJ Glob. Health 4, e000763. https://doi.org/10.1136/bmjgh-2018-000763
- Neal, S.E., Chandra-Mouli, V., Chou, D., 2015. Adolescent first births in East Africa: disaggregating characteristics, trends and determinants. Reprod. Health 12, 13. https://doi.org/10.1186/1742-4755-12-13
- Nelson, A., 2008. Estimated travel time to the nearest city of 50,000 or more people in year 2000.
- Neuman, M., Alcock, G., Azad, K., Kuddus, A., Osrin, D., More, N.S., Nair, N., Tripathy, P., Sikorski, C., Saville, N., Sen, A., Colbourn, T., Houweling, T.A.J., Seward, N.,

Manandhar, D.S., Shrestha, B.P., Costello, A., Prost, A., 2014. Prevalence and determinants of caesarean section in private and public health facilities in underserved South Asian communities: cross-sectional analysis of data from Bangladesh, India and Nepal. BMJ Open 4, e005982. https://doi.org/10.1136/bmjopen-2014-005982

- Nguhiu, P.K., Barasa, E.W., Chuma, J., 2017. Determining the effective coverage of maternal and child health services in Kenya, using demographic and health survey data sets: tracking progress towards universal health coverage. Trop. Med. Int. Health 22, 442–453. https://doi.org/10.1111/tmi.12841
- Niragire, F., Achia, T.N.O., Lyambabaje, A., Ntaganira, J., 2015. Bayesian Mapping of HIV Infection among Women of Reproductive Age in Rwanda. PLOS ONE 10, e0119944. https://doi.org/10.1371/journal.pone.0119944
- Noor, A.M., Clements, A.C., Gething, P.W., Moloney, G., Borle, M., Shewchuk, T., Hay, S.I., Snow, R.W., 2008. Spatial prediction of Plasmodium falciparum prevalence in Somalia. Malar. J. 7, 159. https://doi.org/10.1186/1475-2875-7-159
- Nyamtema, A.S., Urassa, D.P., van Roosmalen, J., 2011. Maternal health interventions in resource limited countries: a systematic review of packages, impacts and factors for change. BMC Pregnancy Childbirth 11, 30. https://doi.org/10.1186/1471-2393-11-30
- Nykiforuk, C.I.J., Flaman, L.M., 2009. Geographic Information Systems (GIS) for Health Promotion and Public Health: A Review. Health Promot. Pract. https://doi.org/10.1177/1524839909334624
- Osgood-Zimmerman, A., Millear, A.I., Stubbs, R.W., Shields, C., Pickering, B.V., Earl, L., Graetz, N., Kinyoki, D.K., Ray, S.E., Bhatt, S., Browne, A.J., Burstein, R., Cameron, E., Casey, D.C., Deshpande, A., Fullman, N., Gething, P.W., Gibson, H.S., Henry, N.J., Herrero, M., Krause, L.K., Letourneau, I.D., Levine, A.J., Liu, P.Y., Longbottom, J., Mayala, B.K., Mosser, J.F., Noor, A.M., Pigott, D.M., Piwoz, E.G., Rao, P., Rawat, R., Reiner, R.C., Smith, D.L., Weiss, D.J., Wiens, K.E., Mokdad, A.H., Lim, S.S., Murray, C.J.L., Kassebaum, N.J., Hay, S.I., 2018. Mapping child growth failure in Africa between 2000 and 2015. Nature 555, 41–47. https://doi.org/10.1038/nature25760
- Ouma, P.O., Maina, J., Thuranira, P.N., Macharia, P.M., Alegana, V.A., English, M., Okiro, E.A., Snow, R.W., 2018. Access to emergency hospital care provided by the public sector in sub-Saharan Africa in 2015: a geocoded inventory and spatial analysis. Lancet Glob. Health 6, e342–e350. https://doi.org/10.1016/S2214-109X(17)30488-6
- Paré, G., Kitsiou, S., 2017. Chapter 9 Methods for Literature Reviews, Handbook of eHealth Evaluation: An Evidence-based Approach [Internet]. University of Victoria.
- Parkhurst, J.O., Penn-Kekana, L., Blaauw, D., Balabanova, D., Danishevski, K., Rahman, S.A., Onama, V., Ssengooba, F., 2005. Health systems factors influencing maternal health services: a four-country comparison. Health Policy 73, 127–138. https://doi.org/10.1016/j.healthpol.2004.11.001
- Patil, A.P., Gething, P.W., Piel, F.B., Hay, S.I., 2011. Bayesian geostatistics in health cartography: the perspective of malaria. Trends Parasitol. 27, 246–253. https://doi.org/10.1016/j.pt.2011.01.003
- Paxton, A., Bailey, P., Lobis, S., Fry, D., 2006. Global patterns in availability of emergency obstetric care. Int. J. Gynecol. Obstet. 93, 300–307. https://doi.org/10.1016/j.ijgo.2006.01.030
- Paxton, A., Maine, D., Freedman, L., Fry, D., Lobis, S., 2005. The evidence for emergency obstetric care. Int. J. Gynecol. Obstet. 88, 181–193. https://doi.org/10.1016/j.ijgo.2004.11.026

- Perez-Heydrich, C., Warren, J.L., Burgert, C.R., Emch, M.E., 2015. Influence of Demographic and Health Survey Point Displacements on Raster-Based Analyses. Spat. Demogr. https://doi.org/10.1007/s40980-015-0013-1
- Pezzulo, C., Bird, T., Utazi, E.C., Sorichetta, A., Tatem, A.J., Yourkavitch, J., Burgert-Brucker, C.R., 2016. Geospatial modeling of child mortality across 27 countries in Sub-Saharan Africa (No. DHS Spatial Analysis Report 13). ICF International, Rockville, MD, USA.
- Pezzulo, C., Hornby, G.M., Sorichetta, A., Gaughan, A.E., Linard, C., Bird, T.J., Kerr, D., Lloyd, C.T., Tatem, A.J., 2017. Sub-national mapping of population pyramids and dependency ratios in Africa and Asia. Sci. Data 4, 170089. https://doi.org/10.1038/sdata.2017.89
- Pilkington, H., Blondel, B., Drewniak, N., Zeitlin, J., 2012. Choice in maternity care: associations with unit supply, geographic accessibility and user characteristics. Int. J. Health Geogr. 11, 35. https://doi.org/10.1186/1476-072X-11-35
- PMNCH, 2011. A Global Review of the Key Interventions to Reproductive, Maternal, Newborn and Child Health (RMNHC). The Partnership for Maternal, Newborn & Child Health, Geneva, Switzerland.
- R Development Core Team, 2017. R: A Language and Environment for Statistical Computing. R Foundation for Statistical Computing, Vienna, Austria.
- Rajaratnam, J.K., Marcus, J.R., Flaxman, A.D., Wang, H., Levin-Rector, A., Dwyer, L., Costa, M., Lopez, A.D., Murray, C.J., 2010. Neonatal, postneonatal, childhood, and under-5 mortality for 187 countries, 1970–2010: a systematic analysis of progress towards Millennium Development Goal 4. The Lancet 375, 1988–2008. https://doi.org/10.1016/S0140-6736(10)60703-9
- Reidpath, D.D., Morel, C.M., Mecaskey, J.W., Allotey, P., 2009. The Millennium Development Goals Fail Poor Children: The Case for Equity-Adjusted Measures. PLoS Med. 6. https://doi.org/10.1371/journal.pmed.1000062
- Requejo, J.H., Bhutta, Z.A., 2015. The post-2015 agenda: staying the course in maternal and child survival. Arch. Dis. Child. 100, S76–S81. https://doi.org/10.1136/archdischild-2013-305737
- Richardson Sylvia, Thomson Andrew, Best Nicky, Elliott Paul, 2004. Interpreting Posterior Relative Risk Estimates in Disease-Mapping Studies. Environ. Health Perspect. 112, 1016– 1025. https://doi.org/10.1289/ehp.6740
- Riebler, A., Sørbye, S.H., Simpson, D., Rue, H., 2016. An intuitive Bayesian spatial model for disease mapping that accounts for scaling. ArXiv160101180 Stat.
- Roberts, D.A., Ng, M., Ikilezi, G., Gasasira, A., Dwyer-Lindgren, L., Fullman, N., Nalugwa, T., Kamya, M., Gakidou, E., 2015. Benchmarking health system performance across regions in Uganda: a systematic analysis of levels and trends in key maternal and child health interventions, 1990–2011. BMC Med. 13, 285. https://doi.org/10.1186/s12916-015-0518-x
- Ronsmans, C., Graham, W.J., 2006. Maternal mortality: who, when, where, and why. The Lancet 368, 1189–1200. https://doi.org/10.1016/S0140-6736(06)69380-X
- Rue, H., Martino, S., Chopin, N., 2009. Approximate Bayesian inference for latent Gaussian models by using integrated nested Laplace approximations. J. R. Stat. Soc. Ser. B Stat. Methodol. 71, 319–392. https://doi.org/10.1111/j.1467-9868.2008.00700.x
- Ruktanonchai, C.W., Nilsen, K., Alegana, V.A., Bosco, C., Ayiko, R., Seven Kajeguka, A.C., Matthews, Z., Tatem, A.J., 2018. Temporal trends in spatial inequalities of maternal and newborn health services among four east African countries, 1999–2015. BMC Public Health 18, 1339. https://doi.org/10.1186/s12889-018-6241-8

- Ruktanonchai, C.W., Ruktanonchai, N.W., Nove, A., Lopes, S., Pezzulo, C., Bosco, C., Alegana, V.A., Burgert, C.R., Ayiko, R., Charles, A.S., Lambert, N., Msechu, E., Kathini, E., Matthews, Z., Tatem, A.J., 2016. Equality in Maternal and Newborn Health: Modelling Geographic Disparities in Utilisation of Care in Five East African Countries. PLOS ONE 11, e0162006. https://doi.org/10.1371/journal.pone.0162006
- Ruktanonchai, N.W., DeLeenheer, P., Tatem, A.J., Alegana, V.A., Caughlin, T.T., Erbach-Schoenberg, E. zu, Lourenço, C., Ruktanonchai, C.W., Smith, D.L., 2016. Identifying Malaria Transmission Foci for Elimination Using Human Mobility Data. PLOS Comput. Biol. 12, e1004846. https://doi.org/10.1371/journal.pcbi.1004846
- Ruktanonchai, N.W., Ruktanonchai, C.W., Floyd, J.R., Tatem, A.J., 2018. Using Google Location History data to quantify fine-scale human mobility. Int. J. Health Geogr. 17. https://doi.org/10.1186/s12942-018-0150-z
- Rushton, G., 2003. Public Health, GIS, and Spatial Analytic Tools. Annu. Rev. Public Health 24, 43–56. https://doi.org/10.1146/annurev.publhealth.24.012902.140843
- Saad–Haddad, G., DeJong, J., Terreri, N., Restrepo–Méndez, M.C., Perin, J., Vaz, L., Newby, H., Amouzou, A., Barros, A.J., Bryce, J., 2016. Patterns and determinants of antenatal care utilization: analysis of national survey data in seven countdown countries. J. Glob. Health 6. https://doi.org/10.7189/jogh.06.010404
- Sandall, J., Tribe, R.M., Avery, L., Mola, G., Visser, G.H., Homer, C.S., Gibbons, D., Kelly, N.M., Kennedy, H.P., Kidanto, H., Taylor, P., Temmerman, M., 2018. Short-term and long-term effects of caesarean section on the health of women and children. Lancet Lond. Engl. 392, 1349–1357. https://doi.org/10.1016/S0140-6736(18)31930-5
- SAS Institute Inc., 2013. SAS version 9.4. SAS Institute Inc., Cary, NC, USA.
- Say, L., Chou, D., Gemmill, A., Tunçalp, Ö., Moller, A.-B., Daniels, J., Gülmezoglu, A.M., Temmerman, M., Alkema, L., 2014a. Global causes of maternal death: a WHO systematic analysis. Lancet Glob. Health 2, e323–e333. https://doi.org/10.1016/S2214-109X(14)70227-X
- Say, L., Chou, D., Gemmill, A., Tunçalp, Ö., Moller, A.-B., Daniels, J., Gülmezoglu, A.M., Temmerman, M., Alkema, L., 2014b. Global causes of maternal death: a WHO systematic analysis. Lancet Glob. Health 2, e323–e333. https://doi.org/10.1016/S2214-109X(14)70227-X
- Say, L., Raine, R., 2007. A systematic review of inequalities in the use of maternal health care in developing countries: examining the scale of the problem and the importance of context. Bull. World Health Organ. 85, 812–819. https://doi.org/10.1590/S0042-96862007001000019
- Schur, N., Hürlimann, E., Garba, A., Traoré, M.S., Ndir, O., Ratard, R.C., Tchuenté, L.-A.T., Kristensen, T.K., Utzinger, J., Vounatsou, P., 2011. Geostatistical Model-Based Estimates of Schistosomiasis Prevalence among Individuals Aged ≤20 Years in West Africa. PLoS Negl. Trop. Dis. 5, e1194. https://doi.org/10.1371/journal.pntd.0001194
- Simkhada, B., Teijlingen, E.R. van, Porter, M., Simkhada, P., 2008. Factors affecting the utilization of antenatal care in developing countries: systematic review of the literature. J. Adv. Nurs. 61, 244–260. https://doi.org/10.1111/j.1365-2648.2007.04532.x
- Snijders, R., Bosker, T., 1999. Multilevel Analysis: An Introduction to Basic and Advanced Multilevel Modeling. SAGE.
- Somefun, O.D., Ibisomi, L., 2016. Determinants of postnatal care non-utilization among women in Nigeria. BMC Res. Notes 9, 21. https://doi.org/10.1186/s13104-015-1823-3

- Sorichetta, A., Bird, T.J., Ruktanonchai, N.W., Erbach-Schoenberg, E. zu, Pezzulo, C., Tejedor, N., Waldock, I.C., Sadler, J.D., Garcia, A.J., Sedda, L., Tatem, A.J., 2016. Mapping internal connectivity through human migration in malaria endemic countries. Sci. Data 3, 160066. https://doi.org/10.1038/sdata.2016.66
- Souza, J.P., Gülmezoglu, A.M., Vogel, J., Carroli, G., Lumbiganon, P., Qureshi, Z., Costa, M.J., Fawole, B., Mugerwa, Y., Nafiou, I., Neves, I., Wolomby-Molondo, J.-J., Bang, H.T., Cheang, K., Chuyun, K., Jayaratne, K., Jayathilaka, C.A., Mazhar, S.B., Mori, R., Mustafa, M.L., Pathak, L.R., Perera, D., Rathavy, T., Recidoro, Z., Roy, M., Ruyan, P., Shrestha, N., Taneepanichsku, S., Tien, N.V., Ganchimeg, T., Wehbe, M., Yadamsuren, B., Yan, W., Yunis, K., Bataglia, V., Cecatti, J.G., Hernandez-Prado, B., Nardin, J.M., Narváez, A., Ortiz-Panozo, E., Pérez-Cuevas, R., Valladares, E., Zavaleta, N., Armson, A., Crowther, C., Hogue, C., Lindmark, G., Mittal, S., Pattinson, R., Stanton, M.E., Campodonico, L., Cuesta, C., Giordano, D., Intarut, N., Laopaiboon, M., Bahl, R., Martines, J., Mathai, M., Merialdi, M., Say, L., 2013. Moving beyond essential interventions for reduction of maternal mortality (the WHO Multicountry Survey on Maternal and Newborn Health): a cross-sectional study. The Lancet 381, 1747–1755. https://doi.org/10.1016/S0140-6736(13)60686-8
- Spiegelhalter David J., Best Nicola G., Carlin Bradley P., Van Der Linde Angelika, 2002. Bayesian measures of model complexity and fit. J. R. Stat. Soc. Ser. B Stat. Methodol. 64, 583–639. https://doi.org/10.1111/1467-9868.00353
- Spiegelhalter, D.J., Best, N.G., Carlin, B.P., Van Der Linde, A., 2002. Bayesian measures of model complexity and fit. J. R. Stat. Soc. Ser. B Stat. Methodol. 64, 583–639. https://doi.org/10.1111/1467-9868.00353
- Stevens, F.R., Gaughan, A.E., Linard, C., Tatem, A.J., 2015. Disaggregating Census Data for Population Mapping Using Random Forests with Remotely-Sensed and Ancillary Data. PLOS ONE 10, e0107042. https://doi.org/10.1371/journal.pone.0107042
- Sylvester, A., Tate, M., Johnstone, D., 2013. Beyond synthesis: re-presenting heterogeneous research literature. Behav. Inf. Technol. 32, 1199–1215. https://doi.org/10.1080/0144929X.2011.624633
- Tanser, F.C., le Sueur, D., 2002. The application of geographical information systems to important public health problems in Africa. Int. J. Health Geogr. 1, 4. https://doi.org/10.1186/1476-072X-1-4
- Tatem, A., Gething, P., Bhatt, S., Weiss, D., Pezzulo, C., 2013. Pilot high resolution poverty maps.
- Tatem, A., Linard, C., 2011. Population mapping of poor countries [WWW Document]. Nature. https://doi.org/10.1038/474036d
- Tatem, A.J., 2017. WorldPop, open data for spatial demography. Sci. Data 4, 170004. https://doi.org/10.1038/sdata.2017.4
- Tatem, A.J., Adamo, S., Bharti, N., Burgert, C.R., Castro, M., Dorelien, A., Fink, G., Linard, C., John, M., Montana, L., Montgomery, M.R., Nelson, A., Noor, A.M., Pindolia, D., Yetman, G., Balk, D., 2012. Mapping populations at risk: improving spatial demographic data for infectious disease modeling and metric derivation. Popul. Health Metr. 10, 8. https://doi.org/10.1186/1478-7954-10-8
- Tatem, A.J., Campbell, J., Guerra-Arias, M., de Bernis, L., Moran, A., Matthews, Z., 2014. Mapping for maternal and newborn health: the distributions of women of childbearing age, pregnancies and births. Int. J. Health Geogr. 13, 2. https://doi.org/10.1186/1476-072X-13-2
- Tatem, A.J., Garcia, A.J., Snow, R.W., Noor, A.M., Gaughan, A.E., Gilbert, M., Linard, C., 2013. Millennium development health metrics: where do Africa's children and women of

childbearing age live? Popul. Health Metr. 11, 11. https://doi.org/10.1186/1478-7954-11-11

- Tatem, A.J., Hay, S.I., 2004. Measuring urbanization pattern and extent for malaria research: A review of remote sensing approaches. J. Urban Health Bull. N. Y. Acad. Med. 81, 363– 376. https://doi.org/10.1093/jurban/jth124
- Tatem, A.J., Noor, A.M., Hagen, C. von, Gregorio, A.D., Hay, S.I., 2007. High Resolution Population Maps for Low Income Nations: Combining Land Cover and Census in East Africa. PLOS ONE 2, e1298. https://doi.org/10.1371/journal.pone.0001298
- Tatem, A.J., Noor, A.M., Hay, S.I., 2004. Defining approaches to settlement mapping for public health management in Kenya using medium spatial resolution satellite imagery. Remote Sens. Environ. 93, 42–52. https://doi.org/10.1016/j.rse.2004.06.014
- Thaddeus, S., Maine, D., 1994. Too far to walk: Maternal mortality in context. Soc. Sci. Med. 38, 1091–1110. https://doi.org/10.1016/0277-9536(94)90226-7
- The Partnership for Maternal, Newborn and Child Health, 2006. Opportunities for Africa's newborns: Practical data, policy and programmatic support for newborn care in Africa. World Health Organization, Geneva, Switzerland.
- Tobler, W., 1993. Three Presentations on Geographical Analysis and Modeling: Non-Isotropic Geographic Modeling; Speculations on the Geometry of Geography; and Global Spatial Analysis (93-1).
- Uganda Bureau of Statistics UBOS, ICF International, 2012. Uganda Demographic and Health Survey 2011. UBOS and ICF International, Kampala, Uganda.
- Uganda Bureau of Statistics UBOS, Macro International, 2007. Uganda Demographic and Health Survey 2006. UBOS and Macro International, Calverton, Maryland, USA.
- Uganda Bureau of Statistics UBOS, ORC Macro, 2001. Uganda Demographic and Health Survey 2000-2001. UBOS and ORC Macro, Calverton, Maryland, USA.
- Uganda Bureau of Statistics (UBOS) and ICF International Inc, 2012. Uganda Demographic and Health Survey 2011. UGIR60.DTA, UGGE61FL.DBF. ICF International [Distributor], Kampala, Uganda and Calverton, MD, USA.
- UN Inter-agency Group for Child Mortality Estimation, 2015. Levels and Trends in Child Mortality: 2015 Report. UN IGME, New York, NY, USA.
- UNFPA, ICM, WHO, 2014. The state of the world's midwifery 2014: a universal pathway, a woman's right to health. United Nations Population Fund, New York, NY, USA.
- UNICEF, WHO, 2015. Countdown to 2015: A Decade of Tracking Progress for Maternal, Newborn and Child Survival. World Health Organization, Geneva, Switzerland.
- UNICEF, WHO, UNFPA, 1997. Guidelines for Monitoring the Availability and Use of Obstetric Services. United Nations Children's Fund, New York, NY, USA.
- United Nations, 2018. The Sustainable Development Goals Report 2018. United Nations, New York, NY, USA.
- United Nations, 2016. Every Woman Every Child The Global Strategy for Women's, Children's and Adolescents' Health (2016-2030). United Nations.
- United Nations, 2015a. The Millennium Development Goals Report. United Nations Department of Economic and Social Affairs, New York, NY, USA.

- United Nations, 2015b. Africa MDG Progress Reports. United Nations Economic Commission for Africa.
- United Nations, 2013. A New Global Partnership: Eradicate Poverty and Transform Economies through Sustainable Development. New York, NY, USA.
- United Nations General Assembly, 2015. Transforming Our World: The 2030 Agenda for Sustainable Development. United Nations.
- US NOAA National Geophysical Data Center, US Air Force Weather Agency, 2014. Version 4 DMSP-OLS Nighttime Lights Time Series (1992–2013; Average Visible, Stable Lights, & Cloud Free Coverages).
- Utazi, C. Edson, Sahu, S.K., Atkinson, P.M., Tejedor-Garavito, N., Lloyd, C.T., Tatem, A.J., 2018a. Geographic coverage of demographic surveillance systems for characterising the drivers of childhood mortality in sub-Saharan Africa. BMJ Glob. Health 3, e000611. https://doi.org/10.1136/bmjgh-2017-000611
- Utazi, C. E., Thorley, J., Alegana, V.A., Ferrari, M.J., Nilsen, K., Takahashi, S., Metcalf, C.J.E., Lessler, J., Tatem, A.J., 2018. A spatial regression model for the disaggregation of areal unit based data to high-resolution grids with application to vaccination coverage mapping: Stat. Methods Med. Res. https://doi.org/10.1177/0962280218797362
- Utazi, C. Edson, Thorley, J., Alegana, V.A., Ferrari, M.J., Takahashi, S., Metcalf, C.J.E., Lessler, J., Tatem, A.J., 2018b. High resolution age-structured mapping of childhood vaccination coverage in low and middle income countries. Vaccine 36, 1583–1591. https://doi.org/10.1016/j.vaccine.2018.02.020
- van den Broek, N., Graham, W., 2009. Quality of care for maternal and newborn health: the neglected agenda. BJOG Int. J. Obstet. Gynaecol. 116, 18–21. https://doi.org/10.1111/j.1471-0528.2009.02333.x
- Varshney, K.R., Wei, D., Ramamurthy, K.N., Mojsilović, A., 2015. Data Challenges in Disease Response: The 2014 Ebola Outbreak and Beyond. J Data Inf. Qual. 6, 5:1–5:3. https://doi.org/10.1145/2742550
- Victora, C.G., Fenn, B., Bryce, J., Kirkwood, B.R., 2005. Co-coverage of preventive interventions and implications for child-survival strategies: evidence from national surveys. Lancet Lond. Engl. 366, 1460–1466. https://doi.org/10.1016/S0140-6736(05)67599-X
- Victora, C.G., Requejo, J.H., Barros, A.J.D., Berman, P., Bhutta, Z., Boerma, T., Chopra, M., de Francisco, A., Daelmans, B., Hazel, E., Lawn, J., Maliqi, B., Newby, H., Bryce, J., 2016. Countdown to 2015: a decade of tracking progress for maternal, newborn, and child survival. The Lancet 387, 2049–2059. https://doi.org/10.1016/S0140-6736(15)00519-X
- Victora, C.G., Wagstaff, A., Schellenberg, J.A., Gwatkin, D., Claeson, M., Habicht, J.-P., 2003. Applying an equity lens to child health and mortality: more of the same is not enough. The Lancet 362, 233–241. https://doi.org/10.1016/S0140-6736(03)13917-7
- Vogel, J.P., Betrán, A.P., Vindevoghel, N., Souza, J.P., Torloni, M.R., Zhang, J., Tunçalp, Ö., Mori, R., Morisaki, N., Ortiz-Panozo, E., Hernandez, B., Pérez-Cuevas, R., Qureshi, Z., Gülmezoglu, A.M., Temmerman, M., 2015. Use of the Robson classification to assess caesarean section trends in 21 countries: a secondary analysis of two WHO multicountry surveys. Lancet Glob. Health 3, e260–e270. https://doi.org/10.1016/S2214-109X(15)70094-X
- Wardrop, N.A., Jochem, W.C., Bird, T.J., Chamberlain, H.R., Clarke, D., Kerr, D., Bengtsson, L., Juran, S., Seaman, V., Tatem, A.J., 2018. Spatially disaggregated population estimates in the absence of national population and housing census data. Proc. Natl. Acad. Sci. 201715305. https://doi.org/10.1073/pnas.1715305115

- Wesolowski, A., Eagle, N., Noor, A.M., Snow, R.W., Buckee, C.O., 2012. Heterogeneous Mobile Phone Ownership and Usage Patterns in Kenya. PLOS ONE 7, e35319. https://doi.org/10.1371/journal.pone.0035319
- WHO, 2016a. Standards for improving quality of maternal and newborn care in health facilities. World Health Organization, Geneva, Switzerland.
- WHO, 2016b. Global Health Observatory [WWW Document]. WHO GHO Data Repos. URL http://apps.who.int/gho/data/node.main.REPWOMEN39?lang=en (accessed 1.20.17).
- WHO, 2015a. State of inequality: Reproductive, maternal, newborn and child health. World Health Organization, Geneva, Switzerland.
- WHO, 2015b. Health in 2015: from MDGs, Millennium Development Goals to SDGs, Sustainable Development Goals. World Health Organization, Geneva, Switzerland.
- WHO, 2014. Every Newborn: an action plan to end preventable deaths. World Health Organization, Geneva, Switzerland.
- WHO, 2010a. Framework for monitoring and evaluation of reproductive health programmes in the Eastern Mediterranean Region. World Health Organization, Geneva, Switzerland.
- WHO, 2010b. Progress Report: Reproductive health strategy to accelerate progress towards the attainment of international development goals and targets (No. WHO/RHR/10.14). World Health Organization, Geneva, Switzerland.
- WHO, 2006. Reproductive health indicators: guidelines for their generation, interpretation and analysis for global monitoring. World Health Organization, Geneva, Switzerland.
- WHO, 2005. World Health Report 2005: Make every mother and child count. World Health Organization, Geneva, Switzerland.
- WHO, HRP, 2015. Strategies toward ending preventable maternal mortality (EPMM) (No. WHO/RHR/15.03). World Health Organization, Geneva, Switzerland.
- WHO, UNFPA, UNICEF, AMDD, 2009. Monitoring emergency obstetric care: a handbook. World Health Organization, Geneva, Switzerland.
- WHO, UNICEF, World Bank, UNFPA, United Nations Population Division, 2015. Trends in Maternal Mortality: 1990 to 2015. World Health Organization.
- Wise, J., 2018. Alarming global rise in caesarean births, figures show. BMJ k4319. https://doi.org/10.1136/bmj.k4319
- Wohlin, C., 2014. Guidelines for snowballing in systematic literature studies and a replication in software engineering, in: Proceedings of the 18th International Conference on Evaluation and Assessment in Software Engineering - EASE '14. Presented at the the 18th International Conference, ACM Press, London, England, United Kingdom, pp. 1–10. https://doi.org/10.1145/2601248.2601268
- World Health Organization, 2018a. Global Strategy for Women's, Children's and Adolescents' Health (2016-2030): Monitoring the Situation of Children and Women [WWW Document]. Glob. Health Obs. URL http://apps.who.int/gho/data/node.gswcah (accessed 11.1.19).
- World Health Organization, 2018b. Births by caesarean section Data by country [WWW Document]. Glob. Health Obs. URL http://apps.who.int/gho/data/view.main.BIRTHSBYCAESAREANv (accessed 11.4.19).
- World Health Organization (Ed.), 2016a. WHO recommendations on antenatal care for a positive pregnancy experience. World Health Organization, Geneva.

- World Health Organization, 2016b. WHO Maternal Mortality [WWW Document]. WHO. URL http://www.who.int/mediacentre/factsheets/fs348/en/ (accessed 12.21.15).
- World Health Organization, 1999. Reduction of maternal mortality. Geneva, Switzerland.
- World Health Organization, 1985. Appropriate technology for birth. Lancet 2, 436–437.
- World Health Organization, Department of Maternal, N., Child and Adolescent Health, 2013. WHO recommendations on postnatal care of the mother and newborn.
- World Health Organization, International Bank for Reconstruction and Development/The World Bank, 2017. Tracking universal health coverage: 2017 global monitoring report: executive summary. World Health Organization, Geneva, Switzerland.
- Yanagisawa, S., Oum, S., Wakai, S., 2006. Determinants of skilled birth attendance in rural Cambodia. Trop. Med. Int. Health 11, 238–251. https://doi.org/10.1111/j.1365-3156.2005.01547.x